A40/Technical Discussions/2

Economic Support for National Health for All Strategies



Background Document

Fortieth World Health Assembly Geneva - May 1987 A40/Technical Discussions/2

Economic Support for National Health for All Strategies



Background Document

Fortieth World Health Assembly Geneva - May 1987

© World Health Organization

y,

This document is not a formal publication of the World Health Organization (WHO), and all rights are reserved by the Organization. The document may, however, be freely reviewed, abstracted, reproduced or translated, in part or in whole, but not for sale or for use in conjunction with commercial purposes.

Contents

G

LACCULIED D	ummary 5	
Introduction	13	
Chapter I	The world economic climate and its implications for health for all 19	
	Background 19	
	Trends in the world economy20The economic recession and the public health sector24	
	The recession and social development28Political determination for health for all in times of economic adjustmentsConclusions37	31
Chapter II	Planning and managing the finances 39	
	Background 39	х.
	Financial master planning 40 Closing the resource gap 41	
	Programme budgeting 44	
	Coordinating revenues and expenditures 45 Conclusions 48	
Chapter III	Mobilizing resources 51	
	Background 51 Criteria for choosing a financing strategy 51 Options for financing strategies for health for all 55	
	Community financing 64 Conclusions 70	
Chapter IV	Making better use of resources 71	
	Background 71	
	Improving accountability 72 Improving efficiency 76	
	Improving cost containment 86	
	Conclusions 90	
Chapter V	Responsibilities and institutional relationships in economic support for health for all 93	
	Background 93	
	Government responsibility 93	
	Individual and family responsibility 96 Role of the private sector 98	
	Community responsibility 100	
	Nongovernmental organizations 101	
	International cooperation 102 Institutional relationships 104	
	Strengthening national capability 105	

3

CANDER STATES AND SEA A

Contents

Chapter VI	Key issues for d	iscussions	109			
	Economic policies f	or equity in health	111			
	Financial planning	113				
	Resources mobilizat	ion 115				
	Making better use o	f health resources	117			
Annex 1	Glossary of Term	ns 119				
Annex 2	Evaluation of the Global Strategy for Health for All by the Year 2000 – Global indicators on resource allocation					

2

and an a start and a start of the and the second second

Sec. Sugar

10 y nga shina Me

2.1

Executive summary

Ten years ago, the World Health Assembly decided that the main social target of governments and of WHO should be the attainment by all the people of the world by the year 2000 of a level of health that would permit them to lead socially and economically productive lives. People must be healthy to contribute to and share in social and economic development and conversely, development goals cannot be achieved without a healthy population. The historic International Conference on Primary Health Care in Alma-Ata, USSR (1978) charted a new course for the health of the citizens of the world. The Conference called for a new approach to health and health care to shrink the gap between the "haves" and "have-nots" and to achieve a more equitable distribution of health resources. The Conference further affirmed that the primary health care approach was essential to an acceptable level of health throughout the world and acknowledged that this could be attained through fuller and better use of the world's resources.

Primary health care is essential health care made accessible at a cost the country and community can afford, with methods that are practical, scientifically sound and socially acceptable. Everyone in the community should have access to it and everyone should be involved in it. Governments together with the World Health Organization in a spirit of social justice have endorsed primary health care as the key to attaining the goal of Health for All by the Year 2000.

The Global Strategy for Health for All, adopted by the World Health Assembly (1981) stressed the close and complex links that exist between health and socio-economic development. Health not only results from genuine socio-economic development, as distinct from mere economic growth, it is also an essential investment in such development. The achievement of health goals is determined to a large extent by policies that lie outside the health sector and in particular the socio-economic development policies. The Strategy therefore emphasized the mutual reinforcement of related policies. These principles were reaffirmed during the Technical Discussions on "Intersectoral Action for Health" (1986) which further clarified the essential elements of equity-oriented development strategies and the respective roles of the crucial development sectors such as agriculture, education and environment in promoting and contributing to health goals. The Technical Discussions also highlighted the health-related policy components in the crucial development sectors.

Political determination for health for all

The overall efforts of governments to develop their health systems in response to their national health strategies has been encouraging, as was revealed by the first report on the evaluation of the Global Strategy for Health for All (1986). A high level of political will is evident. There is a growing awareness of the need for change in the health systems. In some countries impressive efforts have been made to expand the health services infrastructure. Some innovative

In the second second

approaches to reach underserved population groups and to strengthen community-based health services are also noted. The overall benefits however have sometimes been less than expected because of factors such as political instability, natural disasters, and high population growth. In addition, the intervening decade has been marked by economic instability that has greatly impeded social progress in many parts of the world.

Implications of the economic climate for health for all

The global economic situation has changed drastically for the worse since 1977 when the goal of Health for All was adopted. The past seven years have been the most turbulent times for the world economy in over half a century. The recession has had important implications for adjustment policies which governments have had to adopt in order to maintain a reasonable. balance between economic growth and social development. This has not been an easy task for a majority of the developing countries, especially the poorest countries of the world. The health budgets of many countries have been severely reduced at a time when additional resources are required to build and sustain national health systems based on primary health care to meet the priority health needs of all people, especially the vulnerable and the underserved.

The acceptance of the goal of Health for All was accompanied by the fundamental concept of equity in health. Making quality health care available at an affordable level, however, remains a distant prospect in many countries. Aspects such as the impact of available health care as well as concerns with efficiency and cost-benefits are being balanced by countries against a more equitable distribution of health-related resources in order to bring health care to the vulnerable groups of their societies. The danger inherent in this balancing act is that concern with equity can give way to concern for "cost-containment" which, in practice, can be translated to mean reductions in the social expenditures for those who need them most. The manner in which the health goal is incorporated into the overall strategy as a goal of development also affects the allocation of resources and, in turn, the issue of equity.

But external economic factors cannot be entirely blamed for the under-achievement in the health sector. Many governments as a whole have not seriously taken up the strategic actions required to generate and mobilize all possible resources for health. Very few countries have attempted to make an estimate of the magnitude of resources required for their national strategies to achieve the goal of health for all. Altogether, very few new initiatives have been undertaken to mobilize resources internally which can have national impact. Few countries have been able to reallocate their existing health budgets preferentially to primary health care. Inefficient use of existing health resources persists; effective actions to reduce waste or to improve cost-effectiveness have been too few to have a substantial positive impact on the resource situation. The health secto

remains a "weak partner" in influencing socio-economic development policies or in mobilizing effective support from other related sectors for health activities.

Economic support for health for all

How to finance health plans and how to make the best use of resources have both become critical issues in progress towards the attainment of Health for All. While financial cutbacks present major problems in the short run, in the long run the search for additional and new resources, particularly domestic resources, and making the most efficient use of all available resources offer the cest options for financing health for all. Improved financial planning and management and bold administrative ind organizational measures would also be required. The strengthening of national capability especially that of the ninistries of health or equivalent bodies, in developing and implementing policies based on sound economic analysis and strong financial management will be a prerequisite to effective national action.

The purpose of this background document is to focus attention on the options for strengthening and expanding economic support for national strategies for health for all. The document suggests that five broad needs ought to be addressed. These needs are: (i) to project the financial requirements imposed by national health policies and to assess the capacity of currently available revenue measures to meet those needs; (ii) to evaluate the quality of resource mobilization efforts in terms of their equity, adequacy, reliability, impact on the supply and demand for services, and ease of administration; (iii) to increase the productivity of the resources available to governments by focusing attention on efficiency and cost-effectiveness of activities undertaken by the health sector; (iv) to reappraise the roles and responsibilities of potential partners in health including the government, community, nongovernmental organizations and the private sector, and (v) to strengthen national capability in formulating and implementing sound economic policies and approaches in support of national strategies for health for all.

Planning and managing the finances

The elaboration of a well-defined plan of action, including a financial master plan, is an essential part of the strategy for attaining health for all. The financial implications of declared health policies need to be assessed carefully. Financial master planning can provide a framework for assessing the feasibility of implementing the plan of action against the resource availability. It can identify the resource gap and suggest options to close this gap including either increasing resource availability or modifying the implementation objectives.

Such plans would estimate the capital and recurrent costs of implementing the countrywide programmes included in the proposed plan of action and would

Economic Support for National meanin for All Strategies

identify sources of funds to meet these requirements. The process seeks to identify the boundaries within which a health plan could be implemented. Among the essential steps are: estimating the costs of meeting proclaimed health goals and distributing these costs over a period of time between the initiation of the plan and the year 2000; comparing recurrent costs with the revenues likely to become available from existing sources of finance; exploring all possible sources of finance; and reconciling planned expenditure with the revenue both from existing and further sources of finance. If resource gaps exist, either the plan of action should be modified or mechanisms to mobilize new resources should be suggested.

While a financial master plan will help to establish the extent to which a policy can be implemented, programme budgeting is required to enhance the efficiency and effectiveness of implementation. The budgeting system should stress the links between programme objectives and the use of resources and the relationships between capital expenditures and recurrent costs.

Sound financial planning and management are thus essential to the implementation of national health-for-all strategies. Decisions on mechanisms to finance health plans are inevitably political. The task of planners is essentially to develop options for political decision-making. Ministers of Health need to encourage their planning staff to engage in creative thinking, even if some of the options presented prove to be politically unacceptable.

Mobilizing resources

The desired increases in coverage, an the maintenance and improvement o content and quality of health services will require additional resources in many countries. Available governme revenues in most cases will not be sufficient to cover the requirements. New and alternative options will hav to be considered. Strategies for financing health for all will no doubt reflect the various characteristics of national economies. ·1

In choosing a strategy, however, certa criteria should be applied. These include equity, adequacy, reliability, impact on supply and demand, intersectoriality and administrative feasibility.

Many options for financing health services are now being widely considered. First, governments may p for health care from public revenues. This requires that additional resource be allocated for health activities. Second, employers and employees ma be required to contribute to a health insurance scheme, or employers can provide health services for their employees. Third, institutions, public' or privately owned, may be created to attract voluntary insurance contributions and to dispense these revenues to providers of health servic Fourth, schemes of community financing may be developed. Fifth, consumers may be required to pay fo part of the cost of the health services they use. Many variations of these options are being devised. Each of the options has distinctive economic, financial, political and administrative

attributes which need to be carefully examined.

Making better use of resources

There is general concern in all countries that the available resources are not being used most effectively or efficiently. A large share of health resources is wasted because of poor managerial practices and use of inappropriate technologies or human resources. Making better use of resources implies improved accountability, increased efficiency in the allocation and utilization of resources and effective means of cost-containment.

Accountability can be improved by strengthening formal accounting and management information systems and by providing adequate and supportive supervision. Resource wastage due to misappropriation, underemployment or deterioration must be given serious attention by health managers. In many cases communities can also be more actively involved in the management of resources at the local level. They will need to be supported in this task by the central levels.

Efficiency in the use of resources can be enhanced by several means. If access to health services were more equitable, cost-efficiency would generally be greater. The cost to the providers as well as to the individuals must be considered. Many options are available for achieving greater efficiency in the use of human resources. The main objective should be to make rational use of health personnel consistent with the functions of each level of the health system. Appropriate training and supervision would need to be provided to ensure quality and performance. Individuals and families can be informed and educated to take a greater share of responsibility for their health. Careful choice of technologies appropriate to each level of the health care system would also serve to increase efficiency. Cost-effective strategies can be applied to specific health problems and, finally, the strengthening of management support services would be crucial to sustain efficient delivery of health services.

Cost-containment policies have begun to emerge particularly in many developed countries. Many new approaches are being applied to influence both the supply and the demand of health services. These have included a revision of reimbursable fee scales, an imposition of user charges and a regulation of the content of care through review committees. Public education and information on these aspects is also receiving considerable attention. An informed public can be a valuable asset.

Responsibilities and institutional relationships

Health care is a shared responsibility which rests upon the individual, the community and the government. Within the government responsibility, health goals have to be incorporated as part of the sectoral goals of many different sectors including health, environment, education, agriculture and

Economic Support for ivalional meaning for All Surges

housing. Given the magnitude of the task of attaining health for all, and particularly of securing adequate economic support for this, concerted and coordinated action at all levels is indispensable. Collective commitment of all concerned is required in order to ensure the equitable distribution of resources for health care.

A lack of information precludes a thorough analysis of the respective share of the responsibility and contribution of the different entities involved in health matters. But it is clear that, in many developed and developing countries, the public sector controls only a portion of the overall resources available for health care. While no blueprint can be applicable for organizing the institutional relationships of the different entities or sub-systems involved in health matters, it is clear that greater coherence needs to be achieved among these sub-systems and that all sub-systems should reflect primary health care as their major goal.

Collaboration between various institutions and agencies is essential and should be based on a clear allocation of responsibilities in order to ensure the most efficient use of resources. Countries need to examine what is the feasible organizational framework in their own national situation which would enhance such collaboration.

Strengthening national ` capability

Mobilizing economic support for health for all has many implications for national health policy-makers and health administrators. The policy-makers in health need to be strong advocates for promoting soc priorities in economic adjustment policies. They must mobilize commitment and support from othsectors, especially those closely rela to health. The health administrator must increase their capability in defining equitable schemes of finance and of allocation of resources. They must be able to provide policy-mak with different options for mobilizing additional resources. And finally, th must manage their scarce resources most optimally and efficiently.

These implications also suggest the for substantive improvements in information for health planning and management, for the development c capacity for research, processing and analysing economic data, and for the acceptance by senior managers and policy-makers of the importance of these new inputs into the process of decision-making.

Technical capability in the health sein these areas will require consideral strengthening in most developing countries.

Conclusions

The goal of Health for All by the Y 2000 has been called ambitious. The who regard it as such disregard the fundamental principles embodied in primary health care approach, which the key to achieving the goal.

Primary health care emphasizes hea as an integral part of development :

and a second second

thus, a responsibility of not just what is traditionally defined as the health sector, but of people, other related sectors and the community in general. Primary health care calls for the use of affordable, relevant and socially acceptable technologies, and requires that the strengthening or building up of the health infrastructure should begin in the home and at the community level, the other levels of the health system being supportive to these.

Respect for these principles requires a major reorientation of policies and perspectives in the way health is perceived, protected, promoted and delivered.

These principles also apply to resource allocation and distribution policies. A concern for the care and protection of the poor and the disadvantaged groups, the according of high priorities to prevention and promotive actions, and the use of inexpensive yet effective technologies to provide at least the eight essential elements of primary health care, should be reflected in such policies if choices and sacrifices have to be made. Resource allocation policies have to consider not just what goes to health care, but also to other important determinants of health such as education, environment and food, again with due considerations for principles of equity.

The issue is not where the money is going to come from to pay for health care, but rather what broad policy framework is required to expand the economic support for health for all. This support must come from individuals, families, communities, the private sector, nongovernmental sectors and, of course, government sources. The issue is not just how much more resources will be required and how to mobilize them, but also how can those resources that are available be used most efficiently and productively.

Discussions on economic support for health for all should not be clouded by a narrow vision of financing primary health care or medical care by the public or private sector. They should encompass the policy and institutional frameworks which will provide a coherent strategy for the full, active and mutually reinforcing participation of all potential partners. No one should be allowed to escape this responsibility. This is not just a political or social dream, it is an economic challenge.

Achieving health for all will require sacrifices. The mechanisms and methods used to finance and support services will continue to be imperfect. The task of finding long-term solutions is difficult but must be faced if the health of the future generations is not to be jeopardized.

Introduction

The world economic situation has changed drastically for the worse since 1977 when the goal of Health for All was adopted. The health budgets of many countries have been severely reduced at a time when additional resources are required to build and sustain national health systems based on primary health care to meet the priority needs of all people, especially the underserved. The reductions in the health budgets have tended to affect key items such as drugs, equipment, communications and transport, seriously jeopardizing the delivery of essential health care in rural areas. Policies and plans which have been approved in principle at the highest government level have not been fully implemented.

The adjustment policies adopted by governments in times of economic crisis, as in the recent widespread economic recession, are important indicators of the priority which health care receives in the national allocation of resources. Often when resources are constrained the relative low priority for health in resource allocation and the neglect of the social costs of adjustments become pronounced.

Economic support for health for all

It was recognized at the outset that to carry out the Global Strategy for Health for All would mean generating and mobilizing all possible resources. This implied first of all making the most efficient use of existing resources both within and among countries. At the same time, additional resources would need to be generated. Among actions specified for providing economic support for health for all were: careful analysis of the resource needs for the provision of primary health care, particularly for underserved population groups; consideration of the benefit of various health programmes in relation to the cost, as well as the effectiveness of different technologies and different ways of organizing the health system in relation to the cost: estimation of the order of magnitude of the total financial needs to implement the national strategy up to the year 2000; and consideration of alternative ways of financing the health system including mobilizing domestic and external resources.

There is evidence to suggest that governments as a whole have not yet given serious and in-depth consideration to these strategic actions. Very few countries have costed their plans for health for all, let alone identified sources from which they can be financed. Some plans which were costed before the worst effects of the economic recession were known now seem to be over ambitious. Altogether, very few new initiatives have been undertaken to mobilize resources internally which can have national impact. Few countries have been successful in reallocating their existing health budgets preferentially to primary health care. The role of the private sector in national health-for-all strategies has not been fully defined. Even in countries where recent economic improvements have been noted at the individual and family levels, the role of the individuals in sharing the cost of health care has not been reconsidered. With the economic

Economic Support for National freature for All Otheregies

problems affecting the developed countries as well, the anticipated increases in health aid from rich countries to the poor have not come forth.

How to finance health plans and how to make the best use of resources have both become critical issues in progress towards the attainment of health for all. While financial cutbacks present major problems in the short run, it is nevertheless true that, in the long run, the search for additional resources, including new cost-sharing mechanisms, can provide countries with fresh opportunities to look again at how they are using their existing resources. The pros and cons of strategic options for ensuring the future funding of the total capital investment costs and the implied increasing burden of recurrent expenditure are worthy of being analysed. Also, the health policies, the financial management measures and the administrative and organizational structures may need to be reviewed for possible adjustments.

Objectives and scope of the Technical Discussions

To respond to the need felt by Member States to develop and improve their economic capability in the health sector, the Executive Board, at its meeting in May 1985, decided that the subject of the Technical Discussions in 1987 would be "Economic Support for national health for all strategies". The aim of the Discussions will be to clarify issues and identify options for action on ways of mobilizing and using resou optimally when developing or reshhealth systems, to ensure a balance between the cost of effective and practical health plans and the resou likely to be available.

The 1987 Technical Discussions are part of a long-term process aimed a following four objectives:

- Firstly, enabling health officials to communicate more effectively w ministries of finance and plannin especially when designing their health plans and submitting the for future funding;
- Secondly, providing a better understanding of health sector is as well as the broad, health-rela intersectoral issues pertaining to economic and financial aspects (health for all;
- Thirdly, stimulating the use of economic analysis as one of the important tools for better policy decision-making; and
- Fourthly, enhancing commitment to reducing the gap between curren practice and desired actions on priority health issues in resource allocation, mobilization and utilization.
- An additional expected outcome is greater clarity and resolve amon the international agencies on hothey may most adequately suppo countries in their efforts to addre financial and economic issues impeding the attainment of the g of Health for All.

Scope of the background document

In order to support the Technical Discussions, this background document explores the critical issues in planning and managing the financing of the health sector. It also presents and analyses alternatives for mobilizing additional economic support for national health-for-all strategies and addresses the measures for obtaining maximum impact from health resources, with emphasis on social relevance, equity and efficiency. Finally, it explores responsibilities and institutional relationships in economic support for health for all.

No attempt has been made to cover the entire spectrum of issues in the Background Document in order to avoid becoming too superficially comprehensive. To help Member States to address, practically, the economic implications of attaining health for all in a spirit of self-reliance, and as a complement to the 1986 Technical Discussions, on Intersectoral Action for Health, a deliberate effort has been made to concentrate essentially on one vital aspect, namely domestic resources. This does not mean that the question of external resources will not be considered. However, the emphasis will be mainly on the implications of the use of domestic resources on the sustainability of activities initiated, when countries have to support fully their operating costs.

Given the complexity of issues and the diversity of country and sector situations, the document makes no pretension of giving a particular set of prescriptions for effective health financing nor does it suggest that there are universally applicable ways of solving the problem of health financing. The intention is to present a balanced picture of potential measures.

The world economic climate and its implications for health for all

Chapter I describes the global economic climate and its implications for the health sector. Within this macro-economic context, health sector issues to cope with financial problems are pointed out in terms of equity, efficiency and quality of health care. The need for economic support for pursuing equity-oriented health policies is emphasized.

Planning and managing the finances

Chapter II focuses on improved financial management as a way of increasing the effectiveness of available resources. Financial planning is described as a device for determining the financial feasibility and sustainability of strategies and strengthening the health sector's position when requesting funds in the future. Programme budgeting is discussed as a control tool over the use of resources. Accounting practices are reviewed to improve fiscal management. Lastly, economic analysis to ensure adequate financing arrangements is explored.

Mobilizing resources

ร**ะสมกรร**กกระวิทยาส์สามารถ และสามารถสามารถสามารถสามารถสามารถสามารถสามารถสามารถสามารถสามารถสามารถสามารถสามารถสาม

Chapter III presents alternative ways of mobilizing financial resources at the community and national levels for health systems based on primary health care. These include insurance schemes, community financing, cost-sharing and other mechanisms. The criteria by which alternatives might be evaluated are provided. These methods determine the ability of the sector to cope with financial uncertainty to meet particular needs such as importation of drugs and spare parts. Thus the chapter analyses the capacity of different mechanisms to generate funds in order to meet needs for foreign exchange, to accommodate growth in demand, to absorb price inflation, and to service the contractual obligations of the sector.

Making better use of resources

Chapter IV addresses the measures for obtaining the maximum impact from health resources with emphasis on social relevance, equity, efficiency and cost-effectiveness. The chapter first discusses ways of reducing waste in the use of all health resources and mechanisms for improving accountability. It then presents a framework for the efficiency of health services at all levels. Alternatives for cost-effective access to health care and strategies, including the use of appropriate technology and manpower mixes, are examined. Measures for cost-control and cost-containment are considered.

Responsibilities and institutional relationships in economic support for health for all

Chapter V discusses the broad areas of responsibilities of different entities involved in health matters. A central concern in formulating national policies on financing health systems based on primary health care is to ensure that resources are used optimally. This requires that responsibility for performance is clearly defined and that adequate mechanisms are establisl to bring about coordination and effective partnership arrangements

Key issues for discussion

Finally in Chapter VI some issues been selected to help focus discuss on what appears to be critical to the attainment of economic support for national health-for-all strategies. The are based on the analysis offered in above relevant sections.

The background document also contains a glossary of the terms mcommonly used in economics (see Annex 1).

The economic issues influencing he and social development are vast a complex. They are also closely interlinked to the political, social a economic structures and environm of countries. However, there is no doubt that the economic issues are concern equally to the developed a developing countries, and to the centralized or the market-based decentralized economies and politi structures. The composite of issues themselves may be different; for example, in the developed countriwhere the rising cost of health care of great concern, high priority may accorded to issues of cost-containr and cost-sharing, at the same time maintaining levels of quality and effectiveness. On the other hand, i: developing countries, where the provision of at least the essential elements of primary health care to is the central goal, issues relating to achieving equity through sound planning and through efficient and effective use of resources may be paramount.

a kanala an

No single document can address all the issues adequately or in depth, and thus no attempt has been made to cover the entire spectrum of issues. Moreover, serious deficiency in information related to the availability, distribution and use of financial resources poses further limitations on any analysis offered.

Ŧ

It is expected that the Technical Discussions will provide a unique opportunity to the participants to exchange views and experiences and to enrich the debate on the options and strategic actions which would serve to improve the capability of Member States to mobilize and manage economic support for health for all.

HEBARDARDS HURDEI VOOR VIELEN VOOR VAN BROOM VOOR VOOR VOOR VAN BROOM BRO

Chapter

The world economic climate and its implications for health for all

Background

Recent changes in the economic environment have seriously affected the efforts of many countries to implement their strategies for health for all. These economic constraints can at the same time create opportunities for the health sector to play a more dynamic role in dealing with them. The adverse economic situation which still faces some countries in some continents will no doubt continue to challenge policy-makers seeking to achieve a balance between economic and social goals.

Economic and social development are interlinked. The world economic climate has crucial implications both for health and social development. Where there is increasing scarcity of resources, adjustment efforts in macro-economic policies generally force the social sector to adopt stringent priorities. Efforts which aim to achieve economic growth have to be balanced with those aimed at protecting and promoting human development and pursuing equity-oriented policies. Reduced economic support for the health sector can have serious implications for the effectiveness. efficiency and equity of health services.

CODDERRORS STRAD

Identifying the seriousness of these implications and finding new ways to deal with them is the critical challenge for the policy-makers. New opportunities for improving the quality as well as increasing the quantity of economic support for national health-for-all strategies need to be explored and be closely linked to the overall development policies of countries.

The ability of countries to invest in health and social development is no doubt influenced to a great extent by trends in their economic development. Similarly, the ability of individuals to pay for their own health services depends on their level of living. Moreover, levels of living are themselves important determinants of health status, as poverty and malnutrition greatly increase vulnerability to disease and are barriers to the attainment of positive health.

There is growing evidence to suggest that despite the economic turbulence, a central issue of concern to all nations of the world is that of quality of life, of which health is an essential ingredient. And many developed and developing countries have demonstrated their

The economic development of a country is clearly limited when one out of five children dies before completing one year of life, when a high population of children suffer from stunted growth due to malnutrition, when a tenth of a lifetime is shortened by disease, or when a significant proportion of the adult population is beset with disability and disease.

Economic Support for National meaning of An Suarcyies

political determination to maintain a balance between social and economic development. Health care in the past was considered a privilege for those who could afford to pay, supplemented with charity for the poor; today, health is regarded as a basic human right. This clearly arises out of the recognition that healthy people are essential prerequisites for economic development.

This chapter briefly discusses the world economic climate, its implications for health and the need for continued political determination for social equity in the face of these trends. It contains four sections. The first summarizes trends in the world economy in the early 1980s as experienced by both the developed and the developing countries and stresses their growing interdependence. The second reviews the effects of the global economy on public sector priorities, especially the extent to which the health sector has been affected by these trends. The third discusses the implications for social development, presenting key arguments for equity and intersectoral action for health. Finally, there is a call for political determination for health for all in times of economic stringency, providing some conclusions for future policy directions for economic support for health care.

Trends in the world economy

The past seven years have been the most turbulent times for the world economy in over half a century. Many of the governments of countries achieved independence followin Second World War are confront most difficult economic challeng they have ever had to face. The changes in the economic envirohave seriously affected the abilin governments in developing coun implement their health program and have forced many governm reassess their spending priorities

The world's present economic ci began in 1980 with the onset of deepest economic recession sinc Between 1980 and 1983 per capit production declined in country : country. In the most seriously at countries of Africa south of the Western Asia and Latin Americ. recession erased a decade or mc earlier economic development. 1 early 1980s, average living stanc fell by 9 per cent in Latin Amer by 11 per cent in sub-Saharan A Even the industrialized market economies suffered; the United Kingdom, the USA and the Fed Republic of Germany all recorde drop in total economic output ir or more of these four years. Unemployment rose to persisten such as had not been known for century.

In 1984 a recovery of global eco growth took place but with wide distortion in the international p. Growth in the industrialized couwas nearly 5 per cent. Growth ideveloping countries was arouncent, but growth of nearly 8 per Asia conceals the much lower g rates of the least developed couthat continent, and growth in ot regions was only 1.5 per cent to

Contraction of the second

a 1915 - Santa Sa 1917 - Santa Sa

anga ku ku

ONROTONOS!

un de la companya de la compañía

cent. It is anticipated that improvement in the relative performance of the developing countries taken as a whole will continue. However, the poorest countries, especially in Africa, are not expected to be in a position to make desired changes in their balance of payments. Faced with a food deficit, they need to continue to import essential goods and services, thus incurring more debts.

The recession has been compounded in many countries by political instability, armed conflicts, natural disasters and high population growth. In sub-Saharan Africa, the worst drought in 15 years hit many countries in 1982. This devastating drought brought to a head the longer-term adverse trend in food production in this area which has the highest population growth in the world. In the 24 countries most seriously affected by the drought per capita grain production had been falling by 2 per cent per year since 1970.

The economic stagnation which emerged in the early 1980s has been followed by a deepening debt crisis in many countries. The recession in the advanced economies sharply reduced the demand for the goods exported by developing countries. Deprived of part of their anticipated market and thus confronted by a drop in earnings from exports, the heavily-indebted developing countries encountered mounting difficulties in obtaining the foreign currency needed to import drugs, spare parts and replacements of medical equipment, petrol and other essential materials. The substantial burden of servicing external debts incurred earlier has further intensified

the economic problems of developing countries.

contracts and its implications for realth for de

Over the last year, the sharp drop in oil prices has brought some respite to the balance of payments problems of oil importers. But the problems facing certain oil-producing countries such as Nigeria and Indonesia, where oil represents a high proportion of both export revenue and government revenue, are formidable. Moreover, it has added to the onerous debt-servicing problems that face such countries as Mexico.

These economic problems have forced countries to embark upon economic adjustment programmes which have included efforts to restrain demand for imports and to promote exports. A major tool for achieving these ends has been the devaluation of the local currency in order to make imports more costly and exports more competitive in international markets. Many governments have also attempted to balance their public budgets by reducing public expenditure wherever possible. A number of countries have reduced the burden of taxation on firms and individuals in the hope that this would stimulate initiative and private investment. These additional investments in turn were expected to contribute to increased production and thereby lead to greater abundance of both private and public resources.

The world economic climate during the last decade sharply illustrated the economic interdependence of the developed and the developing countries. (See Box 1.1)

響

ŝ

Box 1.1 Economic interdependence

Changes in the pace of economic growth and inflation, protectionism, and shifts in interest and exchange rates in industrialized countries affect developing countries' economic performance through markets for goods and services and through financial markets.

Trade in Goods and Services

The relative importance of the industrialized countries in world markets for goods has declined moderately in the last fifteen years, but these countries still constitute the major markets for exports and the major suppliers of imports for developing countries. Thus events in industrialized countries have a considerable impact on the prices and volumes of goods and services exported by developing countries. A one percent decrease in output in the industrialized economies produces a two percent decrease in commodity prices. Prices of metals and agricultural raw materials which are used as industrial inputs, more closely reflect cyclical fluctuations in industrial production than do the prices of food and beverages.

Growth in real incomes in industrialized countries also increases the volume of goods imported from developing countries. It is estimated that a one percentage point increase in the rate of growth in real GNP in industrialized countries results in an increase of about 3.4 per cent in the purchasing power of exports by non-oil-producing developing countries.

Increased protectionism in industrialized countries also has a direct impact on developing countries' export earnings. Protectionism lowers the demand for their exports and thereby depresses prices as well as lowering sales volumes. The magnitude of these effects depends as well on the demand in foreign and domestic markets for the goods being traded.

Changes in the prices received for exports and in the volume of foreign sales affect economic growth in developing countries in four ways. Firstly, greater exports raise the overall level of economic activity and thereby increase the rate at which productive capacity is used. Secondly greater earnings from exports may also promote domestic production by providing foreign exchange with which to purchase scarce imported inputs. The magnitude of this effect depends on a country's dependence on foreign goods and the severity of its foreign exchange constraint. Thirdly, an increase in export earnings alters a developing country's long-term growth prospects by enabling it to raise its rate of investment Higher export earnings permit larger imports of capital goods; this offers the opportunity of increasing the amount and pace of investment. (Cont

Box 1.1 (Cont.)

Fourthly, an increase in exports may shift resources from the rest of the economy to the export sectors. This will often accelerate growth because the productivity of the economy is greater in the export sectors than in other parts of the economy. Faster export growth may also encourage other sectors to adopt more productive methods and better management techniques.

Financial Markets

¥,

The financial system also transmits economic upheavals from country to country. During the 1970s developments in the banking industry in the industrialized countries enabled major banks to lend more to developing countries. At the same time the demand by developing countries for loans increased rapidly because of sizeable balance of payments problems caused by two oil shocks. The attractiveness of bank loans derived from their flexibility, convenience and low interest rates. However the use of bank loans to finance developments greatly increased the vulnerability of developing countries to events in world financial markets. Interest payments were due irrespective of the uses to which the original loan had been put, and the amounts due depended on world economic conditions as well.

In the 1980s the disinflationary policies adopted by the industrial countries caused borrowing costs to rise. The change in interest rates was especially large when compared with the prices received by developing countries for their exports. The difference between the real interest rate and the increase in the average prices of exports from developing countries was 18 per cent in 1981-82. These changes in the cost of servicing debt significantly affected the developing economies. For example, the debt service payments of Africa amounted to 27 per cent of export earnings in 1983.

The availability of loans to a developing country also depends on its creditworthiness as perceived by banks. When export earnings decline because of a recession in the industrialized economies, the actual and perceived capacity of a developing country to service its foreign debt is diminished. As a result the impact of a recession is magnified by the consequent abrupt reduction in the availability of external financing and the pressure to repay earlier loans.

Source: Adapted from Goldsborough, D. and Zaidi, I. M. How performance in industrial economies affects developing countries. Finance and Development, Vol 23, No. 4, December 1986.

1. 1993年3月1日日日

and a second state of

·>· :::::::::::::::

Economic Support for National Health for All Strategies

The economic recession and the public health sector

The economic recession on the whole has had a profound effect on domestic priorities and programmes in both the developed and the developing countries. The reduced rate of economic growth has resulted in low growth or even a real decline in government revenues.

Developing countries rely heavily on taxes on imports and export earnings to finance the activities of the public sector. The low-income countries collected a smaller proportion of their Gross National Product in taxes in 1983 than in 1973. For the poorest countries and the newly industrialized, middle-income countries, the decline in the value of international trade between 1979 and 1984 was more than twice as large in percentage terms. Thus the contraction of international trade seriously reduced the collection of government revenues in addition to reducing foreign exchange earnings.

Many governments throughout the world have been forced by the economic events to reduce public expenditure and to consider responses which would have the least damaging impact on social development.

In the market-based industrialized countries, it is clear that there has been a considerable reduction in the rate of growth of expenditure on health services in relation to Gross Domestic Product (GDP). For example, in the OECD countries a rate of growth of 4 per cent to 5 per cent per year in the percentage of GDP devoted to health services during the period of 1965-75 was followed by growth of less per cent per year more recently some of these countries health expenditure has been reduced in terms over the past two years. Cost-containment has come to as a critical objective. The high supporting the unemployed, eiti cash benefits or on special sche training or work experience, ha constraint in other fields of pub expenditure. In Europe, cost-sha increased considerably, and ceil have been imposed on public he expenditure and on health expe financed by social security. Long-standing problems in the ; health sector in the developing countries have no doubt been aggravated by trends in the glot national economy. These count faced with the prospects of incr. the utilization of available resou while at the same time making adequate resource allocations ar reducing inequities in both the of costs and in access to service

The key question for developin. countries is whether per capita expenditure in the health sector been rising or falling. Very few countries are in a position to an this question, for a number of r First, public health expenditure divided among several governn departments, levels of governm statutory agencies (particularly security funds), and no data ma available for private expenditu employers, nongovernmental organizations or individuals. Sc a special price index would be to convert current price figures constant price figures, particula period with such abrupt change

and 1983. Comparing these figures with those for 1981, health secured a higher proportion of national budgets in ten countries and a lower proportion in seven. (Table 3).
The trends in public health expenditure,
and the widening gap between public expenditures for health in the developed and developing countries is further indicated in Table 1. Per capita variations in public health expenditures range from below US\$ 2.00 to well over US\$ 500.
Economic crises and the adjustments that result from them exacerbate the "recurrent cost" problem in the health
sector in many developing countries. This problem is the lack of funds in the public sector with which to operate health facilities and programmes in the manner in which they were designed.
When financial limitations affect public health budgets, the cutbacks in recurrent costs mainly affect the critical
inputs required for running services. First cuts are on fuel, vehicles, building
and equipment maintenance, supervision, drugs and other essential inputs. It is easier to cut these than staff salaries. Also lack of foreign exchange for imported drugs, spare or

the world economic climate and its implications for health for all

Table 1: Proportion of public expenditure allocated to health, 1972 and 1982

×.

Countries	1972 %	1982 %
Low Income	6.1	3.0
Industrial Market Economies	9.9	11.7

Source: World Bank. World Development Report 1985. New York, Oxford University Press, 1985.

ABASBARARS

25

and a state of the second state of the second

1.2

 $(\cdot,\cdot) \in \{0,1\}$

and a grant have

Table 2: Current public expenditure on health services as a perc tage of gross domestic product, 1977–83

З

	1977	1978	1979	1980	1981	1982	1983
	%	%	%	%	%	%	%
Developed countries			• •	2.1	3.0	3.0	3.1
Australia	3.2	3.1	3.0	3.1 4.4	3.0 4.5	3.0 4.5	4.5
Austria	4.1	4.3	4.3 3.8	4.4 3.9	4.5	4.1	4.1
Finland	4.0	3.9	0.4	0.5	4.0 0.5	0.5	
France	0.4	0.4 5.7	0.4 5.7	5.9	6.1	5.9	5.8
Federal Republic of Germany	5.7 1.3	5.7 1.4	1.5	1.7	1.8	1.9	2.0
Greece	3.3	3.4	3.6	3.6	3.9	3.9	
Italy	0.4	0.4	0.4	0.4	0.4	0.4	0.4
Japan	3.4	3.9	3.7	3.9	4.0	4.5	4.5
Malta	4.1	4.2	4.1	4.0	4.1	4.3	4.4
Norway	7.0	7.0	7.0	7.3	7.4	7.5	7.3
Sweden United States of America	1.1	1.1	1.1	1.1	1.1	1.2	1.2
	4				2 - 1 - 1 		
Developing countries African Region				•			
Cameroon	0.7	0.8	0.7	0.6	0.6	0.7	0.8
Kenya	1.8	2.1	2.1	2.3	2.3	2.2	2.0
Lesotho*	1.3	1.2	1.8	2.5	1.8		
Mauritania	2.8	2.7	2.4	2.6	2.5	2.4	2.3
United Republic					• •	16	1.5
of Tanzania*	1.8	2.1	1.9	1.7	1.6	1.6 1.8	1.5
Zimbabwe	1.3	1.6	1.4	1.6	1.6	1.0	1.7
Region of the Americas							
Colombia	0.6	0.6	0.7	0.9	0.9	0.9	
Ecuador	0.6	0.7	0.6	0.8		1.0	1.0
Honduras	1.9	1.8	1.8	1.6	1.5	1.4	1.5
Panama	1.8	1.6	1.8	1.5	1.5	1.7	1.7 2.2
Peru	1.5	1.5	1.6	1.8	2.0	2.1	2.2
Saint Vincent					5.0		
and the Grenadines	4.2	4.7	3.9	3.8 1.5	1.9	1.9	2.5
Venezuela	1.9	2.0	1.6	1.5	1.5	1.5	2.0
South East Asia Region	· · · · · · · · · · · · · · · · · · ·					07	0.6
India	0.6	0.6	0.6	0.6	0.6	0.7	0.6 1.1
Sri Lanka	1.2	1.2	1.1	1.1	1.0	1.0	0.7
Thailand	0.5	0.5	0.6	0.6	0.7	0.7	0.7
European Region	1.6	1.6	2.0	1.9	1.8	1.8	1.8
Israel		-					
Eastern Mediterranean							_
Region	1.3	1.2	1.3	1.5	1.6	1.7	1.7
Cyprus	1.0	1.0		1.5	1.3	1.2	
Islamic Republic of Iran	1.0	0.9		1.2		1.3	
Jordan Kuwait	1.4		1.1	1.4			
Ruwan Pakistan	0.7		0.5	0.4	0.4	0.4	0.5
						(
Western Pacific Region	2.1	2.1	1.9	1.9	2.1	2.3	
Fiji	Z. I			0.2	0.2	0.2	
Republic of Korea	2.9	2.7	2.9			3.3	
Tonga	2.0		-				

Notes: * A 'least developed country'. Source: United Nations National accounts, statistics, main aggregates, and detailed tables, 19: computer tapes. New York, United Nations, 1985.

Table 3: Percentage of central government expenditure spent on
health in countries where central government received
90 per cent or more of tax revenue, 1977–84

	1977 %	1978 %	1979 %	1980 %	1981 %	1982 %	1983 %	1984 %	1985 %
Developed countries									
Belgium	1.78	1.79	1.86	1.65	1.70	1.65			
France	14.59	14.83	14.99	15.01	14.72	14.60			
Greece	8.09	9.89	10.48	10.34	10.54				
Italy		7.55	10.47	12.55	10.70	10.64	11.52	11.49	
Luxembourg	2.07	2.27	2.02	2.15	2.37	2.23	2.21		
Netherlands	11.79	11.87	11.71	11.68	11.63	11.62	11.29	10.97	
New Zealand	14.98	15.04	15.21	15.17	14.24	13.52	12.65		
Developing countries African Region									
Kenya	8.16	7.45	7.23	7.83	7.81	7.33	6.96		
Lesotho	5.43						7.18		
Liberia	7.89	8.21	6.13	5.20	7.61	7.17	7.27	6.20	
Malawi*	5.40	5.27	5.30	5.53	5.16	5.23	6.77		
Mauritius	7.98	8.15	8.04	7.48	6.97	7.10	7.84	8.10	
Swaziland	6.48	4.91	6.29	7.15	5.41	7.12	7.37		
Zaïre	4.01	3.94	3.22	2.47	2.61	3.20			
Region									
of the Americas		0.05	0.54		0.54	<u> </u>	F 0F	C 10	
Chile	6.86	6.85	6.54	7.37	6.54	6.80	5.95	6.18	
Costa Rica	3.31	25.44	25.00			32.76	22.48		
Dominican Republic	8.98	9.43	9.07	9.30	9.70	10.66	10.55		
Mexico	4.38	3.97	3.90	2.37	1.86	1.29	1.20	—	
Panama	14.50	15.08	12.15	12.71	13.24	13.14			
Paraguay	2.73	2.64	3.67	3.59	4.51	3.67			
Trinidad and Tobago	7.79	6.86	6.36	5.78	5.91				-
South East									
Asia Region	E 00	6.73	6.39	5.28	6.09	6.96			
Burma	5.88	0.75		3.46			3.80		
Maldives*	E 0E	4 10	5.06		4.45	5.83	5.12		
Sri Lanka	5.95	4.19	5.17	4.88	3.54	3.35	5.12	5.45	
Thailand -	4.69	4.39	4.54	4.09	4.23	4.94	5.11	5.45	
Eastern Mediterranean									
Region									
Cyprus	5.42	5.92	6.02	6.07	6.73	7.26	6.79		
Jordan	3.58	3.71	4.10		3.75	3.76	3.63		
Kuwait	5.90	5.89	6.25	5.12	4.89	5.38	6.25	6.27	
Oman	2.65	3.17	3.24	2.92	3.04	3.09	3.47	4.13	
Sudan*	1.45	1.71	1.46	1.40	0.04	1.34			
Tunisia	7.03	7.27	6.43	7.20	7.65	6.66			
Western Pacific									
Region									
Philippines	5.08	4.74	5.54	4.54	5.01	5.28	6.80		
Singapore	7.37	8.50	7.01	6.88	7.18	6.39	6.41		

Notes: * A 'least developed country'.

Source: IMF Government financial statistics yearbook. Washington, DC, International Monetary Fund, 1985.

replacement parts and oil have often been serious. As a result of the trends in the increased proportion of the recurrent budget for salaries, the effectiveness of staff as well as their morale decreases. Because of lack of quality, facilities are under-utilized, especially in the rural areas where a large part of the population is living. The referral and supervision system at the district level has been seriously jeopardized in many countries by cutbacks in recurrent costs, and consequently the delivery of essential health care in rural areas has been curtailed.

The above brief review of the financial problems of public health services should be completed by an analysis of other sources of financing and provision of services. Central governments in many countries rarely contribute more than half of the total funds spent on health activities. Very important are the contributions from the state and local governments, nongovernmental organizations (profit or non-profit making), communities and households. Over the last decade the proportion of central government expenditures for health care declined in many developing countries while it increased in the developed countries. Data on the allocation of such expenditures from these other sources are neither complete or reliable. But it seems that the largest part of these resources is spent in the private sector on high-cost, mainly curative, care.

On the demand side there are national and regional variations. Overall, however, the demand is increasing for demographic, epidemiological, economic and socio-cultural reasons.

The absolute numbers of children and young people in developing countries have been rising. Aging populations and urbanization are increasing the demand for curative services. In many countries infectious and communicable diseases continue side by side with emerging chronic diseases and health problems related to the environment, behaviour and life-style. An increased awareness of problems through the mass-media and education, especially female education and improved access to services are creating a huge potential demand.

Given the lack of uncommitted recurrent budget resources at all levels of the health sector, the feasible scale of reallocation *within* the public health budget will in most countries be too small to allow major, short-term improvements in effectiveness and efficiency. More resources, and the possibility of different sources of finance for health activities, must therefore be sought as a matter of urgency although this may, in some cases, entail a reconsideration of the scope and responsibilities of governments.

The recession and social development

The impact of the economic recession on the industrialized market economies was clearly indicated by the growth of unemployment to an average of around 10 per cent with one country as high as 21.5 per cent. Hardships are particularly severe in countries which have not yet developed elaborate social security and

social welfare schemes and where most families have only one breadwinner. Evidence is accumulating of the different ways in which unemployment can damage health. The relationship is complex and multidimensional.

Urban unemployment has also reached high levels in developing countries nearly all of which lack unemployment insurance. The unemployed join the overcrowded informal sector in an attempt to make some sort of living. Moreover, the unemployed lose their right to use health services financed by social security schemes and the income with which to pay for private services. This adds to demands on the overstrained government health services.

The emphasis on export crops has diverted land from local food production and the increase in the cost of imports, inflation, high housing costs, the abandonment of food subsidies as well as high unemployment and underemployment have made poverty more severe and more widespread in Latin America and the Caribbean. Low birth weights are on the increase in Brazil and children treated for malnutrition have tripled in Costa Rica. Surveys in 6 Latin American countries have all indicated increased inequality and poverty.

¥

Poverty has also increased sharply in rural Africa, particularly as a result of the deterioration in terms of trade, increased protectionism and the recent famine. Certain African countries have suffered outbreaks of yellow fever and cerebrospinal meningitis as well as the effects of the massive displacement of

refugees. Surveys from Botswana and Ghana showed a clear downward trend in the nutritional status of children and in Ethiopia around half of children aged 1-3 are below the standard weight for age. The infant mortality rate rose to 200 or more per 1000 live births in Burkina Faso, Chad, some parts of Ghana, and in Mozambique.

At least half of the world's poorest people live in South Asia where the growth of poverty has been greater in the rural than the urban areas. The recession hit the least developed countries the hardest.

Some national studies on the social impact of the economic deterioration have pointed out increased infant and child mortality, such as in Chile and Costa Rica, aggravated nutritional status of children as in Ghana and Zambia and adverse health effects on adults having lost employment and income. Of course, the effects of the economic deterioration have been the most evident among the developing countries and within those countries among the poorest people. The health effect of a given total decline in income depends on how the decline is distributed and how the quantity and quality of available health-related services are reduced. Because health determinants are a combination of factors, the health effects of recession and policy readjustments are poorly documented and cannot be easily interpreted.

In general it can be said that the main impact of the recession has been borne by those least able to sustain it, simply because they have neither the political muscle to prevent it nor the economic

Box 1.2 Impact of the world recession on the health of the children

In 1982 UNICEF initiated a study on the impact of the world recession on the health of children. The findings of that study were reported in a special issue of World Development in March 1984. The world economy had undergone two years of what has since been recognized to be the deepest recession in the memory of most living people. The recession persisted for nearly two more years. It was then followed by a deepening debt crisis in much of the developing world outside India and China. In many of the poorest countries, the recovery from the recession has been painfully slow. In a number of African countries, economic growth has not returned to a rate that matches the rate of population growth.

Twelve countries were selected for detailed investigation under the UNICEF study. These countries included all regions of the world and a wide range of levels of economic and social development. Because the recession was only half through when the study was initiated, its effects were presumably not fully manifest. The study found that the greatest suffering seemed to have occurred in Africa. Countries in East and South Asia appeared to weather the first half of the recession with little notable effect on health. A few countries strengthened their programmes to promote health during this period and appear to have registered gains in health in spite of worsening economic conditions.

The UNICEF study underlines the fact that sensitive indicators of deterioration of health are not available even for small children. The inability to identify changes in mortality and morbidity is in part a consequence of the inability of routinely collected data to reflect changes. The countries studied reported only a limited number of diseases and thus it is possible that problems were not captured which are nonetheless important. In addition many of the consequences of economic deprivation are likely to become manifest only after a significant period of time, a longer period than the recession had yet experienced at the time of data collection.

A more likely explanation for the lack of dramatic results is that the quantity of care did not decline as markedly as might have been assumed from reductions in the inflation adjusted budgets of health ministries. Instead, governments reduced non-critical expenditures and allowed staff salaries to decline in purchasing power. Chile for example, had cut its public outlays for development of new health facilities by nearly four-fifths between 1969 and 1980. In addition, payments to staff declined in real terms by about 15 per cent over the same period. (Cont.)

30

un an te staf

Box 1.2 (Cont.)

Expenditures for supplies over the same period increased by about a quarter. Thus the UNICEF study suggests that considerable short-term adjustment can be made in health budgets in order to maintain levels of services. Both public institutions and households appear to be able to reallocate their resources in order to cope with short-term problems of funding. The principal underlying device is to run down equipment, vehicles and buildings by neglecting routine maintenance. These practices may prove costly in the longer term but in the short run allow essential services to be provided on a reduced budget. Moreover, these expedients are only workable for a short period of time.

The conclusion that one can draw from the UNICEF study is that the world recession of the early 1980s did not reduce health status significantly but rather delayed further progress in improving coverage and the quality of services. If this failure to increase the supply of health services were to continue for several years, a widening gulf between needs and responses to those needs would become apparent. The challenge then is to resume the process of expanding the supply of effective health care.

Source: UNICEF. The state of the world's children, 1984.

fat to absorb it. Children have almost certainly suffered most of all in terms of low birth weight, malnutrition, frequency of illness and poor mental and physical growth.

Very few countries have incorporated their health goals into their adjustment policies, in order to minimize the impact of damage to health and protect high-risk groups, most vulnerable to the adverse effects of austerity measures. There is no available evidence that the health sector has suffered from budget cuts more than other social sectors. However, adjustment strategies designed to promote stable economic growth have not taken sufficient account of minimum social needs and equity in the distribution of austerity measures.

Political determination for health for all in times of economic adjustments

The achievement of health goals is determined to a large extent by policies that lie outside the health sector and in particular by policies, whatever their nature, aimed at ensuring universal access to the means of earning an acceptable income. But merely increasing income will not guarantee health. Health authorities will have to display vigilance in identifying aspects of development which can threaten health status and in introducing elements that are essential for health development in their socioeconomic development plans. The linkages between health and development have

MANNERANS STELEN VIEW AND NAMES AND THE STANDARD STANDARD STANDARD THE STANDARD STANDARD STANDARD STANDARD STAN

Box 1.3 Thailand – Incorporating health improvements in social development

Thailand, which is in the lower middle-income category (as defined by the World Bank), had an estimated infant mortality rate of 50 per 1000 in 1983 and a life expectancy of 63 years. It had enjoyed an uninterrupted period of economic expansion during the previous 20 years, sustaining an annual average rate of growth between 7 per cent and 8 per cent. National planners, however, noted that "the social gap between the rich and the poor has been increasing" and that "social services such as health and education" have not been developed appropriately and sufficiently "to reach the low income population, especially in the rural areas".

The regional disparities in income and living conditions have remained high. Between 1960 and 1980 the disparity between the poorest region, the north-east, and the richest, the central region, widened. In 1979, the per capita income in the north-east was 40 per cent of the national average and approximately one-sixth of that of the central region. The main pockets of poverty are in the north-east and north, where the proportions of the population below the poverty line have been estimated at 52 per cent and 23 per cent respectively.

The Prime Minister's office organized the Rural Poverty Eradication Programme in 37 provinces through the country, assigning to four key ministries (agriculture and cooperatives, education, health, interior) the joint responsibility for formulating and executing the programme under the overall coordination of the National Economic and Social Development Board (NESDB). A key strategy in the programme was job creation for the rural poor to narrow income disparity. At this juncture the NESDB established the Social Development Project to support the Rural Poverty Eradication Programme in problem identification, operational planning and management at the village, sub-district and district levels. The Social Development Project formulated basic minimum needs and their indicators and elaborated methods for their use as tools for identifying gaps and proposing priority activities at the village level. The Programme was not initially incorporated in the National Social and Economic Development Plan.

When the Fifth National Social and Economic Development Plan was formulated, the Rural Poverty Eradication Programme was incorporated and renamed the Rural Development Programme, the formulation of which was based on experience gained in the application of the basic minimum needs, mentioned above. Examples are "hygenic nutrition to meet physical needs", "adequate shelter and environmental conditions", (Cont.)

32

TREASURATION AND A DESCRIPTION OF A DESC

Box 1.3 (Cont.)

and "development of pre-school children". The four key ministries were still jointly responsible for the preparation and implementation of the programme plan, with the continued coordination of the NESDB. Job creation was still its main strategy, but in a wider scope and framework, and a larger proportion of the government budget was allocated to the programme. The Social Development Project was terminated at the beginning of 1985. The Government used outputs of the project, particularly the basic minimum needs and their indicators, as a basis for organizing the National Campaign for the Quality of Life (1985-1986), and they have since been incorporated in the Sixth National Development Plan.

Sources: (1) National Economic and Social Development Board, Thailand. (2) Gunatilleke, G., ed. *Intersectoral linkages and health development*. Geneva, World Health Organization, 1984 (WHO Offset Publication No. 83).

been amply demonstrated both by the experiences of developed countries and by the improvement in the quality of ife in several low-income countries. Fen years ago, the Member States of he World Health Organization inanimously resolved that their main social target should be the attainment by all citizens of the world of a level of health that would permit them to lead a socially and economically productive ife. This resolve arose out of a concern o achieve *equity in health*, shrinking he gap in the health status of the eople and countries and ensuring quitable distribution of health resources. This called for a concerted political will and response.

The recent evaluation of the national trategies for health for all undertaken by 147 Member States of the World Health Organization has provided vidence that a positive start has been nade by countries in their quest for ealth for all, against a background of eteriorating economic and social onditions (see Annex 2: Evaluation of he Strategy for Health for All by the

CONTRACTOR CONTRACTOR

Year 2000: Global indicators on resource allocation). No doubt, the prevailing and forecasted economic situation will continue to challenge policy-makers seeking to achieve a balance between economic and social goals.

To protect the poor and vulnerable during the process of adjustment, strategies of adjustment-with-equity are required. Health for all is such a strategy. The reduction of prevailing disparities will require an equitable distribution of health-related resources to bring health care within the reach of vulnerable groups. The causes of disparities in health status can only be removed through intersectoral actions involving the health-related sectors as well as resource allocation policies which give preference to the poor and vulnerable segments of the population, as has been demonstrated by some countries. These aspects were amply discussed during the Technical Discussions on Intersectoral Action in Health (World Health Assembly, 1986). No doubt, the combination of equity,

Box 1.4 Prospects for the health-for-all strategy

Member States in their reports have reaffirmed the validity of the basic principles of the Strategy. None has mentioned the need to modify the Strategy; for some regions in which substantial progress has been made there has been a call for review and modification of the global indicators.

The Strategy is equally valid for the developing and the developed countries; the latter have adapted it to their particular situation and needs. In the final analysis, national and regional variations will become even more apparent as implementation proceeds.

Three scenarios for the application of the Strategy can be distinguished.

At the top end of the scale are the developed countries and a few developing countries which have made substantial progress in improving health status as well as in making available health services for their populations. These countries are now coping with health problems related to increased life expectancy, to life-styles and to the environment, and their attention is turning to measures to reduce the remaining socioeconomic disparities and to cost-containment policies.

In the middle is perhaps the largest group of countries, those which have made significant progress in establishing a health infrastructure based on primary health care. Their situation is changing, but for the time being they have to deal with both traditional and new health problems. They inevitably find themselves facing new demands with strictly limited resources and at the same time trying to improve the coverage of essential primary health care services. The critical challenge for them is to make maximum use of their resources through strengthened management and improved efficiency. They will also have to mobilize additional internal and external resources.

The third group of countries, that with the most critically difficult situation, is still struggling with very high mortality and morbidity, deteriorating socioeconomic conditions and very limited resources to expand the health infrastructure. These countries will require concerted strong support from the international community. They will be called upon to make a serious reappraisal of their national development policies.

Source: Document A39/3 of the Thirty-Ninth World Health Assembly, "Evaluation of the Global Strategy for Health for All by the Year 2000, Seventh report on the world health situation", 1986, paragraphs 476-481.

Box 1.5 China, Kerala, Sri Lanka – The interaction of health and development

In all these areas there was a strong commitment to the goals of equity and concerted efforts to ameliorate the conditions of the disadvantaged and poorer social groups. In all important sectors, the development strategies contained elements aimed at realizing these goals.

The state and public agencies assumed an important role in meeting the basic needs of the people. In China, this was the norm, while in Kerala and Sri Lanka, the supply and distribution of certain goods and services essential to basic needs occupied a central place in public policy and was not left to market forces.

Development policies avoided the urban bias common to the strategies of most developing countries in the early phases of their planning. Consequently, resources for the social and economic infrastructure and investment in development were more equitably distributed. The differences in living conditions between rural and urban areas were not worsened by development. Civic amenities spread to the rural areas. Sri Lanka, for example, was able to maintain a rural/urban balance that limited the internal migration to metropolitan areas.

The political processes were designed such that demands could be formulated and responded to at the community level. In China this was achieved with structures of decentralized decision-making in the communes and lower units. In Kerala and Sri Lanka a highly competitive democratic parliamentary system helped to give forceful expression to community needs and elicit responses from the state.

In economic development programmes, strategies for raising productivity and income in the backward parts of the economy, which contained the poor majority, received priority. Examples are the diversification of the rural economy, and the increase of productivity and output in agriculture, fisheries, energy, and small-scale industry in China, and the drive for food self-sufficiency through programmes for the improvement of peasant farming and small-scale fisheries in Sri Lanka.

In all three areas high priority was given to education. The strategies pursued brought education within the reach of the whole school-age population through a system that provided free or heavily subsidized education. Here again, policies were aimed at the equitable distribution of facilities to provide the rural population with access to education. In all three cases, there was a very high level of female participation in the school system. In China, female participation in primary education was

<u>຺຺຺຺຺຺</u>

CHARGE STRANGER STRAND

(Cont.)

Box 1.5 (Cont.)

97 per cent in 1982; in Sri Lanka, it was 101 per cent.^a According to data available for 1978 for Kerala, the rate was 86 per cent as against 55 per cent for the whole of India.

The improvement in the status of women and the removal of forms of discrimination against females as in the case of education played an important role in enhancing the capacity of the population as a whole for social advancement.

Food security for all segments of the population became an essential objective of public policy. Different policy instruments were used in each case and included state management in the trade in staple foods (China and Sri Lanka), food rationing with food subsidies (Kerala, Sri Lanka), free food supplements for target groups (Sri Lanka), and land reform to provide scope for food production in small allotments (Kerala).

Note: ^a Percentage enrolled in primary school as a percentage of the age group. It can exceed 100 per cent if some pupils are below or above the country's standard primary school age. Source: (1) Gunatilleke, G., ed. Intersectoral linkages and health development. Geneva, World Health Organization, 1984 (WHO Offset Publications, No. 83). (2) Halstead, S. B. et al., ed. Good health at low cost. New York, Rockefeller Foundation, 1985.

effectiveness and efficiency in the context of economic support emerges as a key challenge to the national policy-makers. The alleged trade-off between efficiency versus equity needs to be recast, to demonstrate efficiencyas complementary to equity. Equity represents a highly significant social value, and national political decision-makers cannot isolate themselves or avoid responsibility for decisions which have a profound influence on the level of equity in the assignment of resources and the distribution of health benefits. The making of health policy is a critical task

For the policy-makers in the health sector, the implications are obvious. They need to be strong advocates for promoting social priorities in economic adjustment policies. They must mobilize commitment and support from other sectors, especially those closely related to health. And finally, they must develop their capability in defining equitable schemes of financing and of allocation of resources. Their responsibility will be even greater in managing scarce resources more efficiently.

36

distant of the

for any society. It is an integral part of the political, economic, legal and social structure and is but one part in the decision-making process whereby scarce resources are allocated.

Conclusions

A review of the efforts that have been made to cope with stable or declining resources underlines a serious lack of contingency planning. Little emphasis appears to have been given in the past to planning for variations in economic support for health activities. Instead, governments have approached the management of public finances with considerable optimism about the future availability of funds. Improvements_in economic conditions often have been regarded as if they would continue without interruption. At the same time, major reversals such as those that accompanied the oil shocks of the 1970s have been treated as transient.

Recently, officials in the health sector have had to intensify their demands for both government funds and foreign assistance. The failure to mobilize additional funds has been reflected in the ad hoc deferral of maintenance for equipment and facilities; failure to supply adequate quantities of drugs, chemicals, medical equipment and other essential supplies; and in some countries delayed payment of staff salaries. These problems and responses are similar to those that have appeared in other sectors. Perhaps the most

CONTRACTOR OF CONTRACTOR OF CONTRACTOR CONTRACTOR CONTRACTOR

important lesson to arise from the present economic crisis is that more thorough financial planning is needed than has been practiced in the recent past by public sector agencies. In particular, these experiences suggest a need to devise methods for financing public services that are less vulnerable to short-term economic fluctuations, and a need to develop plans for coping effectively with financial problems whenever they do arise.

Progress towards health for all is not furthered by over-ambitious policies which cannot be financed and fall into disarray when economic development does not progress at the rate which was earlier anticipated. More modest plans which can be sustained are more likely to reach the desired goals. The main aim should be to protect the poor and vulnerable if sacrifices have to be made in response to an economic crisis. But what must never be overlooked is that good health is one of the purposes of economic development and that, in the long run, health promotes economic development.

External economic factors must not be allowed to become the scapegoat for under-achievement in the health sector. There is much that the health sector itself can do to make better use of available resources and, by sound financial planning and stringent managerial procedures, reduce waste and increase efficiency. The health sector needs also to create stronger partnerships with other sectors of developement, with nongovernmental organizations and the private sector, and draw upon the support of local communities by involving them in all its activities at the local level.
Chapter II

Planning and managing the finances

Background

The elaboration of a well-defined national plan of action, including a financial master plan, is an essential part of the strategy for attaining health for all. The sources of finance for the implementation of the plan of action differ widely among countries according to their own particular circumstances. But the need for a national framework to deal with the implementation of strategies and plans of action or their readjustment is especially evident in a period of economic austerity. The recession and the succeeding debt crisis of the 1980s has revealed that the arrangements now being used to finance health-for-all strategies are inadequate in many developing countries.

Chapter I has discussed the impact of the world economic climate on the public sector, especially the public health sector. The facts reveal that the quantity of resources being devoted to health programmes is not expanding as rapidly as was anticipated a few years ago. Some of the poorest countries have relied on continued and even expanded concessionary assistance from others to finance the development of facilities and their operation and maintenance. It is also evident that the developed economies cannot be relied upon to meet these needs, particularly on a long-term basis.

The Global Strategy for Health for All called upon Ministers of Health "to present to their governments a master plan for the use of all financial and material resources". In January 1986,

CAREAR RECORDERSING CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR

the Executive Board of WHO, noting the subject for the present Technical Discussions, urged "those Member States which have not already done so:

- a) to develop further their national strategies for health for all by the year 2000 by producing costed plans for health services and health related activities;
- b) to investigate all possible sources of finance, including the deployment of existing resources;
- c) to ensure that the plans can realistically be contained within the resources expected to be available".¹

This chapter focuses on improved financial planning and management as a way of increasing the effectiveness of available resources and of reducing the vulnerability of the sector to external economic events. The chapter contains five sections. The first discusses financial master planning as a device for determining the financial feasibility of implementing strategies for health for all. The second sets out different possible ways of closing the resource gaps. The third considers the need to improve budgeting and, in particular, to introduce programme budgeting in order to improve budget preparation and control over the use of resources. The fourth examines changes in accounting practices which would strengthen the management of the

¹ Resolution EB77.R11.

39

day-to-day operations of the sector. The fifth discusses choices of sources of revenues to try and ensure the stability of financing arrangements. Finally, the main conclusions of this analysis and their implications for national policy decisions are summarized.

Financial master planning

The financial implications of declared health policies need to be estimated. National plans of action for health for all which are found to be financially unfeasible should be modified, while still ensuring that the compromises and changes are consistent with national health policies. National health plans typically elaborate the investments in manpower facilities, equipment and training that would be required in order to achieve, by a prescribed date, targeted levels of coverage or of access to services. Often the implications of these investments for recurrent outlays are not assessed systematically. Estimates of the future costs of paying staff, maintaining buildings and equipment, providing drugs and supplies, and replacing obsolete and worn out capital items are needed in order to ensure that funds are adequately budgeted. Moreover, there are often competing commitments and promises to provide new hospitals, rebuild hospitals or develop specialized units of which the full long run financial implications have not been calculated. While national health plans specify the expansion of capital and human investments required, they rarely include calculations of the recurrent budgets needed to make full use of these investments. Instead, the

commitments of governments to provide specified services have been conveniently interpreted as implying a pledge to provide the corresponding funds as well.

The events of the past few years have revealed that the financial feasibility of national health plans needs to be appraised in much greater detail. In many countries national health plans have been found to be impossible to fund, and hence to implement. Commitments to provide services often could not be honoured with the financial resources available from government budgets and from such other measures as governments have contrived to generate funds for the health sector. The short-fall in funding has manifested itself in delayed implementation of policies to provide everyone with reasonable access to services, and in a compromised quality of care due to inadequate provision of vaccines, drugs, supplies, staff, transport and maintenance of existing facilities. Hence, well-intentioned commitments to provide universal access to essential health services remain to be fulfilled. In order to meet these pledges improved financial planning and management and better mechanisms for mobilizing funds are needed.

The preparation of *financial master plans* is an essential step towards ensuring that health policies which are reflected in a national plan of action to achieve health for all can be successfully implemented. Such plans would anticipate the capital and recurrent costs of implementing the countrywide programmes included in the proposed plan and would identify sources of funds to meet these

requirements. The process seeks only to identify the boundaries within which a health plan could be implemented. Hence, a master plan may be based on fairly rough estimates of the costs of the plan and compared with modest projections of the growth of national income, tax revenues and other existing sources of finance used to support the health sector.

Among the essential steps are, first, to calculate how the health plans have been financed over recent years, including the contribution of the private sector. The second step is to estimate the costs of meeting proclaimed health goals and to distribute these costs over a period of time between the initiation of the plan and the year 2000. This projection of capital and recurrent costs would then be compared with the revenues likely to become available from existing sources of finance. The third step is to explore all possible further sources of finance. This is discussed further in Chapter III. The final step is to reconcile planned expenditure with the revenue both from existing and further sources of finance. Any gap must be eliminated - if necessary by recasting planned expenditure so that it fits the available revenue by, for example, finding efficiency savings, a less costly manpower mix or using more appropriate technology.

Experience with financial planning

88888888888888888

A few countries have developed full financial master plans. Box 2.1 gives some brief examples.

Closing the resource gap

وريديا المتحجة والمتحدية

While the above countries have found it financially feasible to finance their health plans, many others will find a gap between the resources likely to be available from all possible sources and their first draft of a costed plan. Several options for achieving financial feasibility need to be considered. First, the period over which some parts of the policy are to be implemented may be extended. By taking a longer time to achieve the goals for the content of health care, the extra funds required each year for capital investments and consequential recurrent costs can be reduced. Analysis may also reveal that a greater coverage might be obtained more rapidly if the expansion of coverage to meet the needs of various groups is scheduled more carefully. The cost per person generally will be largest where population densities are lowest and social infrastructure is least developed. Thus coverage can often be accelerated by phasing the expansion, starting with areas most populated. However, considerations of equity and of absolute need may argue for an implementation strategy that maximizes the impact of the government's health activities rather than coverage.

Secondly, the gap between needs and resources may be closed by lowering the goals for the content or extension of services. The gains from reductions in the content of health care are likely to be small since deleting a particular service will significantly affect costs only if it permits the system to operate without a category of staff or a major item of equipment. Targeting the most vulnerable group is also a way to reduce the gap.

1

Box 2.1

Economic Support for ivational meaning of All

Costa Rica plans to achieve Health for All by the year 1990. Despite average income per head of US\$ 1,334, the expectation of life at birth exceeds 73 years and the infant mortality rate is 18 per 1000. The hospital system is already well developed and the use of beds is steadily adjusting to the aging population. One of the main thrusts of policy from 1980 onwards has been to integrate the mainly curative services provided by Social Security with the preventive services provided by the Ministry of Health. In 1985, 65 per cent of the population were serviced by integrated primary health care units: this will rise to 100 per cent by the year 2000. From 1988 it is planned that each member of the population will be able to choose a general practitioner who will provide a 24-hour service as a member of a multidisciplinary team. Services are to be decentralized, local participation increased and intersectoral coordination is to be strengthened. Piped water is to be extended from 88 per cent of the rural population to 100 per cent. The plan envisages an increase of 79 per cent in the current cost of the preventive service provided by the Ministry of Health and of 45 per cent in the mainly curative services provided by Social Security. Within the latter, expenditure on out of hospital services is to grow faster than expenditures on hospital services. The percentage of GNP devoted to health services is to fall from 7.3 per cent in 1987 to 6.7 per cent in 1990 mainly due to a decline in capital construction.

The Netherlands plan is based on a policy document submitted to parliament in 1986 which sets out national policies, targets and strategies up to the year 2000. The plan envisages an annual growth of health expenditures of 1.2 per cent in response to the changing morbidity pattern, particularly caused by the aging population. The different growth rates for the different parts of the health care system are shown below.

	Expenditure in 1986	Annual Growth Rate
	(Guilder millions)	
Administration	2.5	0.5
Specialist care	1.8	0.5
Drugs	3.7	0.8
Hospital care	20.5	1.3
Collective preventive care	0.8	1.5
Primary health care	5.9	1.9

The share of health care in GNP is expected to decrease from 8.3 per cent in 1986 to 7.8 per cent in the year 2000. (Cont.)

42

Box 2.1 (Cont.)

The plan in Sweden is based on a new Act which came into force on 1 January 1983. The policy is to secure good health and health care on equal terms for the entire population. The plan takes account of the growth in the elderly population and provides for effective collaboration between the medical and social sectors. A larger proportion of resources is to be devoted to the prevention of disease and injury, to primary health care, to care in the home and small institutions, and improving the environment. County hospital beds for secondary and tertiary care are to be cut by 28 per cent and psychiatric beds by 50 per cent. Despite these reductions, the cost of the health care system is planned to continue to rise, although at a far more moderate rate than in the previous decades. Health care represented 7.8 per cent of Sweden's GNP in 1983, and will reach 8.0 per cent in the year 2000. Secondary and tertiary care (other than psychiatric) as a share of the total health budget will drop from the 1983 figure of 60 per cent to 48 per cent in the year 2000; psychiatric care will fall from 17 per cent to 13 per cent and primary care will rise from 23 per cent to 39 per cent of the total.

Zimbabwe plans to complete the coverage of the population by rural health centres supported by upgraded and new district hospitals by the year 2000. Among the targets are the reduction of infant mortality to less than 50 per 1000 live births, childhood mortality to less than 20 per 2000 and maternal mortality to 100 per 100 000 with 90 per cent of deliveries attended by trained personnel. The percentage of children immunized is to be raised to 80 per cent. The Financial plan was built up by costing capital developments and their recurrent costs and manpower training requirements, and costing the plan of action so as to add any further expenditures not included in the main programme. About 90 per cent of the additional expenditure required is for primary health care. Health sector spending is planned to grow by 80 per cent by the year 2000 financed by the estimated growth in tax revenue, with the crucial addition of a new compulsory health insurance scheme.

Source: Detailed case studies in finanging health development — options, experiences and experiments, WHO document HSC/87.1.

A third option for reducing the gap between demands for resources and their availability is to reduce the accessibility of care. This option implies either locating facilities further apart or constructing and operating smaller facilities. These reductions in services

HERRERRERRERRERRER

may increase the travel and waiting required for at least some consumers of the health services and thereby discourage their utilization. Whether this will have a serious effect on the health of the people will depend on which demands are left unsatisfied.

43

A fourth alternative for closing the gap is to change the mix of manpower and use lesser trained staff for primary health care. A fifth possibility is to reduce the ratio of staff working in primary health care to the population so as to be able to extend coverage at a faster rate. This may require a tighter definition of the tasks to be performed. Plans vary widely in their manpower complements. For example, one country in sub-Saharan Africa plans to provide primary health care with 6 paid staff per 1000 population and another with 3 paid staff per 1000. Both plans envisage providing all the eight essential elements of primary health care.

The ultimate objective of a financial master plan is to establish mutual consistency of the health goals and funding policies. By establishing the financial feasibility of an overall policy for the sector, governments may avoid inadvertently providing care in a manner which is inequitable and unsustainable.

Programme budgeting

A comprehensive programme to improve financial management also includes strengthening the budget function in order to enhance the efficiency and effectiveness of health care. Striving to increase the efficiency of other public sector programmes would also be useful, but health authorities have little direct influence over them. While a financial master plan will help to establish whether a policy can be implemented, annual budgets identifying those particular investments and activities which

require financial support during the year are required for the implementation of programmes. The budgeting system should stress the links between programme objectives and the use of resources and the relationships between capital expenditures and recurrent costs.

In order to introduce programme budgeting concrete programme objectives are first specified. Detailed plans for achieving those objectives are then devised, and finally the expected accomplishments and the detailed costs of these plans are estimated. During the implementation of the programme objectives, accomplishments and costs are carefully monitored to ensure that plans are realized and/or any necessary modifications are made in the light of experience.

This approach to budget formulation and expenditure control emphasizes achievements rather than resource inputs. It departs from traditional public sector budgeting and accounting in two important respects. First, it stresses the preparation of budgets and summarization of expenditures by objective ("product" or "outcome") rather than by item of expenditure. Thus the primary breakdown of the accounting system is into categories such as "supply of drinking water to homes" or "care of pregnant women", rather than "pipes", "vehicles", or "drugs". Secondly, the accounting system associated with programme budgeting enables managers to monitor the relationship between the costs and the achievement of a particular objective. By recording expenditures according to the purpose for which they are to be used, managers can oversee the efficiency of a programme rather than merely the extent of its compliance with budget authorizations.

Systems of budgeting that stress the relationship of expenditures to activities may be further elaborated in order to produce estimates of the costs incurred by each management unit in meeting defined objectives. The performance of individual management units and providers of services may then be evaluated. For example, the quantity of drugs employed by each rural dispensary in the treatment of a case of malaria might be identified and analysed. Those dispensaries reporting exceptionally high or exceptionally low rates of drug use might then be followed up by supervisors. The effective use of budgeting and cost-accounting in order to strengthen management requires the collection of information about the activities as well as the finances of the sector. In addition, management applications require that information on health activities be collected and assigned to organizational units which have sufficient autonomy and authority to be held accountable for both their performance and for their use of resources.

Improving financial accounting

Conventional governmental accounting systems provide a basis for budgeting and help to ensure that expenditures have been sanctioned, and to guard against the unauthorized use of funds. These purposes may be met by simply recording all spending under the headings used in the government's budget document. However, in order to determine the cost of providing a

FIRE STORES STORES

particular product or service, additional information is needed. Not only must outlays be associated with programme outputs but expenditures must be adjusted to ensure that the quantity of resources used is distinguished from quantities purchased. Resources that are totally consumed during the period in which they were purchased (the services of a health worker, for example) may be charged directly to costs. However, those expenditures which are expected to contribute to output over several reporting periods (such as a hospital bed, a vehicle or a stock of drugs) should be recorded separately and distributed over the activities or periods for which they are inputs. In addition, the replenishment of stocks and the withdrawal of goods from inventories for use should be distinguished. The accounts should identify the purposes for which items were withdrawn, as well.

A well-conceived accounting system will permit managers to determine the full cost of carrying out an activity or meeting an objective. They then may compare the costs experienced over time, at various locations and under different supervisors, in order to identify both good and bad management of resources.

Coordinating revenues and expenditures

The current economic crisis has demonstrated that reliance on periodic appropriations from governments for all or even a substantial part of the funding on health programmes places the sector at the mercies of events

盟

45

i);

beyond its, and even national government's, control. Thus there is a need to plan ways in which funds are obtained which, as far as possible, avoid these risks.

In planning how funds are to be obtained, several principles and distinctions need to be kept in mind. First, interruptions in the flow of funds to an agency can have a catastrophic impact not only on its operations over the short term but also on its capacity to perform in the future. The lack of funds will normally be accommodated by neglecting those expenditures that may be delayed with the smallest legal or political consequences. Formal legal obligations to repay foreign and private domestic debts, or to pay staff salaries. are likely to be honoured, while relatively inconspicuous failures to spend in order to service vehicles or maintain buildings are likely to be used to reduce cash requirements. A deferral of maintenance and a reduction in service quality are the most common responses to financial short-falls.

Authorities in hospitals and clinics delay the maintenance of buildings and equipment in order to save money. Water supply authorities often reduce the use of the chemicals required to clarify and disinfect drinking water while continuing to pay salaries and electricity charges so that the production and distribution of water is not interrupted. Most agencies continue to provide services although the service may no longer be effective or safe.

In the long term such efforts to cope with financial problems are likely to destroy public confidence in a service,

to erode the morale of staff, and to undermine the traditions of health care institutions. The deterioration of facilities and reductions in service quality are eventually recognized by the clients and customers. Perhaps more seriously, the professional staff of the service or agency typically suffer a loss of professional confidence and self-esteem as service quality and customer satisfaction deteriorate. This loss often results in a permanent reduction of professionalism and commitment to technical standards. Moreover, in those instances in which salaries cannot be paid or long periods elapse without adjustments in salaries to compensate for increases in living costs, trained staff are likely to leave the service, thereby wasting earlier investments in their training.

In many of the sub-sectors related to health, the costs of inadequate operation and maintenance can be very high. For example, the failure to lubricate the principal bearings on a large water pump may result in irreparable damage to the shaft of the pump; thus, a saving of less than a dollar may eventually force the agency to replace an item of equipment costing tens of thousands of dollars. Similarly, a failure to paint buildings or to repair leaking roofs may irreparably damage the structures.

In a long-term financial plan sources of funds, their likely yield and their adequacy for meeting the future needs of a programme should be identified. As indicated above, a reliance on periodic appropriations from legislative bodies is an unsatisfactory method of meeting these needs. Competing public

sector priorities and short-term financial and economic crises are likely to result, from time to time, in inadequate allocations to the health sector. These reductions may not be so much the reflection of the priorities of governments as the legal and diplomatic options available to them. For example, in the present debt crisis, countries would lose their access to international credit if they defaulted on their external debts. While reducing the scale of a domestic programme may be very painful, becoming isolated from foreign suppliers would be catastrophic. especially for smaller countries dependent on imports and exports to meet their many needs.

The methods used to secure funds should be coordinated with the requirements for financing. Often governments have found it more expedient to have services, which should be their direct responsibilities, paid for by foreign or domestic lenders, and to pay their external debt obligations instead. The authorities of water supply services have negotiated agreements with governments whereby the cost of providing free services to the poor through public standpipes is financed by a one-time contribution to the capital development programmes of water authorities. The servicing of the debt arising from the contribution then becomes the direct responsibility of governments rather than an obligation of the water authorities, which has to be met from periodic appropriations from the government. Whenever a practical justification can be found, health agencies should minimize their obligations to repay long-term debts.

.

A second measure for managing the financial requirements of an agency is to link revenue generation to the delivery of services. Even countries which are not faced with the problems of external debt or a sluggish domestic economy may face problems in paying for public services because inflation and increasing demand outstrip the yield of the taxes or fees dedicated to financing the service. Local governments face this problem most frequently; often municipalities are unable to pay for conservancy services, solid waste management and sewage disposal with the revenue generated by taxes and fees dedicated to financing these activities. Because higher levels of government establish tax rates and often reassess the value of properties on which tax liabilities are calculated as well, local authorities have little control over the vield of their tax system. As a result, revenues may remain constant as costs rise due to inflation, creating a financial crisis.

Where the revision of fees must be approved by the political authorities, similar problems confront authorities which provide services on a fee basis. For example, attempts to finance water and sewerage services from fees frequently have been frustrated by governments' refusal to approve rate increases in order to meet the cost of inflation. Several countries have authorized the periodic recalculations of fees to reflect changes in the prices of inputs.

A good financial plan seeks to meet the need for additional funds by relying on revenue measures that may be expected to expand in parallel with a

and the second secon

programme's expenditures. The general sales tax and the value added tax, for example, produce revenues which keep pace roughly with both inflation and the growth in incomes, since these taxes are calculated on the basis of the market values of goods and services. A social security contribution related to earnings keeps pace with the growth of earnings. This group of taxes and contributions is therefore ideal for financing services the demand for which is likely to grow with incomes. User charges are attractive ways of financing services where the cost of providing the service depends very much on the volume of services demanded. On the other hand, services whose costs are almost unrelated to the rate of use may be paid for with taxes or fees that provide stable amounts of revenue. Sewerage provides an example. The bulk of the cost of waterborne sewerage is a consequence of the initial capital investment. This investment is usually financed on terms and at prices established for the entire life of the system. Financing these costs from a betterment levy or addition to the real property tax is therefore practical since the financial requirements are known with considerable precision decades in advance. The additional costs of repairs, energy, labour and chemicals are relatively modest. These recurrent operating costs can be met by a surcharge on the water charge or from a separate assessment.

ECONUMINE SUPPORTING

Several important distinctions need to be borne in mind in examining the costs that must be financed and how these costs might best be met. First, are the costs fixed for long periods of time? Costs may not vary where they were incurred in constructing facilities,

training staff or developing a programme. If the annual outlay is then known with confidence many years in advance, the means of finance only needs to be reliable and does not need to expand with inflation or economic growth. The general property tax is a good example of such a revenue device.

Second, are the costs of the programme likely to increase in a systematic manner over time? If the answer is yes, then a financing mechanism that grows for similar underlying reasons is desirable. For example, a programme such as for rural health care has costs which are likely to expand roughly in step with inflation and the growth of population. This pattern of costs reflects the dominance of salaries, drugs and supplies in the overall costs of dispensaries. Therefore, the sub-sector should have a financing mechanism that is responsive to these underlying forces. A tax on incomes or the value of production would meet this requirement. Third, are costs closely linked to choices made by households? The quantity of water used varies widely among households, by season and by income class. Charges based on the volume demanded provide a means to coordinate revenues and costs.

Conclusions

The planners' task is essentially to develop options for political decision-making. Deciding on which mechanisms will be used to finance the health plan is inevitably a political process. Many of the issues are highly sensitive and will require discussion between Ministers of Health, health-related sectors and those responsible for national planning and finance, with representatives of employers and employees and other organizations, and will finally need the approval of the President or Cabinet as a whole. Ministers need to encourage their planning staffs to engage in creative thinking, even if some of the options presented prove to be politically unacceptable.

Good financial planning and management are essential to the implemention of national "health-for-all" policies and strategies. The total financial requirements implied by the strategies have to be compared with the expected flow of resources. Mechanisms for adjusting, either the programmes to be implemented or the financing of the plan, are complex. A continuous dialogue is therefore required between staff at many levels of the health system, socioeconomic development planners, the community, the private sector and nongovernmental ----organizations. The planning process is an essential tool in policy-making and in realizing political commitments to

shift to a more equitable allocation, distribution and utilization of health resources.

The effectiveness of the planning process depends on the extent to which it results in a more efficient and equitable use of resources. Decisions are determined by various interacting factors such as internal political commitment, external influences, nongovernmental initiatives, and government planning processes. Throughout the various stages from policy formulation to the practical implementation of a designed plan, political commitment needs to be steadily reaffirmed.

Planners and managers have to provide relevant information and propose options for political decision-making so as to make the government's health-for-all policy a coherent whole, reflecting the critical social value of equity. A close relationship between planners and decision-makers, and a clear understanding and recognition of their respective roles, is a prerequisite for successful planning.

#

Chapter III

з

9

Mobilizing resources

Background

Chapter II has stressed that financial master planning and improved financial management can reduce the vulnerability of a nation's health plan to macro-economic problems and to shifts in political priorities. Increases in coverage, and the maintenance and improvement of content and quality, however, will in many cases, require that additional resources be committed to health programmes. But the amounts of money that can be allocated by governments to health programmes will depend not only on health needs but on the competing priorities of each country, as well as its economic resources. Even the developed countries, faced with escalating costs of health care, the aging of their population and the questionable value and effectiveness of some technologies, are reassessing their strategies to provide adequate health care to all.

This chapter presents and evaluates the options that are available to governments in order to mobilize resources for health. It first outlines the issues and criteria that should be considered in choosing a financial strategy. It then reviews different

options for mobilizing resources, including community financing, discussing their potential for financing health for all.

Criteria for choosing a financing strategy

Strategies for financing health for all will reflect the various characteristics of national economies. Where the government is predominant in the production and allocation of goods and services, and where the economy is monetised, the level and composition of health sector financing are both identifiable and, in principle, easily controllable. Where, on the other hand, there are multiple agencies of government, together with important private and nongovernmental sectors. the role of the public sector is more complex. Government financing strategies will affect not only publicly-provided services, but also those of the private sector and nongovernmental organizations. Harmonization among the different agencies contributing to health for all will necessitate the following criteria being considered not only for government programmes, but for each

Box 3.1 Criteria for choosing a financing strategy

Equity Adequacy Reliability Impact on Supply Impact on Demand Intersectoral Linkages Administrative Feasibility

Economic Support for National Health for All Strategies

quasi-independent health sub-sector individually. Regulation and coordination, often with the ministry of health enjoying only limited statutory authority, become essential additions to service provision.

Equity

Any strategy for mobilizing resources for health for all must be concerned with equity; this implies that policy-makers must be concerned with not only who gains from but also with who pays for health programmes. Government-financed and operated programmes affect the distribution of services and benefits among individuals. Plans for providing public services which are over ambitious cannot be fully implemented and hence part of the public may be denied access to them, which is inequitable.

The issue of equity also arises in connection with the methods used to obtain inputs for public programmes. Government revenues are obtained ultimately from individuals and households through various forms of taxation, such as taxes on: income; business; production (including the value added tax); land, buildings, vehicles, animals and personal property; fees and licenses imposed on professions and businesses; import and export duties; and excise taxes on entertainment and luxury goods.

Most taxes affect some groups more than others and therefore redistribute after-tax incomes. In considering the impact of a tax on the distribution of income, one must recognize that many taxes can be shifted from the person legally required to pay it to customers, workers, tenants of land or suppliers of raw materials. Such shifting of taxes may then result in a very different impact on the distribution of economic welfare than was originally intended.

The developing countries rely most heavily on taxes on imports and exports and on sales because these taxes are easy and inexpensive to administer. Export duties reduce prices paid to producers, regardless of who is legally responsible for making payments to governments. As prices are set by international markets, taxes reduce what will be paid to producers. Since basic, raw commodities – agricultural goods, timber, ore, fish, and so on - are the principal exports of most developing countries, the burden of financing government services falls mostly on farmers, timber harvesters, mine operators and fishermen. Taxes on imports are also an important source of government revenues in many developing countries. These taxes affect prices paid by consumers for foreign manufactured goods, and thus fall most heavily on persons who spend a large share of their incomes on these imports.

Inflated exchange rates also produce revenues for governments by, in effect taxing exports. Producers of export goods receive less local currency from their sales and importers obtain more foreign goods per unit of local currency. This relatively inconspicuous form of taxation falls especially hard on the rural poor.

Taxes on incomes and wealth (real estate or personal property) are not as readily shifted on to other persons and therefore permit governments to impose taxes in a less capricious and more equitable manner, but such taxes are

52

HARMAN BERTHER BERTHER HARMAN AND A HARMAN AND

difficult to administer in developing countries and may also be easily evaded. Social security contributions, on the other hand, which take a percentage of earnings (often subject to a maximum income beyond which the 'tax' is not levied) are easier to administer because: (a) the task of collection falls on employers, and (b) there are no complicated allowances varying between individuals which exempt the first part of income from taxation. Such taxes can, however, only be readily collected from those with regular jobs working for larger employers. In developing countries these are generally a minority of the working population, but their incomes are on average higher than those in the informal sector or subsistence agriculture. Social security contributions are widely used to finance health services in developed countries and are often retained as a source of income when the same rights to services are extended to the whole population by a 'national health service. They are also a major source of income for the health sector in some developing countries particularly in Latin America – often contributing more revenue than taxes.

In the least developed countries governments rely heavily on indirect taxes, which fall especially heavily on those who must spend a large share of their incomes on taxable goods. Thus usually the poor are affected adversely. For this reason, taxes may be an inequitable way of raising funds to finance national strategies for health for all. This is of particular importance where services are unevenly distributed. Those without access to services may be paying taxes to finance services

i.

enjoyed primarily by the urban population.

Adequacy

The second issue in considering financial strategies is whether the device for producing revenue will yield enough money to meet the needs. This issue of adequacy arises frequently in the least developed countries where those taxes which are administratively feasible often do not produce enough revenue to finance essential government services. If the taxes on goods are raised too much, the underlying activities may become unrewarding. The yield from the tax will then vanish. For example, in one West African country high taxes on exports of cocoa made production unprofitable for farmers, who then stopped cultivating it. The country lost its share of the international cocoa market.

The adequacy of revenue-raising systems also arises as the revenue needs of a health programme develop and mature. A revenue plan that is serviceable now may be inadequate a few years hence unless its methods of revenue generation produce a corresponding growth in funds. In recent years, this problem has emerged primarily because inflation has escalated costs and the yield of taxes did not grow with inflation. Social security taxes, payroll levies, excises on farm commodities, proceeds of lotteries and taxes on tobacco, alcohol and entertainments have been dedicated to financing health activities in some countries. These revenue sources differ widely in their responsiveness to such factors as inflation, population growth and social development. Some may be

885 HELLE

inadequate solutions to the provision of sustained financial support for health services.

Reliability

The revenues produced by a tax, or fee or foreign assistance, may vary considerably with economic conditions. For example, if the revenue system used to support health services is based largely on taxes on the export of a commodity used in industrial production, then its yield is likely to rise and fall with economic conditions in industrialized countries. The funds available to pay for salaries, drugs, spare parts and maintenance of existing vehicles and facilities may become severely inadequate during a recession. (This sequence of events was encountered by a number of developing countries in the early 1980s.) Reliability is likely to be a particularly serious problem if funds are obtained from general government revenues by periodic legislative appropriation.

Impact on supply

Revenue systems produce incentives which may influence the decisions and attitudes of both managers and providers of health services. The principal mechanism for transmitting incentives from the revenue system to the health care institution is the formula used in determining its budget. For example, if additional funds are provided to institutions with a heavy responsibility for specialized problems or endemic diseases, the staff is likely to be more vigilant in identifying these problems. The incentives inherent in a revenue and budgeting system may be harnessed to promote desirable health outcomes. This requires that the incentive mechanisms are fully

recognized and understood, and that the organization of and responsibility for services and for the management of resources be designed to take advantage of these effects. Even if institutions are thoroughly committed to the public interest, the incentives need to be scrutinized to ensure that they confirm and reward desirable behaviour.

Impact on demand

Revenue mechanisms also affect the use of services by the public. User charges link directly the use of services and responsibility for financing the system. This link clearly discourages the utilization of services. The magnitude of this incentive effect clearly depends upon the wealth and income of the potential user of services as well as his/her illness. If consumers are very well informed, the effect will be to focus scarce resources on the most serious problems: however, the capacity of most consumers to make such distinctions in assessing their health care requirements is particularly weak.

A large fraction of the cost of utilizing health care facilities is now borne by households in the form of travel expenses, and earnings lost while seeking care. Often several family members accompany seriously ill persons to hospitals and clinics, thereby increasing the cost of health care to the family. The magnitude of these costs depends on proximity to health care facilities, the number of referral steps that must be made to obtain drugs or competent care, season, and the age and sex of the patient.

The impact of policies on the siting of health care facilities, the range of

services offered at each installation and the referral of patients should be recognized. These factors affect both the costs of supplying services and the burden on households of utilizing the health care system.

Intersectoral linkages

Primary health care includes not only personal medical services but also such activities as health education, water supply, sanitation, nutrition and vector control. Several of these services are provided jointly with services that are not directly under the responsibility of the health sector. For example, some health education is provided concurrently with primary education, and water for drinking and washing is often supplied together with water for industrial processes, landscape gardening and fire-fighting. Health sector financing strategies should stimulate intersectoral action in favour of health, even though responsibility is not assumed for activities in which health is a joint output.

Administrative feasibility

MARSHARARAR RANK

The successful introduction and operation of any mechanism for generating revenues requires that practical methods be available for assessing the obligations of each person, enforcing payment and safeguarding the revenues collected. These requirements are most demanding in countries where the formal bureaucracy and traditions of compliance with tax laws are not highly developed. Tax compliance depends frequently on people recognizing that the burden of paying for government and for specific programmes is being fairly distributed. In addition, the public generally accepts taxes more readily if they are convinced that they are being administered conscientiously. User charges can be more acceptable than taxes based upon income, wealth or economic activity, since user charges are linked to gaining benefits from a programme. Social security contributions may be more acceptable than taxes because people know what they are paying for.

If local officials are not highly accountable to the community or carefully supervised, funds may be diverted to personal use and collections may be unevenly enforced. The administration of a revenue system is less subject to abuse if the scope and quality of services available locally depend on the funds collected. However, tax-payers and users may refuse to make payments if they suspect that others are avoiding payment, or if services are not available.

Options for financing_____strategies for health for all

More than a dozen options for financing services are now being widely considered. These may be grouped into five major categories. First, governments may pay for health care from *public revenues*. These revenues are obtained from some combination of income taxes, sales taxes, export levies, import duties, license fees, foreign assistance, and so on. Second, a variety of insurance schemes for the sharing of risks are possible. Compulsory and voluntary shares are discussed separately. Third, schemes of community financing may be developed. Fourth, consumers through

Box 3.2 Options for further financing

- Attract more tax revenue, possibly from earmarked taxes;
- Attract more external cooperation;
- Introduce or extend compulsory health insurance;
- Require employers to provide defined services;
- Introduce charges or raise charges for government services;
- Encourage fund-raising by nongovernmental organizations;
- Stimulate community financing and voluntary health insurance;
- Find savings from a more efficient use of resources;
- Reorient priorities within existing services, or select less costly methods of services delivery.

user charges may be required to pay for part of the cost of the health services they use. A large number of variants of these themes has been devised, each of which has distinctive economic, financial or administrative attributes. Finally, the roles of nongovernmental organizations and external cooperation are considered.

Direct government financing

Direct government financing of health activities is perhaps the most widespread approach to health financing in the developing world. Governments either provide periodic allocations from general government revenues or assign the proceeds of a designated tax to the health sector or both.

Public revenues are obtained from various sources and then generally are added together. Thus the source of financing for a particular public programme cannot be identified. However, in some cases, governments dedicate the proceeds of a particular tax instrument to the health sector. A dedicated tax provides the sector with a relatively reliable source of income whose likely yield can be estimated several years in advance. Governments (especially in Latin America) have chosen to allocate the proceeds of taxes on tobacco, alcohol, gambling and the like to the health sector. These taxes have produced revenues that expand with incomes and inflation but generally they have not produced large amounts of money. In several countries in the Americas and in Asia, lotteries have been organized to benefit social welfare programmes such as health care. slum upgrading, and primary education. Dedicated taxes should be evaluated on the basis not only of their contribution to the budget of primary health care, but also for their impact on the economy and their administrative costs.

Direct government financing of health activities alone has been inadequate in many countries. While in theory one might be able to develop a system of taxation that would provide sufficient funds to meet public needs, few

56



ÿ

5

BERRY MARCHINE

57

wounding resources

developing countries have the administrative apparatus to make such a system function. Experts estimate that as much as 70 per cent of the economic activity of some South Asian countries is not recorded or recognized for tax purposes. In addition, as noted above, governments of less developed countries rely heavily on indirect taxes which adversely affect the poor who must spend a large share of income on taxable goods.

Health insurance

A variety of insurance mechanisms can be used to help finance the health services rendered to individuals and families. These entail collecting funds directly from potential users of the health care system, either to pay the providers for their services or to reimburse users in full or in part, for payments made to providers.

The advantage of insurance is that it converts unpredictable future expenses into payments which can be budgeted for in advance. A major objection to charging the sick for the services that they require is that the need for health services is a random event. The outlays required to treat even relatively simple diseases may often exceed the cash savings of most households. In order to reduce the financial crisis that accidents or illness might impose, households pool these risks. The agreements convert large, infrequent and unpredictable expenditures into smaller, periodic payments. These payments are collected to form a pool of resources that can be drawn upon to meet the needs of a participant who encounters misfortune.

Mechanisms to share risks and to

facilitate the financing of routine services are needed regardless of the nature of the health care system. Where services are provided by public facilities, these needs are met implicitly by collection of taxes and disbursement of public funds to pay for the costs of providing care. On the other hand, privately-operated health care, financed by fees from patients, requires an explicit programme to promote risk-sharing. This programme must attract persons with a range of risks and needs if it is to succeed in its objective of pooling risks.

Membership in health insurance schemes can be compulsory or voluntary. Thease can be operated by government, by statutory agencies, by profit-making organizations (including company schemes for their own employees) or non-profit organizations, such as cooperatives or benevolent societies. Those private organizations may be tightly regulated by government or loosely regulated simply to try and ensure actuarial soundness. The -insuring agency may employ the providers of health care and own facilities, such as health centres and hospitals (the direct method), or contract with health care providers public or private (the indirect method). Nearly all developed countries which now provide the same rights to health care to the whole population went through an evolutionary stage of voluntary followed by compulsory health insurance.

Compulsory health insurance or social security

These schemes are generally financed by employers and/or employees

- The money is administratively easier to collect than an income tax;
 - The compulsory contribution varies with income and is thus preferable in terms of equity to many of the indirect taxes used in developing countries for a high proportion of revenue;
- Contributions are more readily paid than taxes because the benefit to the payer is readily visible;
- It can be used to negotiate favourable terms for the provision of services from the private sector and require cost-effective provisions such as a limited list of essential drugs;
- If the insured can be made to pay for the full costs of the services they and their dependants use, tax money has only to provide for the non-insured. Thus the introduction of a compulsory health insurance scheme can release tax money to develop and improve services for those unserved or underserved.

contributions, calculated as a percentage of pay roll. There may also be a subsidy from taxation. They may extend also to cover the self-employed on a compulsory or voluntary basis. But in developing countries it is extremely difficult, if not impossible, to collect compulsory contributions from the self-employed and difficult even in developed countries. The persons covered by the compulsory insurance may be the employees or their dependants (spouse and children) as well.

υ

The introduction of a system of collecting compulsory health insurance contributions is a major administrative operation. Several countries have covered first the larger employers, and then gradually have extended the scheme to smaller employers. Many developing countries are already collecting money from employers for provident funds or schemes covering

CONTRACTOR STATES

industrial injuries. The additional administrative cost of expanding this collection system to cover compulsory health insurance as well-is much less than where health insurance is the first or only social security scheme.

winenny neocure

Countries experience with compulsory insurance has revealed a number of common problems. In some countries funds have been used to develop separate health systems, owned and operated by one or more social security funds for insured persons and their dependants - the direct provision of services. This can lead to wasteful duplication of hospitals, lack of equity and problems of coordination as the social security system expands. Services run by social security funds can also, like government services, become inefficient, particularly due to restrictive practices developed by employee organizations.

Where service provision is contracted out (the indirect method), administrative efficiency depends on the terms of the contract and methods of paying providers or reimbursing users of services. A poorly devised and supervised system can lead to false claims by providers, incentives for excessive services and a whole series of corrupt practices on the part of both providers and users.

Social security services, whether provided by the direct or the indirect method, can also generate financial incentives for scarce trained health manpower to leave government services and concentrate on providing curative services in urban areas. However, this can be avoided if terms of service are devised by or in close coordination with ministries of health. There are examples throughout the world of social security financed schemes, using both the direct or indirect methods or some combination of the two, which provide comprehensive, good-quality care efficiently and at an acceptable cost.

Ministries of finance may be opposed to the introduction of compulsory health insurance on the grounds that it will raise prices, generate inflation and damage exports. It may be argued that employers are likely, at least in the short run, to add the cost of their contributions to the prices of the goods they produce and that employees may be induced by the payment of contributions to seek higher pay. These arguments however assume that employers are spending nothing on health care and that employees are not already buying health care in the private sector. To be offset against the

cost of compulsory health insurance are the savings to employers from reducing their existing commitments for the health of their employees and the savings to employees from purchasing in the private sector. Surveys may well need to be made to quantify those savings and convince ministries of finance that these savings really will be made.

Some countries may not yet be in a position to undertake the complex task of establishing a formal compulsory health insurance scheme. Instead they may require by law certain designated larger employers either to provide defined health services or to insure their employees in designated schemes of health insurance where they exist or can readily be established. Such legal requirements have been imposed for many years in some developing countries in the case particularly of mines and agricultural estates. The law may specify that a stated proportion of the pay roll must be spent on the health care of employees. Care is needed to see that these provisions do not conflict with national priorities, are not wasteful of scarce resources, are not socially divisive in local communities, and are coordinated with national developments.

Voluntary Insurance

People can be allowed to be voluntary contributors to a social security scheme, run by government or statutory agencies, which is compulsory for others. Alternatively, they may insure with profit or non-profit agencies or they may join a group scheme.

Groups may be formed by the employees of a firm or the membership

Box 3.4 Singapore's family savings scheme

Faced with mounting costs of the medical services, the Ministry of Health started looking at various options to change the health financing system. The problem was to keep the balance between demand pressures and supply capacity. With growing affluence and greater health consciousness, many people are wanting more and better services.

These considerations formed the underlying basis for the National Health Plan formulated by the Ministry of Health in 1983. Its key proposal, the Medisave Scheme, attempts to impose compulsory savings and restructure the present system of health care financing.

In addition to promoting individual responsibility for maintaining good health, it aims to build up financial resources so as to provide the means to pay for medical care during illness.

Compulsory savings for medical care are regularly set aside by the transfer of six per cent of earnings into a personal Medisave Account. The Medisave Account can be withdrawn to pay for hospital charges and some outpatient procedures, such as minor surgery, which are costly but do not require the patient to stay in hospital. Medisave does not cover general outpatient treatment, for which the cost in Singapore is considered affordable. Nor is it intended to cover long-term chronic illnesses, since other modes of care already provided through subsidized government programmes and by voluntary bodies play a major role.

Medisave can be used to pay for the medical expenses of immediate family members. It covers spouses, parents and children. So there is a shared family responsibility to look after the welfare of its members. It is hoped that Medisave will act also as a financial incentive for the entire family to remain fit and well together, and to avoid incurring medical expenses for individual members.

The Medisave Scheme was first implemented in all government hospitals from April 1984. In essence, Medisave serves as an additional source of personal financing for medical expenditure incurred by individual families. With this shift in public cost-sharing, government tax revenue has been freed to meet other priorities and to improve the public health services, especially as regards preventive and chronic health care.

Source: K. H. Phua. 'Singapore's family savings scheme', World Health, May 1986, pp 11-12.

9**9999988669886**498866666666666

61

of a social organization, or may be determined by the insurer on the basis of age, occupation, sex or other characteristics of the member. An employer may insure all his employees. These characteristics of the group are used to predict the volume of claims likely to be presented. Private insurers must recover the total cost of providing care, as well as the administrative costs, from participants. Thus insurance companies must guard against enrolling only those who are likely to need expensive health care. Insurance schemes typically require the patient to make an initial payment for care before applying for benefits (the "deductible"), and many also require the patient to pay a small share of the additional amount (the "co-payment"). These two devices are intended to discourage over-use of health care services.

Health insurance schemes can operate by reimbursing the consumer, or the insurer may contract and pay selected providers (the indirect method), or employ providers in facilities it owns (the direct method). Where the scheme operates simply by reimbursing the consumer for part of the cost of bills submitted by private providers, claims can become costly. Some insurance programmes have therefore set standard rates for common procedures, and have defined a limited number of "services" for which payment will be made. These moves are intended to control the claims against the insurance fund. The principal criticism of health insurance schemes is that by distributing the risks and costs of care over large numbers of people, the schemes dilute the financial incentives to use health services sparingly and to resist high prices for services. A further criticism is that

individuals with poor health records and thus major health care needs are unable to obtain insurance.

Schemes which offer comprehensive care for a fixed annual charge expect to profit by using preventive services and_ early detection and treatment in order to reduce the need for more costly institutional care. Thus they expect to be able to offer a wider range of health services at a lower cost. If a number of organizations compete among themselves for members, then the one that is most successful in minimizing the costs of providing acceptable care and thus can survive on the smallest fee, is expected to flourish. Higher cost competitors are expected to be driven from the market. Competition for customers is expected to impose discipline and accountability on the provider organizations. However, for competition to continue to impose this discipline, it must be possible for new organizations to continue to enter the market easily. But again there is the problem that persons with major health needs are likely to be denied admission into the programme.

In developing countries with little practical experience with the concepts and structures of personal insurance, participants have tended to withdraw from a scheme if their benefits did not equal their payments over a fairly short period of time. This has left schemes with participants who have greater than average needs and has forced up the cost of premiums. This problem is likely to be greater if insurance is confined to really major expenses such as a hospital stay of over a week. While such insurance may be relatively cheap,

dest de la comp

63

Box 3.5 Community financing in Senegal

Community financing has truly persuaded people to take much more responsibility for their own health. To give just one example, the funds raised by the system of self-management amounted in 1983-84 to 80 per cent of the budgetary appropriations, excluding staff, of Senegal's Ministry of Public Health.

At the national level, receipts during the first year of the project amounted to 303 million CFA francs or US\$ 800 000. Community participation was responsible for the building of 115 maternity units and 49 health posts. In effect, the people and state of Senegal doubled their health care purchasing power.

It is the people themselves who manage this system, which is financed by patients' contributions. It covers all hospitals-health centres (department level), health posts (rural community level) and health huts (village level). The charge is 100 CFA francs per adult (\$ 0.26) or 50 CFA francs per child for treatment in hospitals and health centres; 50 or 25 CFA francs for adults or children attending health posts.

These receipts are administered by health committees which include representatives of every health hut in the village, so they have to learn the intricacies of management; 60 per cent of the receipts are used to buy drugs, 30 per cent for staff expenses (female birth attendant, community health worker) and ten per cent for operational expenses.

The village came into its own again, so to speak, around the health hut. The council of elders, the rural council, the mothers' committees and the health committee all meet under the palaver tree to discuss their problems of health, hygiene and cleanliness. They consider how to replace their stock of drugs, the sale of tickets and collection of receipts, and how to pay, in money or in kind, the community health worker whom they themselves have chosen.

Human resources, community labour and other resources have never been lacking whenever the need arose. Community participation finds expression in a thousand and one ways, including the building of health huts, and payment for care and essential drugs.

Some 90 per cent of the villagers were making use of the health huts. Consequently less time, money and effort were wasted than before, when the villagers had to go to a distant health post.

Source: Sène, P. M., Community financing in Senegal, World Health, May 1986, pp 4-6.

it is likely to be difficult to attract and keep policy-holders.

Community financing

The involvement of individuals and the community, in both developing and developed countries, has been increasingly seen as an important means of decentralizing decision-making in health matters, promoting life-styles conducive to health, and improving effectiveness of health care. A variety of mechanisms have been developed often reinforcing traditional systems. Village development associations, welfare groups, cooperatives and district health committees have been established in order to provide services that communities demand but cannot obtain from governments.

The emphasis of community support in most developing countries has been on providing resources, either financial or material and human, for the establishment or improvement of health and sanitation infrastructure, for example, health facilities, wells, latrines, drug cooperatives, feeding centres - or for payment of parts of health service operations. Some community health programmes - rural water supply, "clean-up campaigns", immunization, nutrition, diarrhoeal diseases and malaria control - have been financed from funds contributed by the beneficiaries.

Spontaneous community efforts to develop or sustain health programmes indicate people's willingness and ability to pay for better services. People who invest substantial energy and resources in creating a local organization and then contribute the funds needed to sustain its activities have demonstrated their enthusiasm. Communities are likely to develop programmes spontaneously only if they recognize needs that are not covered by government programmes and have the necessary resources.

Where the government requires the payment of user fees for the services it provides, some communities have established funds into which village families make regular contributions and the fund pays directly to the health services the cost of those user charges which would otherwise be levied on contributors to the fund. In Thailand, health cards are sold by village volunteers which give defined entitlements to exemption from user charges.

Sometimes community efforts have been motivated by discontent with the reliability or quality of services provided such as availability of drugs in local health centres. Community pharmacies staffed by volunteers have been established which collect funds from beneficiaries and maintain a stock of essential drugs. Alternatively, the government provides an initial stock which is kept replenished out of the revenue from sales (a 'revolving drug fund'). These pharmacies are often operated on a commercial basis; drugs are priced slightly above cost in order to recover the cost of transport and of losses from pilferage.

Communities may obtain the resources needed to sustain programmes by a very wide range of methods. Many projects have imposed fees and charges for the services and commodities they provide. These fees may be paid at the

64

me services are rendered, or may be ollected in advance. Fees for simple "alth activities and for the services of aditional birth attendants may be ollected in the form of produce, dairy oducts or chickens, for example. Cash yments are generally expected by odern health workers. Providers of re throughout the world accept Iditional compensation from patients r special services.

ŝ

formal insurance schemes have also en developed to collect subscriptions om persons enrolled in a health ogramme or to impose taxes on oduction activities. The subscriptions e based either on the incomes of the ember or more commonly on the imber of family members enrolled. oduction-based schemes levy charges ughly in proportion to the income of e beneficiary. Because the ntribution is demanded at the time at people are receiving cash for their ops, collection is greatly simplified. arketing cooperatives, such as a coffee owers group in Colombia and a milk operative in India, have withheld a are of each member's sales as a tax support health programmes for embers. Cooperative rice-polishing ills in Bangladesh and the Republic Korea have imposed a surcharge on [•] processing fee, based on the volume paddy submitted.

ommunity health projects also use outs contributed by beneficiaries. Tese inputs include labour and local nstruction materials as well as cash. -kind contributions have been used ost frequently and successfully in nstructing facilities. Some projects ve employed volunteers to carry out tine activities, but most of these

MARKARASCALL CONTRACTOR

projects have encountered difficulties in sustaining the community's interest.

NIUDINIZING ICOULCES

Community financing of health activities requires community organization. The most serious problems have arisen in trying to sustain contributions to pay for the recurrent costs of programmes and especially for the provision of sufficient logistic support. People frequently have been unwilling to continue to pay for programmes from which they were not benefitting at the time. Recovery of costs has been most successful where charges were clearly related to benefits received. For example, user charges for water supply and waste disposal are widely accepted.

Decentralization of responsibility and allocation of the necessary resources (especially logistics) is a prerequisite for successful sustained community involvement. While the trend is encouraging in many countries, resources still remain largely centralized. With the current budgetary constraints, there appears to be an even greater tendency than before to decentralize decision-making on priorities and allocation of resources. Because of the critical importance of administrative decentralization, some countries have created new local or district administrative or legislative bodies, such as community councils or village community-reliance institutions with well-defined responsibilities and resources.

Conflict within community organizations can be minimized by keeping organizations small, restricting membership to persons with similar interests, and pursuing narrowly defined

objectives. Large organizations are more susceptible to conflict than small ones because the larger number of participants increases the diversity of interests and hence the likelihood of disagreement.

Governments can help local groups to organize workable community financing schemes. Assistance can be offered in setting up the organization and in establishing its system of records and its practices in handling money. Local organizations can be given authority to hold property, and audit systems can be provided to ensure that funds are used properly. In addition, women's clubs, organizations of school children, cooperatives and the like, can be encouraged in order to give community members opportunities to develop skills in operating local organizations.

User charges

A fifth way of financing health care is by charging patients. These charges are levied on those who use services and take a variety of forms. Some form of user charges for health care is now applied in every country in Western Europe, either under compulsory health insurance or a national health service.

Water is often supplied by a public distribution system through metered private connections. The customer is then charged on the basis of the quantity consumed and a schedule of rates. These rates are often designed to permit families to obtain at nominal cost a sufficient quantity to meet basic health needs. Much higher rates are imposed for water used for nonessential purposes such as irrigating a garden or filling a swimming pool. In this way the

charges for water supply are made to vary with the importance of use and roughly in proportion to the ability of families to pay for the service. Water charges may also be levied on the basis of the number of taps or water-using devices in the household, or by the value of the property as determined for rates or property taxes. Property owners are frequently made to pay for the cost of pipelines and production facilities for water supplies as a part of the price of a building plot.

Fees for medical services are even mon diverse. First, the definition of the iten for which a charge is to be levied varie widely. A fee may be required for an encounter with the health care provide an episode of illness or a fixed number of contacts with the health care system A single encounter may be broken down into a number of items representing tests or procedures or goods supplies such as drugs. The rates applied to each of these may also vary widely. A uniform price may be charge for all patients or for all except the poor, children or other categories who are exempt, or a sliding scale of rates may be applied which imposes lower fees on persons of lesser means. Prices may be adjusted by the provider of health care or may be based upon a certificate of low-income from a local official.

The choice of chargeable items is crucial in determining how user charge affect the provision of health services. At one extreme, patients might be required to pay precisely the cost of services rendered. The cost of supplyin a unit of the services of each health worker and each facility, and the cost of supplies and drugs might be

multiplied by the number of units of each item that a patient used. This makes the patient keen to ensure that more services are not provided than are really needed. It also makes the provider aware that his or her actions are throwing costs on to the patient. A set charge per visit would alternatively reward the health system for reducing the cost per visit, but would at the same time encourage further visits. Similarly, a system that charges for an episode of illness or broad activity, such as the management of a normal pregnancy would shift additional responsibility on to the provider. If fees are based upon the procedures offered or the diagnosis, providers may benefit from excessive treatment or reporting multiple diagnoses or complaints.

User charges have the advantage of providing a link between financial responsibility and the provision of services. This link has generally enhanced willingness to contribute to the cost of health programmes and has encouraged both consumers and providers to be cost conscious. In addition, user charges help to control the use of health services by imposing financial disincentives on the consumer. Whether this reduction in health care utilization results in a better use of health resources requires careful evaluation.

The administration of user charges presents serious challenges. The majority of studies in developing countries show that the largest reduction in the use of services as a result of charges occurs among the poor. In this respect charges are in conflict with the aim of health for all, of equity, unless effective safeguards can be

8698888888888888889999

and the second statement of the second se

developed to protect the poor by exemption. But if charging for services where they are provided enables money to be released to develop services for those not currently provided with them. then equity is increased. The major problems are to develop effective systems for exempting the poor and to limit the extent to which bad debts occur which are hard to collect. Checking on people's incomes when they seek health care is very difficult in societies where there is substantial illiteracy, people do not keep records of their income and much of it is in kind. Systems of requiring certificates from local officials may work reasonably well in rural areas but are much harder to apply in an urban situation. Moreover, the need for health care services can be sudden and urgent with no time to be spent on bureaucratic procedures. Where need is urgent, health care systems are obliged to provide care first and seek to recover any charges afterwards. Hence the costly process of trying to collect debts.

Nongovernmental organizations

Relatively unbound by the legislative and policy framework of governments, nongovernmental organizations (NGOs) have the flexibility to experiment with innovative and alternative approaches to solving health problems, often achieving cost-effective breakthroughs which could provide new models for national planning. Nongovernmental organizations in many countries have made significant contributions to human resources development in health and are often able to mobilize critical financial and material resources which can be crucial

Box 3.6 Protecting the poor in Seoul

The Medical Assistance Scheme is a programme for the needy, supported 80 per cent by the central government and 20 per cent by the local authorities, except in Seoul where the cost is shared equally. The Ministry of Health and Social Affairs sets the standards of eligibility for benefits, and the local government is responsible for assessing cases and for day-to-day management.

There are two categories of beneficiary, the indigent and the low-income group. The indigent are those aged over 65, the disabled, children under 18 without parents or with parents over 60, and people residing at welfare facilities; they receive yellow identity cards. The lower-income group consists of those with an average income less than a certain amount and subsistence farmers. The level was set at 40,000 WON (about US\$ 50) per person per month in 1985. People in this group are provided with green identity cards. The selection of eligible persons is made once a year.

The yellow card holders receive medical care free of charge, whilst the green card holders have to pay 20 per cent of the in-patient fees (except in Seoul where they pay 50 per cent). The Scheme enables them to receive primary health care at private clinics designated by the Ministry of Health and Social Affairs or from health centres and community health practitioners, and they are referred to secondary and tertiary hospitals if necessary. About half of the medical facilities nationwide are designated for this purpose.

Though there have been occasional delays in payment owing to the shortage of local government funds, the Scheme has been operating smoothly to the benefit of 3.3 million people – 600,000 indigents and 2.7 million low-income people. This is eight per cent of the total population.

Source: Moon, O.R., Towards equity in health care, World Health, May 1986, p 20.

to many developing countries. Often dealing with specific health problems or areas, they can offer a possibility for concrete technical support to government health strategies. Some national NGOs have been able to mobilize substantial private, local and

international financial resources as we' as human resources (volunteers) whic otherwise would not be available for health programmes. Family planning associations, organized women's group and associations for disabled persons are some illustrative examples of the hidden potential for mobilizing resources in the community. Often a small government subsidy can enable the setting up of a voluntary movement which can more flexibly mobilize resources for specific deprived groups or a health problem which otherwise may not receive the attention it deserves.

\$

While the overall contribution of NGOs in financial terms may be small in most cases, their potential for mobilizing people and strengthening their self-reliance should not be overlooked. They can particularly support primary health care efforts by taking up innovative actions, especially at community level.

External cooperation

Of the approximately US\$ 35 000 million annually available for external financing of general development, between eight and ten per cent is devoted to health financing. External financing is generated mostly through development-oriented institutions such as bilateral agencies, multilateral organizations and banks, and NGOs (for simplicity and uniformity these are referred to as "donors"). Financial cooperation is generally channelled to a central authority in the recipient country, such as the ministry of finance or central planning organization, which coordinates foreign aid. In some cases, funds may be routed directly to particular ministries, agencies or NGOs. Conventions differ among countries on the extent to which external cooperation received from government services is shown in government accounts.

Considering the large volume of external funds involved and the multiplicity of donor agencies, there is a need for coordination between donors and recipients to develop realistic and effective health sector strategies at the country level, and for both to abide by these strategies. Where donor efforts have not been well-coordinated recipient governments have been confronted by contradictory requirements and approaches by donors. Donors have, in some cases, disagreed among themselves on necessary sector reforms. However, there has been some movement towards consensus positions on issues such as: the importance of sector planning; cost recovery; institutional and human resource development; community participation and the role of women; sanitation; and health education.

Where donors sponsor similar projects in a recipient country, coordination of training efforts and strengthening of national capacities by collaborative institutional development are essential to secure better returns on investments. Countries with well-formulated proposals rooted in a clear, national strategy, and with plans scrupulously in line with declared national priorities, are more likely to be successful in attracting external financing.

Because external financing is more readily available for capital investments than for recurrent costs, recipient countries which have already received assistance for capital expenditures may later find themselves facing substantial recurrent cost obligations. This becomes a critical problem when current budgets are suddenly reduced due to economic stringency.

69

The mobilization of external resources requires a working knowledge and adaptation of the development financing procedures which have evolved over a period of more than 30 years. Donor agencies note that the health demand has by no means yet reached its eventual potential. However, the health sector must organize itself effectively for the task of mobilizing external financial resources to realize the goals of health for all.

Conclusions

In so far as tax revenues for the health sector are likely to be tightly limited in all countries irrespective of their level of socioeconomic development, possible sources of financing include charges to certain categories of users or for particular services, compulsory health insurance (where it has not already been extensively developed), formal voluntary insurance and informal local insurance, revolving funds, and voluntary contributions in cash and kind. Local financing as a part of community involvement can effect change, and decentralization can strengthen managerial capacity at the local level. All these financing mechanisms to provide economic support to the national strategies for health for all should be considered and decisions taken on the basis of the criteria that has been suggested above. The search for further financing is, as shown in Chapter II, an integral part of the construction of a financial master plan.

What is clear is the overwhelming importance of examining every possible new source of internal financing and finding more cost-effective ways of using resources to achieve health-for-all goals.

hapter IV

ž

Making better use of resources

3ackground

Primary health care addresses the main health problems in the community and iccords priority to those health nterventions that are most effective, cientifically sound, affordable and esponsive to the felt needs. Considerable energy has been devoted over the past decade to refining those nealth technologies which are applicable it the family and community level either by people themselves or by health personnel with limited yet idequate training. An essential haracteristic of a health system based on primary health care is that the system should be organized in such a vay that it provides the essential elements of primary health care to the ommunity level. More complex health problems and specialized health ervices should be dealt with at ntermediate and central levels. Drganized in this fashion, health esources can be effectively and productively used. Furthermore, a rimary health care strategy requires oordinated efforts with other relevant ectors, such as agriculture, education, nd environmental health. Thus, dverse impacts on development programmes can be mitigated; and ctivities can be synergistic, thereby ncreasing overall cost-effectiveness.

Aaking better use of resources thus is n implicit objective of primary health are. This is equally true for developing s well as developed countries. Health are systems in the latter have become omplex and costly enterprises. The flectiveness of their impressive health ufrastructure and high technologies for neeting the emerging health problems f their people is debatable.

Cost-containment policies have begun to surface in many countries, reflected in health promotion strategies which emphasize and support individual self-reliance.

There is general concern in all countries that the available health resources are not being used most effectively or efficiently. Comprehensive, scientifically sound studies of the cost-effectiveness of resource use in the health sector are needed. A large share of health resources (up to fifty per cent in some countries) are wasted because of poor management practices, use of inappropriate technologies, excessive and unnecessary routine care norms or other factors and forces, not all of which are amenable to rapid change.

The issues are not only managerial but also ethical, political and attitudinal. However, the magnitude of the problem demands that urgent attention be devoted to this area in order to increase the impact of available resources on health outcomes.

This chapter first describes some of the major causes of resource waste and suggests how better control of the resources devoted to health care can be established. Proposals are then made for improving efficiency in the utilization of human resources, technologies and disease control strategies. The third section is devoted to cost-containment. The final section focuses on policy issues and responsibilities in this area.

Improving accountability

The major causes of waste of health resources are their misappropriation, underemployment and deterioration.

Misappropriation

The magnitude of waste due to the diversion of health inputs is difficult to assess. However, managers know in general terms how important this waste is, especially in the case of drugs. In some countries, the loss of imported drugs can be as much as 70 per cent from the time they are delivered to the country to the time they are supplied to the peripheral health posts for distribution. Inadequate transport systems cripple both the supervision and the distribution of drugs and supplies. Lack of regular provision of drugs and other supplies not only undermines the functions of services, but also discourages the utilization of peripheral health facilities, such as dispensaries and health posts. Relatively minor gaps in supplies can have a substantial negative effect on theproductivity of the entire health care system.

The delivery of health services requires inputs which are valuable for other uses: they can be used for other purposes or readily sold in grey or black markets. For example, motor vehicles provided to the health sector can be diverted to personal use by health workers or unauthorized use in the transport of goods and passengers. While these alternative uses can be valuable to the community or individual they nonetheless diminish the supply of transport available to the health sector. Stocks of spare parts, lubricants and fuels for motor vehicle can be stolen or sold for private gain. In remote rural areas where these goo are particularly scarce and administrative supervision is often weak, these items can be easily misappropriated. Drugs, bandages an dressings supplied by the government for free distribution or at nominal co can be diverted, either by health workers or by patients to private markets, where they are sold to the public or used in licit or illicit private practice. Where unwarranted resale of goods works well, its major effect is to enrich those who divert and resell the goods. However, control over the prescription and quality of drugs typically deteriorates when unqualified and unregulated retailers assume responsibility for the distribution and the value of drugs is partially wasted : well.

Underemployment

Despite the scarcity of health care availability in much of the world, ma health workers and health facilities (including village health posts) are seriously underutilized. In many instances in rural areas, health faciliti operate for only two or three hours a day, serving a fifth or less of those people needing care. Part of the explanation for underutilization of peripheral facilities is that too few people live within a reasonable distan of a health facility to fully occupy the facility and its staff. Where transport not available (or perhaps not affordable), rural health posts and dispensaries can serve only those who live within a reasonable walking distance. However, self-referral to higher level facilities (district hospital

typically) is also a major explanation for underuse of local health facilities. This by-passing of local facilities is often a consequence of the public's lack of confidence in the quality of services available locally. Other reasons include lack of supplies and equipment; harsh and uncaring attitudes from health care providers; high costs or imposition of unofficial "surcharges for services"; decentralized decision-making; poor managerial capacity at the local level; and inadequate support and supervision services.

Deterioration of resources

ŝ

Delays and/or bottlenecks from procurement to distribution, may cause health workers to receive unusable medical supplies and equipment, including out-of-date drugs. Because of the lack of an adequate operating budget, health facilities can also suffer from insufficient maintenance and deficiencies in hygiene as well as substandard storage of materials. Local and intermediate levels of health care systems often lack the expertise required for proper maintenance and repair of equipment and vehicles which eventually become unfit for use.

Formal accounting and management information systems can be implemented to enable supervisors to detect waste and misuse. Such systems are designed to record the quantities of goods made available and the activities undertaken by health facility or health worker. These systems must be integrated in order to cross-check the accuracy and credibility of reports and allow rapid analysis in order to identify anomalies for follow-up by managers. The development of inexpensive personal computers has made

computerization of these tasks a practical option. Computerized accounting and cost-control have had a major impact on the management of hospitals, drugs supplies, and district health programmes. Formal accounting and management information systems are also essential tools in the assignment of health manpower and the distribution of facilities and medical supplies.

Another means by which resource accountability can be improved is through the community. Community involvement in health services management can take various forms, from the simplest at the primary health care level to the most sophisticated (e.g. consumer unions; representation of local people in management committees) close direct supervision of providers of care by the formal bureaucracy is difficult, irregular and costly. However, communities through their representatives can successfully oversee some aspects of the operation of health facilities such as hours of service, use of vehicles, availability ofdrugs, management of funds collected or allocated, maintenance of facilities, and so on. For such management to succeed, the authority of the group responsible must be recognized by the community and respected by higher levels of the health administration. Prompt, effective follow-up of complaints from the community by higher authorities is essential.

The third point at which accountability can be improved is at the level of the health worker. Based on a clear definition of his or her professional responsibilities, the health worker has to be both accountable to the

73

Box 4.1 Kenya – a new drug management system for rural health facilities

Efforts to strengthen the drug distribution system in Kenya illustrate how improvements in organization, record-keeping and management can improve the quality and reduce the cost of health care.

Only a few years ago, many rural health facilities (RHFs) in Kenva health centres and dispensaries - were prevented from giving a proper service by severe shortages of basic, life-saving drugs. Sick patients might be forced to travel 100 km or even more to the bigger hospitals, suffering both physical and financial hardship.

In 1980, the Ministry of Health selected a list of essential drugs for the RHFs and tested a pilot drug supply programme in two regions - one at the coast and one in the highlands. In 1985, the rural drug distribution programme functioned well all over Kenya serving all rural inhabitants within a much shorter distance of their homes.

To save money, the quantities of drugs supplied must correspond as closely as possible to the needs of the RHFs. Standard drug rations have been worked out for a workload of 3000 new cases at a health centre and 2000 new cases at a dispensary. These quantities form the basis of the drug ration kits.

Ration kits

To avoid wastage and loss, the standard drug rations are packed at the factory into strong, sealed cardboard cartons which are opened by no one until they reach their final destination. The headquarters drug management team is responsible for dispatching the kits to the district hospitals by government-chartered lorries. The district health management team uses its own vehicles to transport the kits to RHFs, the health centres and dispensaries, in its area.

Health centres receive two kits containing a total of 38 drugs; dispensaries receive two kits containing a total of 31 drugs. The difference reflects the fact that clinical officers who are in charge of health centres, have a more advanced training than community nurses, who are in charge of dispensaries. Since the system of kits was introduced, wastage and loss have been negligible.

Control

Drug supplies for the rural programme are subject to strict controls at all stages of their journey from supplier to user. The suppliers receive (Cont.)

Box. 4.1 (Cont.)

precise instructions as to the drugs to be supplied, their packaging, labelling and quality. Random sampling is also performed by the headquarters management unit, which regularly takes out one package of each item in a kit received at the Central Medical Stores, for independent analysis organized by UNICEF.

Out-patient Register

Registers kept at every RHF record how the health worker is managing the patient. They include all the patient's particulars, the diagnosis, the treatment given and details of any possible referral. Part of the District Clinical Officer's responsibility is to check the register and discourage symptomatic treatment (treatment should be given only for a specific diagnosis and provide on-the-job training of health workers). Polypharmacy – giving more than one drug at a time – is also discouraged.

Benefits

The new drug management system has cut down the cost of supplying drugs to the entire population of Kenya to US\$ 0.29 per person, due to such factors as bulk procurement in generic form, standardized treatment schedules which reduce the number of drugs prescribed, and the virtual elimination of wastage and loss. The service given at RHFs of which there are over 900 has improved and the staff find greater satisfaction in their work.

Control of each individual drug on the essential drug list for the RHFs has become possible for the first time. Finally, the supply to the RHFs is constant. Even the problems of the rainy season can be overcome by making extra deliveries in advance.

Adapted from "Kenya: rural drug distribution programme", WHO Action Programme on Essential Drugs and Vaccines, 1985

community and report to his or her supervisor for delegated responsibility. This managerial requirement is common for all types of service delivery and at any level in the hierarchy of the system. In primary health care, this requirement is particularly important is health staff, including village health igents, have to work in areas where close relations with their supervisors is limited and irregular. Thus, much of

the responsibility for overseeing resource use and quality of services remains with the health worker. Efforts to recognize their performance, to instil pride in services and develop skills are likely to enhance professionalism and a sense of personal responsibility and accountability. Support services required to improve effectiveness in primary health care are developed in greater depth later in this chapter.
The success of all these measures will depend on the extent of delegation of responsibilities in the health management system. Appropriate delegation of responsibilities requires decentralization in decision-making, adequate managerial capacity at the local level, and a strong support and supervision system. The crucial point here is to give sufficient authority to health managers to achieve accountability for responsibilities assigned to them.

Improving efficiency

The factors which affect efficient use of resources by the health sector are many and varied. Some of these are discussed in this section.

Cost-efficient access

If access were more equitable, the efficiency of health care would generally be greater. The importance of this issue varies widely. Where people have no access to services, the first challenge is progressively to provide to the majority of people at least the essential health services which have a low cost but are effective. When coverage has increased. efforts should focus on the underprivileged and the most vulnerable risk groups. Priority-setting for health programmes and the allocation of funds and resources should be based upon judgements of political viability, financial feasibility, technical efficacy, socio-cultural acceptability, and administrative and managerial capacity.

Decisions about the location and the scale of new facilities are based

generally on the costs to the responsil agency. The costs to households of travel, food and lodging required to obtain health services or drinking-wat are usually not considered, though the may be very large in relation to incomes. If these costs are recognized in planning investments, higher levels of service are frequently found to be justified. For example, the labour cost of collecting water for domestic use in arid areas of Africa and South Asia comprise several hours a day. Improving the accessibility of supplies by increasing the number and proxim of wells would substantially improve the well-being of households, even if they were required to pay the addition costs. Public standposts in urban areas are often too few in number and inconveniently located. Increasing the number of delivery points and reducin the distance from the standpost to the place where water will be used would in many instances substantially reduce the costs to households of meeting bas needs for water. Improvements would reduce both time and transport requirements.

The geographic maldistribution of health services also imposes heavy coon consumers. Often not only patients but also other family members who accompany the patients must be transported to and maintained at a health facility. If, as is typical in rural areas of developing countries, transpo facilities are limited and public service are infrequent, then the costs of seekin care may be very large for the family. The frequent bypassing of dispensaries and rural health centres and over-use of district and regional hospitals is largely due to the fact that the cost to patients is generally lower despite

greater distances. These lower-level facilities also frequently have to refer patients upwards because of lack of skills or of drugs and supplies. Careful analyses are needed to identify the optimal range of services to be provided locally, and rigorous efforts are required to ensure that local health facilities, such as dispensaries and health centres, are supplied and staffed adequately to be able to meet a high proportion of demands.

Efficient use of human resources

The issues underlying various strategies for improving efficiency in the health manpower field are largely economic. That is, the cost of training and using a stated type of health service may be excessive because the service is furnished by personnel more expensive than necessary or because it uses patterns of work which are wasteful. In other words, the goal of efficiency in health manpower implies that the cost of providing a given service should be no higher than is necessary or reasonable for its proper performance. Hence, if a briefly trained and modestly salaried health worker can give an immunization satisfactorily, it is wasteful and inefficient to have an expensively trained and highly paid health professional do it. The use of a less expensive health worker may, indeed, require closer supervision and various other conditions for assuring that the work is properly done; and the costs of these factors must be recognized.

Even in the most affluent developed countries, there have been many reasons for governments and community leaders to search for

THE CONTRACTOR OF THE CONTRACT OF THE CONTRACT.

maximum efficiency in the use of health personnel as well as in their training. In many developing countries, insufficient human resources on the one hand and their inequitable distribution on the other can create a two-edged problem of inefficiency: poor utilization of health facilities due to lack of trained health personnel and underutilization of professional health workers due to their excess in the urban areas. In a growing number of countries in Latin America and Asia in particular, a large number of physicians in urban areas are either underemployed or unemployed. Considering the investment in their training, this is a serious wastage of resources which no country can afford.

In spite of these known deficiencies, most countries have not adopted adequate policies or measures to reduce waste in order to improve efficiency or performance. Many options are available for achieving greater efficiency in the use of human resources. These in fact form the fundamental principles of primary health care strategy.

For example, one option is to promote the role of individuals and family in preventive and health promotion measures. This requires education and information support. In a number of developed countries increasing attention is being given to health promotion aimed at improving life-styles and health practices of the individuals thus reducing health risks and ultimately the demand for health care services.

A second option is to make rational use of health personnel consistent with the functions of each level of health care system. For example, at the home and community levels use of community

с базабраата бола с на се таректарараната не ороскора се состо на состо

44.44

Box 4.2 The Egyptian dayas

In Egypt, traditional birth attendants called dayas, have for a long period of time demonstrated a great capacity for providing midwifery services among the population. In many rural places the ratio of dayas per 1000 population is 0.4-0.7 and some surveys have shown that they perform roughly between 50 and 75 deliveries per year. They have a high status in the community and their advice is highly regarded.

Therefore, in order to reduce neonatal and maternal mortality through improved antenatal and postnatal care and safer delivery practice, the Government decided to develop an important programme to make better use of dayas. This programme started in 1982 on a trial phase in four districts and now the process for extension is going on the whole very smoothly.

The objectives of this programme are:

- i) IIIII to upgrade the knowledge and skills of the dayas in midwifery techniques as well as in other areas of maternal and child health (immunization, management of diarrhoeal diseases, family planning, maternal/child nutrition, community health education);
- ii) to foster stronger and closer working relationships with the health services personnel; and
- iii) to promote a broader role of the dayas as health agents towards primary maternal care.

The following are some striking characteristics of this programme as observed in a recent assessment:

- a) The health personnel at all levels have become extremely positive, the doctors both in the Rural Health Unit (RHU) and in the hospital are supportive and motivated. This is a definite sign of success of the programme.
- b) The dayas have been eager to participate and acknowledged the need to correct bad practices. They have also given multiple examples of timely referrals for complications. They have been very willing and able to accept a wider voluntary role in health, particularly in relation to immunization, antenatal care and family planning.
- c) The RHU supervision of dayas, the motivation of the programme and the perceived improvements seem to have been greatly rewarding. Receipt of certificate and distribution of midwife kits were essential incentives. (Cont.)

Box 4.2 (Cont.)

- d) The process for developing the curriculum for the dayas training has been a flexible and participatory one. The learning objectives have deliberately been kept somewhat general in order to allow this process.
- e) The daya's training programme was not only making use of the daya and linking her with the health team, but was also a real strengthening of the supervision capacity and of the back-up capacity at the first referral level.

This last result is extremely important, since it is clear that, in the future, the number and the use of dayas will decrease with the "modernization" and the demand for midwifery and obstetric services from the rural health unit will increase. This training programme is really preparing a much more comprehensive move towards primary care with the strong support of the medical district team. However, the use of dayas is currently considered as the most cost-effective way to provide primary maternal care but also the most acceptable for the rural population.

Source: EDSTROM, K. Assessment of the expansion phase in Beheira Government. UNICEF-supported Dayas Training Programme in Egypt. 3 July 1986.

health workers and traditional health practitioners, such as traditional midwives (wherever applicable) can be promoted. This option has to be carefully costed taking into account costs of training, supervision and remuneration.

Auxiliary health workers and middle-level health professions (nurses, midwives, public health inspectors) have been used successfully at the local and intermediate level of the health care system. Yet in many others, these health workers are insufficient or inadequately prepared. Consequently, fully qualified doctors are carrying out activities which an adequately trained lower-level health worker could perform.

No universal formula for the most cost-efficient health manpower mix can

a aparappoper appoint a

be proposed. However, a health team approach, especially at the intermediate level of health care system, can be advocated. A health team can facilitate the performance of functions of all personnel in the least wasteful and most efficient manner. The roles and functions of each member of the team should be clearly outlined. The training as well as supervision requirements should also be considered. In this manner an adequate, appropriate and cost-efficient health manpower mix can evolve.

The potential gains from a more rational use of health manpower vary widely. In some countries the cost of outpatient care has been reduced by as much as a third by making maximum possible use of nurse practitioners, midwives, therapists, surgical assistants

and social assistants. Health post attendants, dispensers and family welfare workers offer the possibility of not only less costly care but much more rapid expansion of coverage, particularly in the lower-income, developing countries.

at the unclear and the main

Increasing the use of lesser trained manpower requires action by governments. Training opportunities have to be provided, and in some instances, appropriate certification or licensure have to be introduced. Career opportunities would also require expansion. Pay scales may have to be revised and grades affixed to the new professions. Regulations and laws regarding the prescription of drugs and practice of medicine may have to be reviewed to ensure that the responsibility given to new categories of manpower do not violate the law.

Attitudinal changes are also required both of the health professionals and the community to achieve a more rational use of health manpower. Opportunities to train the new health workers in situations in which the full health team is present can help to modify old roles and assignments of responsibilities. Health administrators and doctors would have to acquire respect for the capabilities of other members of the health care team and support them fully in their roles and functions. Communities also need to be informed and educated about the roles and functions of the various members of the health team. Supervision and continuing education are also critical at each level to motivate field workers to provide a continuing service of high quality and to enhance their credibility.

Cost-efficient technologies

Expensive and labour-saving equipment which requires sophisticated maintenance workers and highly controlled environments have often been purchased by countries (including developed countries) which cannot afford it. In much of the developing world the adoption of inappropriate technologies has sometimes resulted from aggressive marketing by manufacturers of equipment and from foreign economic assistance programmes that stress capital expenditures and tied procurement. Sometimes, decision-makers have also actively sought highly sophisticated facilities in part because they received their training at prestigious international universities which prided themselves on having the most advanced facilities possible. Finally, in many instances the world market for simple low-technology equipment for the health sector has been too small to attract interest among manufacturers.

-Excessive investment in hospital technology has attracted discussion throughout the world. Highly specialized facilities for preparing angiograms, computer-assisted tomography and the like, are examples of facilities which are very expensive to acquire, operate and maintain, and which may not be used frequently enough by many owners to maintain the proficiency of operators. In the developing countries equipment has been purchased which is difficult to operate and maintain because of lack of trained technicians, inadequate electric power supplies, faulty installation and high cost. Because of

adverse operating conditions and inadequate maintenance, the equipment has frequently not functioned properly and has deteriorated rapidly, thus requiring premature replacement.

The choice of technologies is usually made at high government levels. Indeed, the purchase of an individual item of equipment frequently requires approval at a high level because of the magnitude of the expenditure. Therefore, it is at this level that pragmatic decision-making is needed.

Decision-making about the choice of appropriate technologies should be part of the overall planning and management process. Its rationale is to start with the identification of priority objectives, feasible and affordable programmes, and then define functions required for staffing and equipping health facilities. Unfortunately, in many cases, decision-makers in both the developed and the developing countries have to cope with a heavy inheritance of large hospitals. It is not easy to restructure the hospital network and convert the health system towards outpatient centres. Some developing countries have changed their investment policy by substituting health centres for planned new hospitals. In many developed countries, the results of hospital planning have been to concentrate high technology in a limited number of specialized hospitals and to devote new investment to ambulatory care centres. For instance, there is a trend towards redistributing resources within geriatric care from institutional care to care of the elderly in their own homes. This is expected to lower the total cost of the care of the elderly as well as to improve quality of life.

Within overall planning, options for the choice of technology should be identified on the basis of the following criteria. First, the cost of an item of equipment needs to be assessed over its entire life in relation to its likely health impact. Then comparisons should be made with other technical options available, especially in the local market, for carrying out the same function(s). Thirdly, the requirements for operation and maintenance should be examined to ensure that these requirements can be met in the future. If a technology is to be applied on a mass scale (e.g., jet injectors for mass immunization or handpumps for rural water supply), an evaluation of its performance under local field conditions should be made and design modifications considered where serious deficiencies are identified.

Because technical choices determine to a large extent subsequent requirements for operation and maintenance, they must be made with these implications clearly in mind if maximum performance is to be obtained. Similarly, because capital equipment often has a lengthy life expectancy, earlier imperfect choices may be unavoidable for many years. Hence dramatic improvements may not be achievable until existing equipment needs replacement.

Cost-effective strategies

The efficiency of strategies (curative, preventive, educative or promotional) might be improved either by reviewing the design of strategies or by strengthening support services for their implementation. Knowledge in health

81

Box 4.3 Cost-effective treatment of tuberculosis

The cost-effectiveness of alternative treatment strategies for tuberculosis has been studied in Botswana. The study compared the cost-effectiveness of a short-course tuberculosis treatment regimen using rifampicin or ethambutol with long course regimens based on thiacetazone and isoniazid. Short-course regimens were more costly per case treated, but were half the cost per person effectively treated by isoniazid-based regimens because of higher patient compliance. In addition, outpatient treatment was much cheaper than a combination of inpatient and outpatient care. The study estimated that treated 80% of patients through an outpatient based short course regimen would have cost one-third of the pre-1984 treatment pattern of combined inpatient and outpatient care and a long course regimen. Furthermore, the number of people complying and covered by the programme would have doubled.

Source : Barnum, H.N., "Cost savings from alternative treatment of tuberculosis", PHC Technical Note 86-11 World Bank, 1986, cited in "Survey of cost-effectiveness and cost-benefit analyses of key primary health care projects in commonwealth countries". (processed no date) by Drummond, M.F. and Mills, A. page 22.

sciences and technology, with concomitant cost and effect implications, has been improving steadily. Recent developments in diagnosis, treatment <u>or primary</u> prevention have raised the need to reconsider the control strategies for several diseases.

Control of diarrhoeal disease and of schistosomiasis offer examples of what can be done. The development of oral rehydration therapy has revolutionized diarrhoeal disease treatment. This safe, effective and inexpensive treatment has sharply reduced not only the cost of rehydration but also the seriousness of the disease. Similarly, the development of an easily administered anti-schistosomal drug has made treatment an economically attractive alternative to the interruption of transmission through the frequent application of pesticides to the breeding areas of the intermediate vector, the snail.

A large number of studies have been made on the cost-effectiveness of existing and new diagnostic tests, drugs. treatment regimens, logistic supply (i.e. materials for ensuring the cold chain in immunization), mass screening, excreta disposal, water pumps, approaches to promoting healthy life styles and so on. Appropriate practical research in the field of management as well as that of technology is of great value for formulating and implementing national health programmes. What is required through this research is creative thinking to develop ways of solving problems.

82

₩₽.CTHEREDE.CO.CTHEREDE.CO.CTHEREDE.CO.CTHEREDE.CO.CTHEREDE.CO.CTHEREDE.CO.CTHEREDE.CO.CTHEREDE.CO.CTHEREDE.CO.C

Box 4.4 An innovation in management development in Malaysia

Team Problem Solving at the District Level

In 1985, the Director-General of Health Services in Malaysia called upon his Institute of Public Health to develop a new style of management training which would raise the ability of staff at State and District level to diagnose and resolve their own problems, and thereby increase the efficiency and effectiveness of their services without requiring additional resources.

The Institute undertook an action, learning-by-doing approach, involving, in the first instance, four state-district teams in the analysis, solution-design and, most importantly, the implementation of solutions to four assigned health service problems (immunization coverage, food poisoning, malaria case-finding and management of high-risk pregnancies).

In February 1986, the teams were brought together for a ten day workshop to facilitate their analysis of the assigned problem and to design and plan the implementation of their solution. This workshop included the analysis of existing data and some field survey activity to obtain more information needed. At the end of this workshop each team presented its proposal for action to a panel of directors who critically reviewed each. The teams were then given permission to proceed to implement their proposals.

After a period of eight months the teams rejoined for three-day evaluation workshop during which they reported to each other and to the same panel of directors the results of their efforts to resolve the assigned problem. The early findings are impressive:

- all teams succeeded in remaining active and were able to implement their planned activities more or less as intended.

 one district was able, through special defaulter follow-up procedures, to increase third dose DPT coverage by 50% and triple measles immunization coverage.

 a second district was able, through extensive health education and promotion of general practitioners and outpatient departments, to quadruple the amount of notification of food poisoning cases.

(Cont.)

 A spectra spectra and spectra s per spectra s per spectra spect

Box 4.4 (Cont.)

- a third district increased the blood examination rate of fever cases in its malarious areas by 80% through the use of community workers, schools and maternal and child health staff.

- finally, the fourth district focused its attention on priority high-risk women and instituted an obstetric flying squad in its efforts to reduce maternal deaths, particularly those due to post-partum haemorrhage. At the time of the evaluation there had been no deaths from this cause in 19 cases, compared with seven of the ten maternal deaths in 1985.

Impressive as these immediate results were, the programme directors were most interested in the ability of the districts to keep their teams active and working together, in addition to all the normal work to be done. Efforts are continuing by the Institute of Public Health to expand this type of management development and to assess the longer term effects of such training in the first districts to be involved.

Source: Workshop for state and district health managers. Problem solving through the team approach. Evaluation workshop, 20-22 October 1986. Institute of Public Health, Kuala Lumpur (Malaysia).

The modification of control strategies sometimes requires little more than a policy decision. A major overhaul of the administration of the sector is not typically required, and problems of patient acceptance and compliance are seldom major. Ministries of health should continue to scrutinize the strategies being used to control specific diseases. These examinations should be especially rigorous when new technologies emerge, such as a vaccine or a safe drug. They should also be undertaken whenever the cost of supplying services changes significantly, more qualified manpower becomes available, or communications and transport improve significantly.

Strengthening of management support services

The strengthening of management support services (in-service training,

TORKEN ADDRESS AND ADDRESS ADDR

supervision, drugs, logistics, transportation, organization), is crucial to sustain efficient delivery of health services. The provision of these support services has been difficult in many countries because the need for efficient managerial practices to deliver these services to numerous and scattered health units has not been adequately met. Support services are highly dependent on transportation and distribution of medical supplies and drugs. In many countries, serious efforts have been made to improve the management of essential drugs at each of the following stages: procurement, storage, distribution, supply and utilization. Strengthening the managerial capacity of health administrators, managers and health providers is crucial to make programmes workable. Another approach to increasing efficiency and effectiveness of primary

Box 4.5 A strategy to improve supply of drugs

Most developing countries import essential drugs, often from a few major multinational suppliers. WHO describes essential drugs as those that are most needed for the health of the majority of the population. There are about 250 in a model list developed by WHO.

An essential drugs strategy can permit existing government health budgets the capacity to increase significantly the quantity of drugs purchased and thus improve the availability of drugs to the mass of the population. There are four important steps that could help lower costs:

- Selecting drugs and estimating needs. Based on the health problems in the country and the type of facilities available to treat patients, a basic list of necessary drugs and their quantities should be drawn up and kept up to date by the ministry of health. WHO's model list of essential drugs, which is revised and updated on a regular basis, can serve as a guide in this process. The ministry of finance, in its capacity as overseeing fiscal agency for all sectoral ministries, must ensure that the basic drug needs of the public sector are met, while at the same time being aware of the private sector's demands.

– Procurement. The lowest prices can usually be obtained by international competitive bidding, or by purchase through UNICEF. In both cases, foreign exchange must be made available on a regular basis to avoid expensive "emergency" local purchases of imported drugs.

- Distribution. Careful planning and purchase of drugs can help meet the basic needs of the population. However, unless careful thought is given to distribution, the drugs may not reach the rural areas where they are needed most. Packing the drugs in preset amounts in tamper-proof boxes that are then shipped direct from the central medical stores to the rural facilities is one way to prevent theft or misdirection.

— Use of drugs. Doctors and other prescribers must be made aware of the costs of their prescriptions, both to the community and to their patients. The use of standardized treatment schedules for the most common diseases is helpful, as is the publication of drug formularies and information sheets by governments.

Source : Catsambus, T. and Foster, S. "Spending money sensibly: the case of essential drugs". Finance and Development, December 1986, pp. 30-32

85

health care programmes is through their periodic in-depth evaluative reviews, focusing particularly on the managerial and operational aspects. Such reviews can take managers to every level of the health service and into the communities, in order to assess programme performance (including coverage, quality and effectiveness), and to identify problems which may be hindering performance. Such programme reviews were initially widely employed by the Expanded Programme of Immunization but were soon extended to include other elements of primary health care including maternal and child health/family planning, diarrhoeal disease control, essential drugs, environmental health and essential care. This approach of joint programme reviews as a form of monitoring and evaluation has been found useful by many countries and has enabled them to improve the managerial efficiency in the delivery of primary health care.

Improving cost-containment

Cost-containment has become a major issue in many developed countries which are confronted with ever-increasing costs of health care. Reasons for rising costs of care are many, for example, resources may be misdirected by systems of health care financing which provide incentives for excessive servicing. Insurance schemes which pay private suppliers of services on a fee-for-service basis are another major example. If the patient is not required to pay part of the bill, then neither party to the health care encounter has any financial interest in controlling costs. On the contrary,

providers of care profit from charging higher fees and delivering more extensive procedures. Moreover, more drugs are frequently prescribed and more doctor-induced consultations are provided. Where insurance has been introduced without safeguards against these potential abuses, costs have risen rapidly. In addition, the use of ever-increasing diagnostic procedures and greater resort to costly hospital-based care have been common

The rising cost of health care, without commensurate improvement in the health of those served, has led many developed countries to seek practical solutions to their fiscal crises. Recent figures show that a marked decline in the growth of health expenditures in relation to gross national product is already visible. Many factors have contributed to this decline. They include a decrease in hospital building. a fall in hospital use, a relative decline in health professionals' income, more use of day surgery, a higher use of outpatient care and day hospitals for the elderly and the mentally ill, the control of pharmaceutical prices, the greater use of positive or negative drug lists, changes in the pricing of medical procedures, and an increase in the introduction of user charges. More and more countries have been placing budget limits on the growth either of total health expenditure or of hospital expenditure.

Influencing the supply side

One approach has been to revise the scale of fees paid by or reimbursable by the insurance scheme. Fee scales must be carefully designed to ensure that

Box 4.6 Public health finance and planning in the Soviet Union

The Soviet Government takes full responsibility for protecting and improving the health of the national population, under obligations set out in the programme of the Communist Party of the Soviet Union. The right of every citizen to health care is established in the Constitution of the USSR. The principles underlying health service provision thus include an emphasis on the preventive approach, the priority of making qualified medical assistance available without payment, the planned nature of health care, and the involvement of public opinion and organizations in measures connected with health.

The greatest proportion of financial resources allocated to health care comes from the State (made by the federal and union republic budgets), with lesser amounts coming from other sources including the collective farms, public enterprises and other public bodies. The Union Republics contribute the majority of State expenditure on health, insofar as their principal concerns are with financing sociocultural measures and developing the local economy. In 1960 they provided 85 per cent of all State health funds, and this proportion rose to nearly 95 per cent by 1981.

There is a wide range of health spending among the Union Republics; in 1981 the Russian Soviet Federal Socialist Republic expended over 8 billion roubles on health, while the small republic of Estonia spent only 106 million roubles. In 1980 Azerbaijan spent 37.8 roubles per head on health, while Estonia spent 71.9 roubles per head; for Moscow this figure was 57.6 roubles per head.

State management bodies have pointed for some time to the need for improved efficiency in the use of resources in all branches of the national economy. This requirement has been addressed in several ways within the health sector. The principal measures involve the refinement of financial standards and the extension of the authority of the heads of medical establishments over the use of budget allocations.

In 1977 the national ministries of health and finance made joint proposals to State control bodies for changes in the financial standards, to make them consistent with developments in medical science and the medical industry, and in criteria for patient care. The Government accepted these proposals and increases were made in expenditure standards for catering and the purchase of drugs and dressings in nearly all health care institutions: maternity homes; provincial, territorial and

(Cont.)

87

Box 4.6 (Cont.)

republic children's hospitals; the clinics of research institutes; cancer hospitals; urban and district centre hospitals; heart, chest, surgical and other specialized institutions. A further increase in expenditure standards was made in 1983 for the purchase of equipment and drugs and for the feeding of patients in specialized care wards. Local health service management bodies and the corresponding subdivisions of the Ministry of Finance monitor the use of budgetary allocations by all establishments, to ensure conformity with the standards that have been set out.

Source: Golovoteev, V.V. and Pustovos, I.V., "Public Health Finance and Planning in the Soviet Union," World Health Statistics Quarterly, Vol. 37, No.4, 1984, pp 364-374.

some procedures are not exceptionally remunerative and therefore encouraged. Very high hourly rates of compensation have been implied by fees paid for surgical procedures as compared with other forms of treatment and in some countries may have promoted excessive surgery. Relatively high rates for diagnostic procedures have also encouraged their use.

A second possibility is to shift the basis for payment from procedures to either encounters with the health care system or an episode of illness or a hospital stay. This device pays providers for a collection of activities from which they may choose; thus the economic incentive is to expend as few resources as possible. The defect in this approach is that providers are rewarded for neglect and inadequate care.

Finally, direct regulation of the content of care has been used as a way of controlling excessive use of health resources. Review committees have been established by hospitals to examine patterns of diagnosis and treatment by individual practitioners and to evaluate the outcomes of care. These professional review procedures have not been fully successful largely because of the reluctance of peers to accuse colleagues of abuses or malpractice. Statistical analyses of patterns of resource use have also been undertaken by insurers in order to identify excessive use or possible fraud.

Influencing the demand side

One method for cost-containment and reducing unnecessary procedures has been to require patients to bear part of the cost of care. A nominal fee may be charged for access to the health care system and a share of all subsequent costs may be charged directly to the patient. This sharing of the costs of care restores some responsibility to the patient to use resources judiciously and, at the same time, heightens the patient's interest in containing costs.

Providers of health services frequently claim that health facilities are overused by the "worried well" and those seeking to obtain exemption from work or other responsibilities. Patients also fail to complete courses of drug therapy, to return for evaluation of treatment, and to follow through on referral to other sources of care. The magnitude of losses

88

Box 4.7 Cost-containment in some industrialized countries

Cost-containment in health care is feasible and indeed, has already been achieved in some industrialized countries. It is a question of political will, acceptance by the public, cooperation of providers and the use of effective methods. In a number of countries, Australia, Belgium, Canada, Finland, Norway and the United Kingdom, health expenditures, as a percentage of gross national product, have levelled off.

The following have been the most important cost-containment measures taken on demand or on supply. Measures operating on *demand* are described as cost-sharing – giving the term a wider use than is current in the United States. Cost-sharing means that the user has ultimately to pay or have paid on his behalf part of the cost, normally at time of use. This payment may be intended to discourage user demand or act indirectly on the doctor or dentist who authorizes the use of resources through the knowledge that the user will have to pay. The share the user pays may be a flat rate charge or a proportion of the cost or both. There may be related to income. It may represent that part of the cost which is not reimbursed to the patient – the *ticket modérateur* in France. The extreme case is where the scope of what is provided or covered by insurance is reduced. This throws the whole cost on the user (e.g., spas or phamaceuticals which can be purchased without a prescription).

Systems of cost-containment operating *on supply* are of even wider variety and have in recent years taken more complex and ingenious forms when applied to contracted services.—They include the following:

Short-term direct controls

Budget ceilings (e.g., for hospitals) Controls on staff numbers (e.g. in hospitals or clinics) Controls on levels of remuneration Controls on prices (e.g., per bed/day or per pharmaceutical product) Controls on quantities (e.g., the maximum number of items in a prescription).

Short-term indirect controls

Changing relative value scales (e.g., to make it less remunerative for a doctor to order diagnostic tests).

A positive list or negative list for pharmaceutical products or chemist substitution (to stop the prescribing of more expensive products where there are less expensive products considered to be equivalent). Restrictions on sales promotion of pharmaceuticals to doctors. (Cont.)

and the second second

sont.)

Box 4.7 (Cont.)

Information to doctors on relative prices of pharmaceuticals which are equivalent or near-equivalent.

Doctor profiles (e.g., indicating the number of visits per patient, the prescribing of drugs or the requesting of diagnostic tests per patient or per consultation) with or without sanctions for excessive use.

Medium-term direct controls and incentives

Controls over the construction or extension of hospitals. Controls over the installation of heavy or expensive medical equipment. Controls and incentives to develop substitutes to traditional hospital care.

Long-term direct controls on manpower

Controls over the number of students entering medical or dental education or passing into the second year of the course.

Controls over entry to specialist training.

Source: Abel-Smith, B. Cost containment in 12 European countries. World Health Statistics Quarterly, vol. 37, No. 4, 1984, pp. 351-363.

due to such misuses of health resources is not known.

Misuse of health resources by consumers is in large part a consequence of insufficient knowledge of the proper use of health care facilities. Problems of non-compliance may also result from lack of knowledge. Health care providers must also shoulder some of the responsibility since they often do not adequately explain the rationale and expected benefits from a course of action. Better education in the use of health care and improved health education would reduce the misuse of resources by consumers.

Conclusions

Making better use of resources implies improved accountability, increased efficiency in the allocation and utilization of resources and effective means of cost-containment. 5

To reduce waste, underutilization and misuse of resources, accountability must be improved. Information and education of the public are important measures to achieve better accountability.

Secondly, to increase the focus on the performance of individual health providers and of health care institutions, a relevant accounting and management system is required. Information systems have to be set up

in such a way that responsibility for performance can be clearly established. Detailed, accurate information on costs should be made available to the providers and consumers of health services. Earlier chapters of this document have also suggested that programme budgeting and cost-accounting systems need to be introduced. This implies relevant training in accounting and management for health care managers and better organizational arrangements to review costs and performance in terms of effectiveness and quality.

ŝ

Thirdly, to generate required changes in the delivery of services, lines of command need to be streamlined; and real authority is required in decision-making at each level of supervision and reporting. The potentiality for the delegation of

THORE BREAK BREAK

a and share to the state of the second

responsibilities must depend upon the potential capacity of the various partners: individuals, communities, nongovernmental organizations, private institutions. Public oversight is needed especially for public health interventions such as vector control, epidemiological surveillance and immunization. In order for consumers to oversee the performance of the sector more effectively, formal local organizations, such as health committees or boards are needed in many instances.

In summary, the use of resources by the health sector can be improved by increasing the availability of information about performance and enhancing the responsibility and managerial capacity of individuals, communities and health managers to evaluate and act upon this information.

91

Responsibilities and institutional relationships in economic support for health for all

Background

Health depends substantially on decisions made by individuals, families, groups, and organizations, all of whom contribute to its maintenance and improvement. Moreover, important roles in promoting and maintaining health are played by the health-related sectors such as environment, education, agriculture, and housing. Participation in health for all is not just a question of democratic rights, it is an essential requirement for the effective and equitable planning and allocation of resources. Commitment precedes action, and the collective commitment of all concerned is required in order to ensure the equitable distribution of resources for health. Given the magnitude of the task of attaining health for all, and particularly of securing adequate economic support, concerted and coordinated action at all levels is indispensable. Action taken within any one component affects the action to be taken within the others. Hence, the shared responsibility which rests upon the individual, the community, the government, other national agencies and international agencies consists of specifying not what is possible, but what is necessary; not what can be done, but what must be done.

This chapter examines the responsibilities of different entities involved in health matters, the roles to be played by each in pursuing the goals of health for all, and the relationships

between them. These include the government health agencies and other relevant agencies; people; the private sector; the community; the nongovernmental organizations; and the external agencies.

Commitment on the part of each entity is reviewed in the context of providing universal health care coverage, particularly where responsibility is shared. A coherent strategy for their participation in planning and carrying out programmes in primary health care and in the management of those resources which are committed to the programmes is advocated so as to maximize economic support for national health-for-all strategies. Finally, the implications for the strengthening of institutional relationships and capability are identified.

Government responsibility

Governments, throughout the world and under all forms of political and economic organization, have responsibility for the health of their people which can be fulfilled only by the provision of adequate health and other social measures. Therefore, much of the action required to meet the goals of health for all falls within the responsibility of government. While no universal blueprint can be applied to all governments, the essential characteristics of health systems stated in the Global Strategy for Health for All

define the means of developing infrastructures for optimum health delivery based on primary health care.

unume Support for National realth for An

Governments' responsibility for health originates in the realization that individuals, voluntary organizations and the private sector are unable to meet countrywide health needs through their own efforts alone. Similarly, services which benefit the country as a whole should be provided without charge to individuals by the government. This would, for example, include mass immunization (the successful smallpox eradication campaign is an excellent case in point), and the control of vectorborne diseases.

National profiles of health and disease indicate the vast inequalities in health status which exist in many countries. The underprivileged form part of every country's population. People may be landless, members of minority ethnic groups, disadvantaged because they are old, sick, homeless, jobless, or inhabitants of remote marginal areas or displaced persons. Most of the underprivileged in health terms are women and children whose well-being requires special attention within primary health care. The elderly are also vulnerable in many countries. Particularly in developing countries, these underprivileged elements of society are politically weak and are often too sick, uneducated, or geographically dispersed to become politically active. Thus, the responsibility for improving the prospects for these groups will fall upon the more privileged groups. Concern for the care, protection and promotion of the health of the vulnerable and

underprivileged groups must be the task of governments, which have, after all, the responsibility for ensuring an equitable distribution of society's resources based on the principles of social justice.

The primary health care approach, focusing on the "eight essential elements", evolved in response to the mis-match between the distribution of health resources and health needs. Disadvantaged people not only suffer from poor health but receive poor health care. Conversely, wealthier people enjoy not only better health but better access to health-related facilities. Development strategies which are compatible with the primary health care approach contrast with single-purpose quests for economic growth regardless of the human consequences. Strategies which at the same time promote economic growth and reductions in social inequalities and increases in social services are consonant with the goals of health for all.

But government money alone will not guarantee the achievement of health-for-all goals. Political commitment and support is also critical to the success of health for all. Indicators of the degree of political commitment are, of course, qualitative in nature and comprise the following: official declaration of high-level commitment; allocation of adequate resources (the most important indicator); the degree of equity in the distribution of financial resources; the degree of community involvement; and the establishment of appropriate organizational frameworks and managerial processes (including

monitoring and evaluation) to move step by step towards health for all.

The importance of government intervention in the distribution and utilization of health resources varies with the service to be provided. The extent to which responsibility can be imputed to the government for delivering medical services versus the extent to which it should remain with the individual is continuously debated. At the heart of this debate between individual responsibility and social concern is the relative desirability of enhancing overall quality versus improving equality of access.

Improving the health status of the population cannot simply be achieved by governments expanding or leveloping the health services. revention and control of disease and he promotion of health must be part of a combined effort for the mprovement of well-being as a whole. fhus, health care must be supported by mprovements in the overall social and conomic structure and contributions rom sectors other than health. The lealth sector has the important role of oordinating these intersectoral efforts. y defining major problems, by uggesting preventive strategies, by roposing shifts in priorities and source allocations, by encouraging ositive action in other sectors, and by articipating in advisory committees sponsible for these efforts.

he main motivation for government volvement in primary health care has een to ensure that no one is denied tre on the basis of income or wealth. owever, this purpose is frequently not hieved, since lack of funds limits the

level of medical service provided to significant proportions of society. Additionally, the increasing pressure of recurrent costs on public budgets is forcing governments to review the mechanisms used to finance health care.

Three areas of consideration are apparent. First, limitation of resources tends to diminish the quality of services provided. Next, financing through general tax revenues may cause a redistribution of economic resources from the unserved to those obtaining services. Third, those services which are available are not being distributed equitably.

One proposal for mobilizing additional resources is to impose charges on non-emergency users of hospitals who have not been referred by a lower-level health facility. In addition to generating revenues, such user charges would also discourage the bypassing of primary care facilities. In effect, the charge would represent the additional cost of managing a health problem at a hospital rather than at a dispensary or health centre.

Redistribution of consumption in the health care system could also be rationalized by introducing a compulsory health insurance system for paying for health services. Such insurance systems would levy insurance contributions on those with regular jobs who generally have access to services, but would exclude those who do not.

Individual and family responsibility

Primary health care originates with people and their health concerns. With this dominant role in health, people have to be actively involved rather than passively receiving care "from above". The rights and privileges of the people include the right to equal opportunity for health; the right to health care; the right to be informed; and the right to be involved. As partners in the health-for-all strategy people should have an opportunity to share responsibilities and to contribute actively at individual, family, group and community levels. They should develop the ability to define and express their needs, with the awareness of when and how to use health care to satisfy those needs. They have the right to use the opportunities that exist to obtain the required information, to analyse it and to draw their own conclusions. Once people are aware of health problems they should have access to the knowledge and information which allow people to make choices in health care. The right of people to be actively involved in health care ensures that satisfactory prerequisites for health exist for all people; that their environment is healthy; and that their health care system is responsive to their needs.

People pursue good health and provide health care themselves through a wide range of activities. Self-care, for example, is the primary health resource in the health care system. Included are informal health activities and health-related decision-making by individuals, families, neighbours, comprising self-medication,

ACCORDING OF DESCRIPTION OF DESCRIPT

self-treatment, social support during illness, first aid, etc. Another term, "lay care", describes all health care given by lay people to one another in both natural and organized settings. In addition to family units, self-help groups are voluntary, small group structures for mutual assistance in satisfying a common need, overcoming a common handicap or problem, and bringing about desired personal or social change. Groups may provide emotional support, information and advice, direct services, social activities, and pressure group activities. There is, of course, a continuing debate over the issue of self-help in both the political and professional fields, which could lead to a critical assessment of the quality and efficiency of the health care system in general.

э.

In many developed countries national health-for-all strategies have identified individual and family responsibility as a key element. This arose out of the fact that health profiles of the population have changed considerably over the past three decades. Aging populations, health problems related to life-styles, changing cultural patterns, and burgeoning industrial development are a few of the factors conducive to rapid change in health profiles. Health promotion strategies under preparation in developed countries aim at enlarging the individual's responsibility in pursuing and maintaining improved health. Particular emphasis is placed on hypertension, cardiovascular diseases, tobacco-related diseases, etc., as conditions which are conducive to changes in life-style. Some countries, such as the USA, have achieved significant decreases in hypertension and tobacco-related morbidity and also

Box 5.1 Partnerships for health promotion – United States of America

In 1979 the first Surgeon General's Report on Health Promotion and Disease Prevention, *Healthy People*, was issued. The report chronicled a century of dramatic gains in the health of the American people, reviewed present preventable threats to health, and identified fifteen priority areas in which, with appropriate actions, further gains could be expected over the decade. The report established broad national goals for the improvement of the health of the Americans at the five major life stages: specific and quantified objectives were established for the attainment of these broad goals and each of the 15 priority areas defined in the Surgeon General's Report.

These health promotion objectives for the nation comprise the health of the Americans by 1990. The objectives are not primarily about disease treatment and control, they are about preventing disease and encouraging people to protect their health. The target areas fall into three groupings: personal preventive services, health protection, and health promotion.

Preventive services are those provided by physicians, hospitals, and other health care providers, and the targets are high blood pressure control, family planning, pregnancy and infant health, immunization, and control of sexually transmitted diseases.

Health protection includes efforts by governments, industry, and other organizations to reduce health hazards in the environment. Targets include the control of toxic agents, occupational safety and health, accident prevention and injury control, water fluoridation and dental health, and surveillance and control of infectious diseases.

Health promotion aims at educating the public about the risks involved in health abuses and to increase public commitment to sensible life-styles that can add years to life. Targets include the health risks of smoking, alcohol and drug misuse, and stress and violent behaviour, along with the benefits derived from good nutrition and physical fitness.

The implementation strategy for these objectives and targets stresses a pluralistic process involving public and private participants from many sectors and backgrounds. Health officials and health providers must be joined by employers, labour unions, community leaders, schoolteachers, communications executives, architects and engineers and many others

and the second sec

(Cont.)

97

Box 5.1 (Cont.)

in efforts to prevent disease and promote health. While the Federal Government must bear responsibility for leading, catalysing and providing strategic support for these activities, the effort must be collective and it must have local roots.

Sources: "Promoting Health, Preventing Disease, Objectives for the Nation", Department of Health and Human Services, USA, 1980 (reprinted 1984).

in traffic accidents through developing and implementing strategies which emphasized individual responsibility and included active health promotion measures. The economic aspects of these modified strategies have not as yet been given sufficient attention. On the other hand, the benefits which have accrued already through individual responsibility in health make a strong case for economic support to such health promotion efforts.

Role of the private sector

The private sector constitutes an important component of the health system in many developed and developing countries. It is not just limited to the provision of medical care, but also involved in the manufacture and supply of drugs, and medical supplies and in the provision of water supply, sewage and solid waste disposal services. In the new approaches which are being designed to produce and finance primary health care, a broad spectrum of institutional arrangements to involve the private sector are possible. Different roles have been assigned to government agencies, parastatal organizations and private firms. The suggestion that autonomous and profit-seeking institutions might play an important role in primary health care may seem surprising.

However, profit-seeking organizations already manufacture and sell drugs, build and equip health care facilities, water treatment plants, etc. Thus the issue is not whether private and public health institutions ought to work together, since they already do; but, rather, how can governments shape relationships with private agencies in order to achieve social and particularly health-for-all goals promptly and more efficiently. ç

Private health professionals have an important role to play in all countries. Their expert knowledge and the influence they have in the health secto relative to politicians and the general public can be instrumental in mobilizing support and initiating change for the health-for-all movement Further, they can accord higher priorit to health promotion, disease prevention, care and rehabilitation that is often the case at present.

Of course there are some training, regulatory and information roles that only the public sector can perform in overseeing and guiding the activities o private providers. The public health sector needs to take the lead in trainin health workers, testing them for competency, and licensing private facilities. The government must set standards and regulations to protect th

98

Box 5.2 Philippine medical care plan

The Philippine Medical Care Plan was conceived, formulated and realized through the initiative of the private medical profession represented by the Philippine Medical Association. The Plan was developed in response to the realization that about 75% of Filipinos did not receive medical attention prior to death due to the maldistribution of medical manpower and the lack of hospitals. In addition the low-salaried employees of government, private business and industry were recognized to be crowding out the truly indigent from government "charity" beds.

In 1963 the Philippine Medical Association launched the MARIA Project (Medical Aid to Rural Isolated Areas) to settle young medical graduates in communities who pleaded for medical assistance and doctors. The Project entered into agreements with town officials and the people themselves to help the young doctor in his work, provide a place for the clinic or whenever possible to set up a 5-10 bed hospital, and to cooperate in any health or community project that the town leaders might embark upon to uplift the socio-economic condition of the community. In this way the MARIA Project, invoking the cultural tradition of "Bayanihan" or helping the member of the community, and the spirit of self help, became the foot in the door for total health care and community development. Between 1963 and 1968 the Project was able to establish 35 medical clinics and hospitals without government fiscal and material support, but through private initiative by the medical association and the people themselves.

The Committee on National Health Planning of the Philippine Medical Association was organized in 1965. It conceived and formulated the Philippine Medical Care Plan of 1969 which was passed by Congress in 1969 as Republic Act 6111.

The medicare plan was a social health insurance scheme to which members contributed according to their income levels in order to build a health insurance fund. From such a fund contributors may be reimbursed their expenses for medical care. In this social health insurance concept all contributors received equal benefits regardless of the amount of their contribution.

The plan was also designed to be comprehensive. It was to cover all the people and possibly include some aspects of preventive care such as immunization and procedures for population growth control.

(Cont.)

Box 5.2 (Cont.)

The health providers such as the doctors, dentists, hospitals and drugstores were to be accredited for participation in the medicare programme. They were expected to abide by the rules and regulations, and limitations of the Plan.

The Medicare Programme ensured free choice of hospitals and doctors within reason, and was on a fee-for-service basis, although such reimbursable fees may be limited by rules and regulations and the patient may have to cost-share depending on the nature or gravity of his illness, and the corresponding expenses.

It was also conceived for practical purposes that there were to be two programmes: one for the formally employed in government and the private business and industry sector, and the other for the self-employed such as the small entrepreneurs, farmers, fishermen, tenants, etc.

Source: Tamesis, J. V. "Partnership and responsibility in financing health activities: The Medicare Programme of the Philippines". A position paper for the World Health Organization, 26 August 1986.

people from unethical or untrained practitioners, especially where such supervision is not yet adequate through professional associations. Governments need to develop a legal framework for health insurance systems, and disseminate information to inform and educate the consumers.

Precise information is lacking in most developing countries about the extent of the private sector's involvement in health care. Even the term "private" is given different interpretations in different countries. Available data show that in many developing countries, the private sector finances over half of the health care services, especially in the urban areas. The role of the private sector or the methods for its full participation in national health-for-all strategies have not been defined in most developing countries.

At the same time, new modes of cooperation are also emerging both in the developed countries and in some o the developing countries. These are generally based on the financing mechanisms being applied in providing and extending health care. Some of these have been discussed in Chapter III. These experiences need to be carefully reviewed by countries in orde to apply solutions which are appropriate to their own socio-political and economic situation, and which serve to expand the real economic base for the achievement of the goal of health for all.

3

Community responsibility

It is generally agreed that community participation is an essential principle of the primary health care approach, and

no declaration on the subject by a national government or an international organization appears to overlook this requirement. However, development proposals generated by remote government officials often ignore these grass-roots systems. At the same time, there is widespread agreement that communities can play an important role in the identification of health problems and in the search for appropriate and cost-effective solutions. Where public funds are insufficient to extend basic health services to those without access, communities may be called upon to share the burden through contributions in labour, materials, or money. Community members have often been involved on a voluntary basis in defence of their own health. Some of the mechanisms being used by communities in financing health services have been discussed in Chapter III.

The political and administrative structure of some countries lends itself to local community involvement. Many countries, especially those with large rural populations, have long-standing traditions of community participation in all local development activities. The support of a variety of local mechanisms, often reinforcing traditional systems, has contributed to progress in local primary health care development. Conversely, low levels of education, differing customs or beliefs concerning the causes and nature of ill-health, and long dependence on central government for all action and resources breed passivism and forestall successful community involvement. The emphasis of community support in most developing countries has been on providing resources, either financial or

material and human, for the establishment or improvement of health and sanitation infrastructures – health centres, latrines, drug cooperatives, feeding centres, or for the payment of community-based health workers. Important contributions have been made by communities in the form of volunteer services. Sustained community involvement, however, has remained difficult in most countries.

A clear division of responsibilities between the community and the health system, and effective mechanisms for mutual support and communication as well as administrative decentralization, are of critical importance in mobilizing full partnership for health promotion from the local communities.

Nongovernmental organizations

Many nongovernmental organizations (NGOs) have grown into national bodies, such as associations, clubs, and trade unions; and some have become transnational, principally through the rapid development of communications. They differ widely from each other and represent the most varied activities and interests. They are independent of governments and possess their own funds. Generally their aim, in the context of health care, is to assist people in the lower-income segments of the population of countries in which they operate to organize and utilize better their own resources, both human and material, with special emphasis on the local community level.

While it is difficult to arrive at accurate estimates of the financial contributions

101

by NGOs, health financing in 1982 by these agencies accounted for eighteen per cent of all health contributions and equalled the total input by the United Nations health-related specialized agencies.

Many NGOs deal with problems that are intersectoral in nature, growing out of such basic causes as poor education, poor health and nutrition, inadequate community infrastructure, high population growth rates, low productivity, insufficient income-earning opportunities, lack of effective community organization, and attitudes of over-dependence and apathy that frustrate problem-solving competence at the local level. Since their motives are generally not suspect, NGOs can deal with these problems with effective, timely and flexible inputs of relevant resources.

The growing awareness and wider acceptance of the role NGOs play in health care have resulted in a number of successful initiatives and projects which demonstrate the value of the partnership approach. Such initiatives include small-scale local voluntary organizations which may focus on socially deprived or under-privileged groups in a given geographical area, providing essential medical care, nutrition, community education services and promoting self-reliance. Or they may be large-scale national-level organizations to whom the government has assigned almost full responsibility for a specific health programme. National family-planning associations provide excellent examples of such cooperation in several countries. Organizations for the care of the disabled also play a critical role in an

aspect of health care which is generally very difficult for government health services to cope with in many developing countries.

The Technical Discussions on the Role of Nongovernmental Organizations in the Strategy for Health for All, held in 1985, suggested many practical measures for enhancing partnership relationships between governments and nongovernmental organizations. Nongovernmental organizations as operational partners can make a crucial contribution to the national health-for-all strategy and to international cooperation.

International cooperation

Many external agencies provide support to health programmes. These include bilateral agencies, multilateral agencies. including those of the United Nations system, lending agencies such as the World Bank, or international nongovernmental organizations and private foundations. In some of the developing countries, such support can be very significant.

Each agency works within its specific policy framework and operational procedures. These may not always be in concordance with each other and sometimes may even be in apparent conflict. When these efforts have not been well coordinated, recipient countries have been confronted by contradictory requirements and approaches. Sometimes, the national priorities are influenced by the level of external inputs which are intended for a specific health problem or programme.

Box 5.3 Planning for equity in health – Zimbabwe

As part of the process for developing a health policy consistent with the economic policy embedded in "Growth with Equity", and thus based on achieving equity in health, the Government of Zimbabwe undertook a review of the health sector. The following is a summary of some of its findings on the overall structure of the health sector and some of the policy directions.

Overall structure of the health care sector

The Government of Zimbabwe inherited a fragmented health care system. Five sub-sectors predominated in the provision of "modern" health care: the Ministry of Health, local governments, missions, industrial medical services, and the private medical sub-sector. Various voluntary organizations, also, were making contributions to health care, most in specific problem areas. Each provider had its own character, structure and system of financing, and there has been little coordination among them. The most marked characteristics of the health care sector of Zimbabwe were the inequitable distribution of facilities geographically, and the discriminatory access to care on the basis of race and the ability to pay. The system had ensured excellent standards of care for the few, while seriously neglecting the mass of the people, especially in the rural areas. The style of practice stressed curative interventions by doctors, while the preventive and community aspects of health care had been weak, despite the broad scope for action to eliminate avoidable disease.

The primary health care approach

The Government adopted a policy to achieve equity in health. The policy emphasized the primary health care approach, broadening the social base of health activities and restructuring the health service from the base upwards. The policy also stressed the need for a coherence between development and health policies and included as a priority for the government the strengthening of its institutional mechanisms for integrating development and health planning and for coordinating the work of the agencies responsible for health-related sectors, notably: health care, water supply and sanitation, local government and housing, rural development, agriculture, community development and women's affairs, and education.

Political decision-making and interministerial coordination for health

Given the close dependence of health promotion on socioeconomic conditions the Government has proposed institutional mechanisms for coordinating health with development policies action. It is recognized that the discussion and decision-making on broad policy lines on (Cont.)

103

Box 5.3 (Cont.)

health-related issues takes place at high government level. To provide a supporting technical mechanism to this level, the formation of a small intersectoral group of health and development planners is proposed. This planning group would call on the country's available expertise on health-related issues, drawing into a collaborative network the appropriate departments of other government agencies (e.g., Finance, Local Government and Housing, Material Resources and Water Development, Women's Affairs, and Education), resources of unversity and popular organizations such as trade unions, women's and youth groups.

Source: "Planning for equity in health, A sector review and policy statement", Government of the Republic of Zimbabwe, Ministry of Health.

There is a need for effective coordination between external agencies and countries to develop realistic and effective national health strategies. Some countries have established coordinating mechanisms under one single agency, such as the planning agency, which serve to harmonize national health practice and external agency inputs and to coordinate international cooperation. Such mechanisms are needed in all countries.

Institutional relationships

Health care activities are supported and directed by both formal and informal organizations. In developing countries formal organizations usually comprise the ministry of health and its affiliated agencies, although organized professional groups may also have an important role in health care. Informal organizations are groups of people who organize themselves in order to establish health care priorities and to mobilize local resources. Frequently, the local organization for health care is one that was established for other reasons. These may be agricultural cooperatives, women's clubs, labour unions or local units of a political party. Because marketing cooperatives are organized to promote local interests and have achieved acceptance, they have been successful. In some countries, administrative responsibility for local or district services does not rest with the ministry of health, but rather, with the ministry of local government or the territorial administration.

A lack of information precludes a thorough analysis of the respective share of the responsibility and contribution of different entities involved in health matters. However, it is clear that in many developed and developing countries the public sector controls only a portion of the overall resources available for health programmes. What strategies, therefore, can be proposed for a coordinated and coherent approach to the provision of health care to the population which respects the principles of equity and primary health care?

104

Collaboration between various institutions and agencies that comprise the health sector is essential and should be based on clear allocation of responsibilities in order to make the most efficient use of resources. Countries should examine whether such coordination would be valuable, feasible and acceptable and determine the most effective mechanisms for achieving it.

nesponsionales and motivational relations.

No universal blueprint can be applicable for organizing institutional relationships of different subsystems or entities involved in health matters. This, no doubt, will vary with the socio-political organization and policies of the countries. What is clear is that more coherence should be achieved among the various subsystems and all subsystems should reflect primary health care as their major goal.

The successful pursuit of the strategy in countries will depend on one authority being responsible for directing and coordinating national health matters. In most countries, this will be the ministries of health. This does not necessarily imply direct administration of all health resources and facilities. It does imply the responsibility for channelling activities into the national strategy for health for all and coordinating them on behalf of the government, both within the health sector as well as within other sectors through the appropriate channels.

Experience is already emerging from several countries which have undertaken administrative and legislative reforms in order to pursue the integrated development of a health system based on primary health care. The short-term and long-term

୶**୶୷୶ୄ୷୷ୄ୷୷୷୷**୷୷୷୷୷୷୷୷୷୷୷୷୷

implications of these reforms and measures need to be examined in detail. But this goes beyond the scope of this paper. It would be sufficient to suggest that a careful review of the institutional responsibilities and mechanisms to bring about a coherent relationship among them would be an important ingredient of success in mobilizing economic support for the goal of Health for All.

Strengthening national capability

In earlier chapters it has been suggested that countries need to strengthen their financial planning and management by establishing programme budgeting, introducing management by costed objectives, developing unit costs for a whole range of objectives, improving their capability in cost-effectiveness analysis and for estimating costs of their plans for health for all with identified sources of finance which are realistic and administratively feasible. This will require major extensions in the information needed for health planning and management, a development of capacity for processing

and analysing financial and economic data and the acceptance by senior managers and planners of the importance of these new inputs to the process of decision making.

Many ministries of health in developing countries lack the skills for this type of work, whether it is in devising new costing systems, using macro-economic data, applying systems of national accounts to the health sector, or making economic projections. Such skills may be concentrated in government departments concerned with economic affairs whose staff lack a knowledge of health concerns and who are reluctant to see the significance of health in socioeconomic development. Training in public health throughout the world, with few exceptions, has neglected the study of financial management and health economics, and particularly their application to the public sector. Where both economic planners and health planners tend to be particularly inadequately prepared is in the search for alternative sources of financing. Little may be known about ways of financing health services which have come to be widely used in other countries, and their effects.

Exchanging and sharing experiences and information are clearly needed among the national policy-makers and health administrators in this important aspect of their work. Training is also needed for administrative and technical staff who will work in this area. Skills are needed in the establishment and handling of financial statistics, in health economics, in national accounting and public finance and in social security planning. 2

Research and development efforts on resource issues need to be intensified particularly in developing countries. Such issues can be related to the allocation and use of health resource, the financing of health programmes especially through new types of options, and cost-effective and economic analysis of health programmes. Results of such pragmatic research can then be applied to formulating or modifying policies related to economic support for health care.

Technical capability in these areas will require considerable strengthening in most developing countries. Research and health policy analysis will require close coordination in order to provide

As countries reorient their health systems to primary health care, the need for change is evident. These changes go beyond physical design and may involve the redefinition of the objectives of the principal health institutions, the reallocation of responsibilities, and even the revision of the power structure. This presents important political and managerial challenges to the national leadership. Carefully yet courageously pursued, they would serve to expand the economic support for national strategies for health for all.

intelligent and sound information to the health policy-makers. Innovative approaches are needed to strengthen national capability in these aspects.	national and international resources, the commitment and involvement of people in shaping their future is a key factor. Where a mix of private and public agencies controls health care, close coordination and balanced decision-making must be assured.
Conclusions The strategy for health for all calls for organization and strengthening of health infrastructure based on primary health care, beginning with the individual and reaching to top government levels. Together with the mobilization, coordination, and rationalization of	Policy and institutional frameworks ar needed in countries to provide a coherent strategy for full and active as well as mutually reinforcing, participation of all potential partners i health care. National capability for dealing with economic aspects of healt strategies will require considerable strengthening.
·	

......

nesponsionnes and more and relations, relationships in contentio support to recent to an

Ê

unapter VI

8888889999944444444

Key issues for discussion

Economic issues influencing health development are varied and complex. They are closely interlinked to political, social and economic structures and environment of countries. No attempt has been made to cover the entire spectrum of issues. The following issues have been selected to focus discussion on what appears to be critical to the attainment of economic support for national strategies for health for all for a majority of countries. They are based on the analysis offered in the relevant sections of the background document.

109

Economic policies for equity in health

Ten years ago, when the Member States of the World Health Organization unanimously adopted the goal of Health for All by the Year 2000, they endorsed the principle of *equity in health*, that is, shrinking the gap in the health status of the people and countries and ensuring equitable distribution of health resources. This called for a concerted political will and response.

The past seven years have been the most turbulent for the world economy in over half a century, seriously affecting domestic priorities and programmes in both the developed and developing countries. The adverse economic situation, which still prevails, challenges policy-makers seeking to achieve a balance between economic and social goals.

To protect the poor and vulnerable during the process of adjustment, policies and strategies of adjustment-with-equity are even more strongly required. The causes of disparities in health status can only be removed through intersectoral actions involving the health-related sectors as well as resource allocation policies which give preference to the poor and vulnerable groups of populations.

Policy-makers in health need to be strong advocates for promoting social priorities in economic adjustment policies. They must mobilize commitment and support from other sectors and they must enhance their capability in defining equitable schemes of financing and of allocation of resources.

It is hoped that this discussion group will be able to share experiences in

AND CONTRACTOR OF THE AND A CONTRACT OF THE AND A CONTRACT OF THE ADDRESS OF THE

formulating and maintaining policies aimed at achieving equity in health. It will be useful to discuss the process of arriving at such policy measures, including any difficulties encountered and how these were overcome.

- I. In the review of national experience, the following issues may be addressed:
 - a) What has been the impact of the economic recession on public health policies in general (i) in terms of priorities; (ii) in terms of availability and allocation of resources?
 - b) Are there examples where evidence of progress towards equity in health (in terms of coverage, target groups and provision of essential health services) can be given?
 - c) How has economic support been given to make such progress:
 - by shifting resource allocation?
 - by changing the financing system?
 - by targeting resources at specific groups or health problems?
 - by other means?
- II. Where the public sector controls only a portion of the overall resources available for health, what policies and strategies can be proposed for a coordinated and coherent approach to the provision of health care to the population, which respects the principles of equity and primary health care?

111

an in

æ

з

III. What specific actions would be required to improve the capacity of ministries of health in the analysis and evaluation of economic aspects of their national health policy objectives and in the formulation of policy options?

المرجوعة والأولية الممالية المعادر المراجعين

112

Issues paper INO. 2

Financial planning

Financial planning for health for all is an integral part of the managerial process for national health development. Policy formulation, programming and implementation should occur interactively, supported by technical information, programme budgeting and evaluative feedback. But ministries of health are often poorly equipped to plan, allocate, budget and control their own resources. To a large extent these difficulties derive from government budgeting procedures which are more concerned with internal audit than with policy objectives, such as equity and efficiency. Certain other sectors share these problems, while some such as water supply - which may be health related - have better management information systems in which financial and economic data are important.

Where expenditure data are exclusively linked to budget line-items, it is difficult for health agencies to describe what they are doing in a clear way, consistent with stated policy objectives. Unit costs, even of major facilities, are often lacking, inhibiting analysis of the recurrent cost consequences of investment decisions. In turn, forward planning, the appraisal of sustainable capital projects, the identification of wastage or efficiency, and the monitoring of progress to greater equity, are all frustrated. A centralized tradition of expenditure responsibility compounds these information weaknesses. National health development plans, which may outline a strategy for health for all, sometimes omit cost considerations, or build on unrealistic resource assumptions.

CONTRACTOR CONTRACTOR CONTRACTOR

To sustain a credible dialogue with ministries of finance and planning, and even more importantly to improve the value obtained from limited domestic resources for health, ministries of health should incorporate considerations of existing resource use, and of likely resource availability in their planning.

It is hoped that this discussion group will be able to share experiences in financial planning for national health-for-all strategies. It will be useful to discuss the process of developing such plans, the measures taken to close the resource gap, the difficulties encountered and how these were overcome.

The following issues may be specifically addressed:

- Have countries been able to estimate the financial requirements

 for their national plans of action to achieve health for all? How have these been estimated? What period
 do these-generally correspond to (5)
 - years? up to the year 2000?).
- II. If a gap between the requirements and resources likely to be available is detected, what options have been considered and what measures have been applied for closing the resource gap? What have been the positive or negative consequences of these measures? Have these led to further readjustments of policies?
- III. How have countries planned for the recurrent cost implications of their national plans of action to achieve health for all?

113

IV. What measures have been utilized by countries to improve their financial management? What specific actions would be required to strengthen the national capacity in this area?

3



THE CONTRACTOR OF THE STREET STREET
Issues paper INO. 3

Resources mobilization

In a period of declining health budgets, meeting the cost of health services is for many countries becoming increasingly difficult. This is true as it applies to extending primary health care coverage to all segments of a population, but it is also true for sustaining the existing infrastructure of the health services. This situation has led many countries to explore ways of mobilizing additional resources to assure a financing base for the capital and recurrent public health budget and to permit more effective use of the services and resources in the family, community and private sector.

In the mobilization of domestic resources many financing mechanisms have been tried such as: government financing directly from public revenues, or through compulsory health insurance schemes (social security); private health insurance programmes, including those related to employment, cooperatives and benevolent societies; prepaid medical coverage, health maintenance organizations; direct user charges for services; community participation in cash and in kind, self-help.

Mobilizing resources in new ways will have its effects – some positive and some negative – on the range of services which can be provided, and on their availability, quality and cost. Moving into this area will therefore call for careful examination of financing policies and strategies, as well as an assessment of the impact of the financing mechanisms themselves on health status and on utilization patterns, especially among the poor.

It is hoped that this discussion group will be able to review the positive and

negative lessons learned from the experience of countries in mobilizing new resources by applying different financing mechanisms. It will be useful to study the measures countries have taken to correct or avert the ill-effects of certain financing schemes. The applicable experience of nongovernmental organizations may also shed light on financing and revenue generation.

The following issues may be specifically addressed:

- I. What has been the positive impact, and what have been the weaknesses or defects of each financing system with regard to:
 - a) equity in availability and accessibility of care, social costs?
 - b) impact on providers and the quality of care?
 - c) efficiency and effectiveness of service, reliability?
 - d) adequacy of financing available? -
- II. Which financing methods have been preferred, by type of service, e.g., disease prevention and treatment, water supply, sanitation, maternal and child health care, family planning, referral and secondary medical care?
- III. Which organizational, administrative and institutional arrangements have been required for changes in financing mechanisms? What problems have been encountered, and what has been required to correct them? Where are the hidden costs in each

115

financing method? Do the financing mechanisms actually increase resources available?

ھ

IV. What has been the impact of external financing on the national capacity for mobilizing domestic resources? When external resources have been used for capital investment, what have been the recurrent cost and debt service implications, the consequences in terms of sustainability, the demands for mobilizing new domestic resources, maintaining programmes after the phasing out of external support?

STORE STO

مەمىيە ئىلغانلىق () . C

issues paper ino. 4

Making better use of health resources

In addition to issues concerning the economic implications of various health policies and strategic options, and the problems of resource mobilization. allocation, planning and budgeting, the management of existing resources and services can be improved with benefit in terms of social relevance, equity, efficiency and effectiveness. In particular, improved management of existing resources will lead to higher efficiency (greater coverage with critical services) and high quality, both of which will result in greater health improvement for a given level of public and private investment.

This discussion on the management of health resources can be focused on the three issues of efficiency, cost-effectiveness, and resources accountability and cost-containment.

Among the points that might be discussed under these issues are:

- All health service administrators have at-one time or-another expressed concern that more services and health benefit could be derived from existing resources if only they could be organized and managed better.
 - a) How have those areas of the service with least efficiency been identified?
 - b) What steps can be taken to raise efficiency? Is there a conflict between efforts to improve efficiency and the desire to increase equity in the accessibility of services? How have problems of coordination,

₩₩₩₽₽₽₽₽₽₽₽₽₽₽₽₽₽₽₽₽₽₽₽

.

over-specialization and integration of services been tackled?

- Ensuring that those services and technologies which are given high priority during planning and resource allocation actually receive priority attention in the delivery of services is a continuing management responsibility.
 - a) What methods have been used for choosing the most cost-effective and appropriate technologies and strategies for use within health services?
 - b) How have programme managers been able to ensure that their service providers continue to emphasize priority services?
 - c) How have the most cost-effective mixes and uses of health manpower been arrived at and implemented?
 - d) How has the *quality* of government health services been monitored, and what means have been employed for improving quality?
- III. Contemporary health administrations in most countries are forced to deal with increasing responsibility in the face of escalating inflation and more stringent government budgetary practices. A waste of existing resources is an over-riding problem which all responsible administrators are seeking to reduce.

.....

angala tip dalam

6.35

Line Construction and a second

118

1

à.

CONTRACTOR STATES STATES

2

- a) Do we have examples in which governments have successfully identified causes of waste in resource use and implemented corrective actions?
- b) Are there particular cost-containment strategies which can be recommended?
- c) What has been the effect of decentralizing managerial and financial autonomy to peripheral levels of the health services? Does the authority to utilize the fees collected improve efficiency and cost-control at these levels? What examples are there of community control over health resources?

Annex 1

Glossary of terms

Accounting

The recording of income and expenditure, and of the balance between them; over a period of time.

Benefits

Gains, whether material or not, accruing to an individual or a community.

Budget

A plan in financial terms for the carrying out of a programme of activities during a specified period of time.

Capital costs

Expenditure on physical assets, such as land or buildings, that provide benefits over a period of years.

Capitation

A tax or fee levied per head of a given population, or a grant per head of a given population entitled to a certain benefit.

Cost-benefit analysis

A form of economic analysis in which all the costs and benefits of an activity are expressed in common monetary terms. This_ analysis aims at assessing the activity's ratio of benefits to costs against those of other alternatives or against established criteria.

Cost containment

Measures taken to reduce the rate of growth of expenditure.

Cost-effectiveness analysis

A form of economic evaluation in which this analysis seeks to determine the costs and the effectiveness of an activity, or to compare similar alternative activities, in order to determine the relative degree to which they will achieve the desired objectives. The preferred action or alternative is that which requires the least cost to produce a given level of effectiveness or provides the greatest effectiveness for a given level of cost. The costs are expressed in monetary terms but some of the consequences are expressed in physical units (e.g., life years gained or cases of disease detected).

Deductible

A fixed sum, specified in an insurance policy, that is deducted from any claim made under that policy (and that is therefore paid by the beneficiary of the policy), the remainder of the claim, or a portion thereof, being paid by the insurer.

Demand

The type and quantity of service or commodity that consumers wish and are able to buy in a given period of time.

Economic evaluation

A process whereby the costs of programmes, alternatives or options are compared with their consequences in terms of improved health or savings in resources.

Effectiveness

The extent to which an activity achieves its objectives (e.g., the adequacy of delivery of a treatment to the members of a population that could benefit from it).

Efficiency

The extent to which implementation of an activity produces the greatest product at a given cost or a specified level of production at the lowest cost. Efficiency is therefore a measure of how well inputs are converted into outputs.

Fee-for-service

A payment per item of service.

Fixed costs

Production costs which tend to be unaffected by variations in the volume of output.

Goods

Tangible commodities such as cars, coats, or food.

Gross national product

The value of goods and services produced within a nation plus net income from abroad (the value of exports minus that of imports) during a specified period of time.

Incremental cost

The additional cost of one programme, alternative, or option over and above another.

Inputs

Goods, services, personnel, and other resources provided for an activity with the expectation of producing outputs and achieving the activity's objectives.

Investments

Strictly, in economics, expenditure on real capital goods (e.g., equipment, buildings, or roads); in everyday use, the term applies also to expenditure on other assets from which a return is expected (e.g., stocks and shares).

Marginal cost

The additional cost of producing each successive increment of output.

Monitoring

Continuous oversight (in the sense of supervision) of the implementation of an activity, with the objective of ensuring that input deliveries, work schedules, and other required actions are proceeding, and that targeted outputs are being achieved, according to plan.

National income

A measure of the monetary value of the goods and services becoming available to a nation from economic activity during a prescribed period.

Nominal fee

A fee considerably below the fee normally charged on the open market (by analogy to "nominal value" is the value given to shares in companies when they are created 2

Output

The product(s) that an activity is expected to produce from its inputs in order to achieve its objectives; the quantity of good or services produced in a given time period

Prepayment

Payment made in advance giving a guarantee of eligibility to receive a service when needed at reduced or zero additional cost at time of use (e.g. insurance premiums, membership dues, crop share contributions).

Programme budgeting

Making a budget for a programme.

Real terms

Taking account of, or adjusting for, the changing value of money.

Recurrent costs (or operating costs)

Expenditure that recurs i.e. the costs of running an activity such as salary and raw material costs. That is not devoted to the acquisition of capital assets.

Revenue planning

Plotting future income.

Social cost

Costs to society and not merely to the individual or agency carrying out the activity that do not appear in financial accounts (e.g., the costs of air pollution, noise, congestion).

Subsidy

A grant of money to an organisation or an individual from a government or other agency.

ciussary un rennis

121

Unit cost

1

Cost per unit of production

User charges

CHARGER BROCK AND COMMON CO

A payment made by the user of a service; it may or may not be related to cost.

ATTICX 2

1

CONCERNMENT AND A DESCRIPTION OF A DESCR

Evaluation of the strategy for health for all by the year 2000

Global indicators on resource allocation

Two global economic indicators on resource allocation have been selected for the monitoring and evaluation of the Global Strategy for Health for All: the proportion of GNP spent on health and the proportion of national health expenditures devoted to primary health care. In the first evaluation of the Strategy, the use of these indicators raised some problems relating to definition, data availability and interpretation. Thus, the information that has been collected is not as valuable as expected, and in any event, changes in resource allocation in quantitative terms cannot be measured at this stage. There is still a need to develop additional and more sensitive and appropriate indicators for monitoring and evaluating resource allocation and utilization.

The following table shows the number of countries in which at least 5% of the GNP is spent on health, but the number of countries achieving the 5% target is certainly over-stated. Indeed, some countries have given ministry of health expenditures as a proportion of the national budget rather than public and private health expenditures as a proportion of the GNP.

The data provided for the second global economic indicator is summarized in the following table. This shows the number of countries in which a percentage of the national health expenditures is devoted to primary health care. Because of the great variations in the interpretation of "local health services", no valid conclusions can be drawn from the table.

Proportion of GNP spent on health, by WHO regions

Percentage of GNP spent on health	Number of countries									
	Africa	Americas	S.E. Asia	Europe	Eastern Medit.	Western Pacific	TOTAL			
Less than 5% and more	21 3	13 14	9 1	9 26	15 5	5 7	72 56			
Sub-total No Information	24 20	27 7	10 1	35	20 2	12 8	128 38			
Total	44	34	11	35	22	20	166			

123

Proportion of national health expenditures devoted to primary health care, by WHO regions

ter en	Number of countries									
- Percentage	Africa	Americas	S.E. Asia	Europe	Eastern Medit.	Western Pacific	TOTAL			
Less than 20%	4	6	· · · · ·	3	2	4	19			
20.0%-29.9%	4		1	5	4	3.	17			
30.0%-39.9%	3	· · · · · ·	3	2	3	1	12			
40.0%-49.9%	5		1	3	3	1	13			
50.0%-59.9%	3	1	- 1	1	1	5	12			
60.0%-69.9%	2		2	·	1		5			
70.0%-79.9%										
80% and more	<u></u>	1	1	1	1 at		4			
Sub-total	21	8	9	15	15	14	82			
No information	23	26	2	20	7	. 6	84			
Total	44	34	11	35	22	20	166			

۶

Source: WHO. Evaluation of the Strategy for Health for All by the Year 2000. Seventh report on the world health situation 1986, document A39/3, pp 50-51.

wooded well with with the state of the state

Annex 3

Bibliography

This document has drawn on a wide variety of sources. Major relevant publications and documents are listed below by chapter.

Introduction

- 1. WHO/UNICEF. Primary Health Care. Report of the International Conference on Primary Health Care, Alma-Ata, USSR, 6-12 September, 1978. Geneva, World Health Organization, 1978.
- 2. WHO. Intersectoral action for health, Background document to the Technical Discussions at the Thirty-ninth World Health Assembly. Geneva, 1986.

Chapter I

- ELBAZ, S. Economic and financial constraints and its impact on health provision in Egypt. Draft paper for the Technical Discussions at the Fortieth World Health Assembly, 1987.
- GOLDSBOROUGH, D. and ZAIDI, I. M. How performance in industrial economies affects developing countries. Finance and Development, Vol. 23, No. 4, 1986.
- GUNATILEKE, G., ed. Intersectoral linkages and health development. Geneva, World Health Organization, 1984 (WHO Offset Publications, No. 83).
- 4. HALSTEAD, S.B. et al., ed. Good health at low cost. New York, Rockefeller Foundation, 1985.

₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩

- JOLLY, R. and CORNIA, G.A. The impact of world recession on children. Oxford, Pergamon Press, 1985.
- 6. KOHLER, L. and MARTIN, J., ed. Inequalities in health and health care. Göteborg, The Nordic School of Public Health, 1985.
- 7. INTERNATIONAL MONETARY FUND. Government financial statistics yearbook 1985. Washington, DC, 1985.
- UNCTAD. Trade and development report – 1986. Geneva, UNCTAD, 1986.
- 9. UNDP. Adjustment and growth with human development. Salzburg statement, North South Round Table. New York, 1986.
- 10. UNITED NATIONS ECONOMIC AND SOCIAL COUNCIL. Overall socio-economic perspective of the world economy for the year 2000. Report of the Secretary-General. Document 1985/102, 4 June 1985.
- UNICEF. Within human reach. A future for Africa's children. New York, 1985.
- UNICEF. The state of the world's children – 1984, 1985, 1986, 1987. New York, Oxford University Press, 1984-1987.
- 13. UNITED NATIONS. United Nations National accounts statistics main aggregates and detailed tables, 1982, and computer tapes. New York, United Nations, 1985.
- WORLD BANK. World Development Report 1985 and 1986. New York, Oxford University Press, 1985 and 1986.

125

- 15. WHO. The role of health economics in national health planning and policy making. Copenhagen, WHO Regional Office for Europe, 1987.
- WHO. Global Strategy for Health for All, "Health for All Series", No. 3. Geneva, WHO, 1981.
- ABEL-SMITH, B. Cost-containment in 12 European countries. World Health Statistics Quarterly, Vol. 37, No. 4, 1987, pp 351-363.
- WHO. Repercussions of the world economic recession – Provisional report by the Director-General. Geneva, WHO, 1986 (document A39/4).
- 19. WHO. Evaluation of the Strategy for Health for All by the year 2000. Seventh report on the world health situation. Geneva, WHO, 1986 (document A39/3).

Chapter II

- 1. Plan Financiero Maestro para el Plan Quinquenal 1986-1990. Caja Costariccense de Seguro Social, San Jose, Costa Rica, 1986.
- 2. CUMPER, G. The costs of primary health care. Tropical Doctor, 14, 19-22, 1984.
- 3. DRUMMOND, M.F. and STODDART, G.L. Principles of economic evaluation of health programmes. World Health Statistical Quarterly, 38(4), 1985.
- 4. GRIFITHS, A. and MILLS, M. Money for health: A manual for survey in developing countries, 1982.

TERRET A CENTRE RECORDER DE CONTRACTOR DE CONTRACTOR DE CONTRACTOR DE CONTRACTOR DE CONTRACTOR DE CONTRACTOR DE

5. LAGERGREN, M. Sweden's plan fo 2000. World Health, May 1986, pp 7-9. . .

- 6. LEE, K. and MILLS, A., ed. The economics of health in developing countries. Oxford, Oxford University Press, 1983.
- 7. MACH, E.P. and ABEL-SMITH, B Planning the finances of the health sector: A manual for developing countries. Geneva, WHO, 1985.
- 8. MERKLE, A. The cost of health for all. A feasibility study from Upper Volta. Eschborn, German Agency for Technical Discussion, 1982.
- 9. MWABU, G. Financial health planning, Kenya (Nairobi, 1986). Di paper for the Technical Discussions the Fortieth World Health Assembly 1987.
- 10. Financial survey of health and social welfare 1987, including a cost estima 1987-1990, Health for all policy document by 2000. The Hague, <u>Ministry of Welfare, Health and</u> Cultural Affairs, 1986.
- 11. PARET, H. Planification de la santé en Afrique. L'Harmettan, Paris, 198
- 12. Public Finance International. Financ management in the health sector of developing countries. Background paper for the World Bank. London, Public Finance International, 1982.
- 13. THOMAS, M. Problems of recurrer costs in the health of developing countries. WHO/SHS document, 19
- WHO. Managerial process for natio health development. Guiding principles. "Health for All Series", 1 5, Geneva, WHO, 1981.

- 15. WHO. Guidelines for costing primary health care development. Unpublished document SHS/84.3 (1984).
- KLECZKOWSKI, B., ROEMER, M. and VAN DER WERFF, W. National health systems and their orientation towards health for all. Guidance for policy-making. Geneva, WHO, 1984 (Public Health Papers, No. 77).
- 17. KLECZKOWSKI, B., ELLING, R. and SMITH, D. Health system support for primary health care. Geneva, WHO, 1984 (Public Health Papers, No. 80).
- WHO. Regional Office for South-East Asia. Financial planning for Health for All by the Year 2000. New Delhi, WHO, 1984.
- 19. WHO. Programme budgeting as part of the managerial process for national health development. Guiding principles (document MPNHD/84.2). Geneva, WHO, 1984.
- 20. WHO. Planning of the finances of Health for All. Report by the Director-General. Geneva, WHO (document EB77/INF.DOC./1, November 1985), 1985.

;

- 21. UNICEF/WHO. National decision-making for primary health care. Geneva, WHO, 1981.
- 22. WILSON, R.F. et al. Planning and management of primary health care programmes. Report of a Workshop. Geneva, The Aga Khan Foundation, 1984.

and the second second

Chapter III

- COMMONWEALTH. Development cooperation for health in Commonwealth countries. Part I and II – Meeting reports, Nassau, The Bahamas, 13-17 October 1986.
- 2. CUMPER, G. Health sector financing, estimating health financing in developing countries. London, London School of Hygiene and Tropical Medicine, .
- 3. de FERRANTI, D. Strategies for paying for health services in developing countries. World Health Statistical Quarterly, Vol. 37, No. 4, pp 428-450, 1984.
- 4. GROSS, P.N. et al. Revolving drug funds: concluding business in the public sector. Social Science and Medicine, Vol. 22, No. 3, pp 335-343, 1986.
- KAL HONG PHUA. Singapore's family saving scheme - World Health, May 1986, pp 11-12.
- 6. MUSGROVE, P. What should consumers in poor countries pay for publicly-provided health services? Social Science and Medicine, Vol. 22, No. 3, pp 329-333, 1986.
- MWABU, G.M. and MWANGI, W.M. Health care financing in Kenya

 A simulation of welfare effects of user fees. Social Science and Medicine, Vol. 22, No. 7, pp 763-767, 1986.
- 8. OK RYUN MOON. Towards equity in health care. World Health, May 1986, p. 20.

127

-

- RUSSELL, S.S. and REYNOLDS, R.J. Community financing. PRICOR Monograph Series. Issues Paper 1. PRICOR, USA, 1985.
- RUSSELL, S.S. and ZSCHOCK, D.K. Health care financing in Latin America and the Caribbean. HOFLAC, State University of New York, Research Report No. 1, Stony Brook, New York, 1986.
- 11. SENE, P.M. Community financing in Senegal, World Health, May 1986, pp 4-5.
- 12. SHEPARD, D.S. and BENJAMIN, E.R. Mobilizing resources for health. User fees and health financing in developing countries. Paper for the second Takemi Symposium on International Health. Boston, Harvard School of Public Health, 1986.
- STINSON, W. Community financing of primary health care. Primary Health Care Issues, Series 1, No. 4, Washington, DC, American Public Health Associates, 1982.
- 14. WHO. Guidelines for preparation of the country health resource. Utilization Review Document. Unpublished WHO document COR/HRG/82.1, 1982.

Chapter IV

- 1. ABEL-SMITH, B. Cost-containment in 12 European countries. World Health Statistics Quarterly, Vol. 37, No. 4, 1984, pp 351-363.
- AMERICAN PUBLIC HEALTH ASSOCIATION. Primary health care: progress and problems. An analysis of 52 projects, assisted by the Agency for International Development of the United States. Washington, DC, 1982.

AND MARKED DESCRIPTION OF DESCRIPTION OF

- 3. BANKOWSKI, Z. and FULOP, T. Health manpower out of balance: conflicts and prospects. Geneva, Council for International Organization of Medical Sciences, 1987.
- BARNUM, H.N. Cost-savings from alternative treatment of tuberculosis. Washington, DC, World Bank, 1986 (PHN Technical Note No. 86.11).
- BLANPAIN, J.E. Cost-containment has already been achieved in some countries. World health forum, Vol. 6. p 11 (1985).
- 6. BLUMENFELD, S.N. Operations research methods: A general approach in primary health care. PRICOR Monograph Series. Methods Paper 1, PRICOR, USA, 1985.
- CATSAMBUS, T. and FOSTER, S. Spending money sensibly: the case of essential drugs. Finance and Development, Vol., pp 30-32, 1986.
- CREESE, A.L. and HENDERSON, R.H. Cost-benefit analyses and immunization programmes in developing countries. Bulletin of the World Health Organization, 58(3), 491-497, 1984.
- CULYER, A.J. and HORISBERGER.
 B. Economic and medical evaluation of health care technologies. Springer Verlez – Berlin Heidelberg, 405 p, 1983.
- 10. DRUMMOND, M.F. and MILLS, A Survey of cost-effectiveness and cost-benefit analyses of key primary health care projects in commonwealth countries. Commonwealth Secretariat. London, 1986.

11. EDSTROM, K. Assessment of the expansion phase in Beheira government. UNICEF-supported Dayas training programme in Egypt. 3 July 1986.

18

- GOLOTEEV, V.V. and PUSTOVOJ, I.V. Public health finance and planning in the Soviet Union – World Health Statistical Quarterly, Vol. 37, No. 4, 364-374, 1984.
- 13. GROSSE, R.N. and PLESSES, D.J. Counting the cost of primary health care. World health forum, Vol. 5, pp 226-230 (1984).
- GWATKIN, D.R., WILCOX, J.R. and WRAY, J.D. Can health and nutrition interventions make a difference? Overseas Development Council Monograph, No. 13, 1983.
- 15. JORDAN, P. Schistosomiasis: comparison of control strategies. Cambridge University Press, 1985.
- KAEWSONTHI, S. Cost and performance appraisal-of malariasurveillance and monitoring measures. Final Report of the WHO/UNDP/World Bank TDR Project, No. 800435, 1983.
- MILLS, A. Economic evaluation of health programmes – Application of the principles in developing countries. World Health Statistical Quarterly, Vol. 38, No. 4, 1985.
- MILLS, A. and THOMAS, M. Economic evaluation of health programmes in developing countries. A review and selected annotated bibliography. London, London School of Hygiene and Tropical Medicine, 1984.

- 19. OVER, M. The effect of scale on cost projections for a primary health care programme in a developing country. Social Science and Medicine, Vol. 22, No. 3, pp 351-360, 1986.
- 20. PAUL, S. Managing development programme: the lessons of success. Westview Special Studies, Boulder Colorado, USA, 1982.
- 21. PIACHAUD, D. Medical equipment in Third World countries. Journal of Tropical Medical Hygiene 85, pp 183-185, 1982.
- 22. **REYNOLDS, J.** and **GASPARI, K.C.** Cost-effectiveness analysis. PRICOR Monograph Series: Methods Paper 2, PRICOR, USA, 1985.
- 23. ROBERTSON, L. et al. Service volume and other factors affecting the costs of immunizations in The Gambia. Bulletin of the World Health Organization, 62(5), 729-736, 1984.
- 24. SCHIEBER, G.J. The financing and delivery of health care in OECD countries: past, present and future. OECD Report, Tokyo, 25-28 November 1985.
- 25. WALSH, J.A. and WARREN, K. Strategies for primary health care. Technologies appropriate for the control of disease in the developing world. Chicago, University of Chicago Press, 1986.
- CREESE, A. EPI Programme costing guidelines. Weekly epidemiological record, 55, 281-288 (1980).
- 27. WHO. Guidelines for health care practice in relation to cost-effectiveness. Copenhagen, WHO Regional Office for Europe, 1981 (Reports and Studies No. 53).

129

- WHO. Control of health care costs in social security systems. Copenhagen, WHO Regional Office for Europe, 1982 (Reports and Studies No. 55).
- WHO. Drug policies and management: procurement and financing of essential drugs. Report of a Meeting, Madrid, 22-26 October 1984. Unpublished WHO document DAP/84.5, 1984.
- 30. WHO. Kenya: rural drug distribution programme. The new drug management system for rural health facilities, Kenya. WHO Action Programme on Essential Drugs and Vaccines. Geneva, WHO, 1985.
- 31. WHO. Financial and economic aspects of environmental management and its cost-effectiveness as vector control measures. Draft report of the Sixth Annual Meeting of the panel of experts on environmental management in vector control, 7-12 September 1986, Geneva.
- 32. WORLD BANK. World development report – 1985. Washington, DC, World Bank, 1985.

Chapter V

RRR

nore.

1. ALFILER, M.C. Partnership and responsibility in financing health activities in the Philippines. Draft report for the Technical Discussions at the Fortieth World Health Assembly, 1987 (Manila, October 1986).

TORNAL DE LE CONTRACTOR DE

- BERCHE, T. A propos d'une O.N.G. de developpement sanitaire: L'église catholique en Afrique et les soins de santé primaires. Sciences Sociales et Santé, Vol. III, No. 3-4, novembre 1985.
- 3. DAVIS, K. Recurrent health cost issues related to effectiveness and equity: the US experience. Draft paper for the Technical Discussion at the Fortieth World Health Assembly, 1987.
- MAHLER, H. The meaning of "Health for All by the Year 2000" World health forum, 2(1), 8 (1981).
- 5. MWABU, G.M. Health care decisions at the household level: results of a rural health survey. Social Science and Medicine, Vol. 23, No. 3, 1986.
- 6. ROEMER, M. Private medical practice: obstacle to health for all. World health forum, Vol. 5, No. 3 pp 195-211, 1984.
- RONDINELLI, D.A. et al. Decentralization in developing countries. Washington, DC, Work Bank (Staff Working Paper, No. 581, Management and Development Series, No. 8).
- 8. TAMESIS, J.V. Partnership and responsibility in financing health activities: the Medicare programm of the Philippines. Draft report for the Technical Discussions at the Fortieth World Health Assembly. 1987 (Manila, October 1986).

- 9. Promoting health, preventing disease, objectives for the nation. Department of Health and Human services, USA, 1980, reprinted in 1984.
- von THUN, F. and ULLRICH, G.J. (rapporteur and editor). Fighting rural poverty through self-help: International Conference Report. Deutsche Stiftung fur Internationale Entwicklung, Feldafing, FRG, 1985.

2

- Planning for equity in health A sector review and policy statement. Harare, Ministry of Health, 1986.
- 12. WHO. Collaboration with nongovernmental organizations in implementing the Global Strategy for Health for All. Background document for the Technical Discussions at the Thirty-eighth World Health Assembly, 1985 (A38/Technical Discussions 1).

131

i i po